

LOW-LEVERAGE STATES IN INTERNATIONAL MEDIATION: A COMPARATIVE CASE STUDY OF NORWAY AND QATAR

by

Ida Bary

A thesis submitted to the University of Birmingham for the degree of

DOCTOR OF PHILOSOPHY

Department of Political Science and International Studies

School of Government and Society

College of Social Sciences

University of Birmingham

June 2022

UNIVERSITY^{OF} BIRMINGHAM

University of Birmingham Research Archive

e-theses repository

This unpublished thesis/dissertation is copyright of the author and/or third parties. The intellectual property rights of the author or third parties in respect of this work are as defined by The Copyright Designs and Patents Act 1988 or as modified by any successor legislation.

Any use made of information contained in this thesis/dissertation must be in accordance with that legislation and must be properly acknowledged. Further distribution or reproduction in any format is prohibited without the permission of the copyright holder.

ABSTRACT

Although there has been considerable research on various aspects of international mediation, few studies have specifically addressed the issue of low-leverage states and their potential as mediators. Low-leverage states do not fit neatly into the traditional categories of "pure" or "power" mediators because of their unique status as states who often have an interest in the conflict and yet lack the same level of leverage possessed by powerful states to use in their mediation.

Through their engagement in the international arena, low-leverage-states adopt different approaches to international mediation. This thesis contributes to the existing knowledge and the understanding of the scholarly literature by presenting two different approaches for low-leverage states in international mediation: norm-driven mediation and interest-driven mediation. Those two approaches to mediation are represented by Qatar and Norway. While Norway represents norm-based mediation, Qatar represents interest-based mediation. To find out the reasons behind the different types of engagement by low-leverage states in international mediation, this thesis answered the following main research question: Why do low-leverage states engage differently in international mediation? In addition, the following sub-questions were asked: What are the motivations and the capabilities of interest-based low-leverage mediators? What are the motivations and the capabilities of norm-based low-leverage mediators? Those questions were answered through a qualitative comparative case study of Qatar and Norway. This thesis makes a significant contribution to our knowledge by drawing on more than 30 semi-structured

interviews with elite representatives, including diplomats, political leaders and politicians from different countries such as Norway, Israel, Qatar, Sudan, South Sudan, Darfur, the Palestinian Authority and Hamas.

Keywords: low-leverage-states, international mediation, mediation strategy, power mediation, pure mediation, impartiality, norm-based mediation, power-based mediation.

ACKNOWLEDGEMENT

I would like to express my sincere gratitude to my supervisors, Dr Asaf Siniver and Dr Stefan Wolff for the continuous support of my PhD studies - for their patience, motivation, and immense knowledge. Their guidance helped me throughout the research and writing of this thesis. I could not have imagined having better supervisors and mentors for my PhD thesis. Thank you for always being there to support me. I would like also to express my sincere thanks to the Norwegian loan Fund" Lånekassen" for funding my study. I am thankful to Stig Jarle Hansen for supporting my study.

I am grateful to the Qatari minister of Foreign Affairs, His excellence Sheikh Mohammed Bin Abdulrahman Al-Thani. I would also like to express my sincere thanks to Mr Jaber Al-Harmi for his help and support. Without his support, I would not be here. I am incredibly grateful to Mr Khaled Mishaal for being supportive. I would like to express my sincere thanks to all the people who accepted to be participants in my thesis.

I would like to thank my little man, my son Omar, for being supportive and for motivating me. I still remember his encouraging words: *Mum, do not give up - you are almost there!*

Thanks to my best friend Kira for being my friend and for being there whenever I need help. Last, but not the least, I would like to thank my husband Tarek Eslim for supporting me spiritually throughout writing this thesis and my life in general.

TABLE OF CONTENTS

CHAPTER ONE: INTRODUCTION AND METHODOLOGY5
CHAPTER TWO: LITERATURE REVIEW AND CONCEPTUAL FRAMEWORK 29
CHAPTER THREE: NORWAY VERSUS QATAR IN INTERNATIONAL MEDIATION 65
CHAPTER FOUR: CASE STUDY ONE: NORWAY AND THE OSLO ACCORD 1993 . 116
CHAPTER FIVE: CASE STUDY TWO: NORWAY'S ROLE IN THE COMPREHENSIVE
PEACE AGREEMENT (CPA) IN SUDAN 2005 163
Norway, Troika, IGAD in the mediation process175
CHAPTER SIX: CASE STUDY THREE: THE DOHA DOCUMENTS FOR PEACE IN
DARFUR (DDPD),2011207
The Qatari International mediation approach for Peace in Darfur 216
CHAPTER SEVEN: CASE STUDY FOUR: FATAH-HAMAS DOHA AGREEMENT (2012)
The Qatari International mediation approach in mediation between Hamas and Fatah
CHAPTER EIGHT: FINDINGS AND CONCLUSION 269
The mediation strategy in Oslo Accord and in the CPA275
REFERENCES286

CHAPTER ONE: INTRODUCTION AND METHODOLOGY

Introduction

This thesis examines the role of low-leverage states as mediators in international conflict. Since the end of the Cold War, an increasing number of conflicts have provided opportunities for mediation (Bercovitch, 2011b). At the same time, mediation by low-leverage states has grown significantly (Egeland, 1998; Goetschel, 2013a). Norway, Sweden, Finland, Denmark, Ireland, Austria, Vatican City, South Africa, Switzerland, and Qatar are examples of low-leverage states that have been involved in a disproportionate number of international mediations (Slim, 1992). However, few studies on international mediation have focused on the mediation conducted by those states (Schulz, 2016; Slim, 1992). This thesis considers Norway and Qatar as examples of case studies. Norway and Qatar are both low-leverage states. However, they are engaged in international mediation in different ways. While Norway is a normbased mediator, Qatar is an interest-based mediator.

To understand the ways low-leverage states engage themselves in international mediation, this thesis sought to answer the following main research question: Why do low-leverage states engage differently in international mediation? In addition are the following sub-questions: What are the motivations and the capabilities of interest-based low-leverage mediators? What are the motivations and the capabilities of norm-based low-leverage mediators?

The research questions are answered through a qualitative comparative case study of two low-leverage states, Norway and Qatar.

International mediation scholars (Smith,1994; Svensson, 2007b; Zartman & Touval,1985; Zartman & Touval,1996) suggest two categories of mediators: pure mediators and power mediators. A power mediator has both interest and leverage to push the disputants towards embracing a compromise (Herriott & Firestone, 1983). A typical pure mediator lacks both leverage and specific interest in the conflict (Svensson, 2007b). While scholars have focussed on the power mediator who possesses considerable leverage and the pure mediator who lacks it, there have been fewer studies of the mediators who possess limited leverage and are represented by low-leverage states. Low-leverage states do not fit neatly into the categories of "pure" or "power" mediators because of their unique status as being full states who often have an interest in the conflict (Zartman & Touval, 1996) and yet lack the same level of leverage possessed by many larger states to use in their mediation (Slim, 1992).

The mediators are classified into two categories based on interest: Normbased mediators and interest-based mediators. The norm-based mediation means that the mediator is unlikely to have a vested interest in the conflict or the mediation outcomes. Additionally, the mediator is driven by norms and values and aims at promoting international peace and ending the conflict. Those mediators lack specific interest in the mediation process and are driven by liberal normative motivation (Leira, 2013).

In direct contrast to norm-based mediation is interest based-mediation. Interest-based mediators usually engage themselves in mediation due to interest-based motives. These interests may take the form of either a security strategy or a way of increasing status (Leira, 2013). This is not to say that interest-

driven mediation is better than normative-driven mediation or vice versa, but it is important to note that each of them possesses unique tools in international mediation, given their different sources of motivation.

This thesis seeks to contribute to the existing knowledge and increase understanding of the scholarly literature in different ways. First, it discusses low-leverage states in international mediation, a topic that is insufficiently covered in the current literature. Taking this a step further, their location is placed on the spectrum of leverage. Secondly, the behaviour of low-leverage states in international mediation is explored, where they have a limited level of influence and behave in a certain way. Thirdly, the research explored which strategy/ies low-leverage states employed in their mediation process.

This research contributes to the existing knowledge by evoking two types of mediation approaches by low-leverage states. The approaches are interest-driven and norms-driven, represented by Norway and Qatar, respectively. Focusing on two specific low-leverage state mediators with different approaches, Qatar and Norway, the thesis aims to examine those two approaches: a norm-driven approach and an interest-driven approach.

Two cases represent Norway's mediation experience: its mediation leading up to the Oslo Accord 1993 and its mediation role in the Comprehensive Peace Agreement (CPA) in Sudan 2005. Two cases also represent the Qatari mediation experience: the Qatari mediation in The Doha Document for Peace in Darfur (DDPD) 2011 and Qatar's mediation role in reconciliation between Fatah and Hamas, which was accumulated by the Fatah- Hamas Doha Agreement

or what is known as "The Doha Palestinian Unity Agreement".

The context of the study and problem statement

Since the end of the Cold War, low-leverage states have played a more active role in international relations than in the past, especially in security affairs and promoting peace (Steinmetz, 2016). By the end of the Cold War, the increasing number of ethnic and internal conflicts was parallel to the growing role of low-leverage states in international mediation and provided opportunities for mediation (Bercovitch, 2011b; Goetschel, 2013b). The specific problem of interest was that the role of low-leverage states in mediation had not been given adequate attention in the literature. International mediation literature has conceived of mediators as being either "pure" or "power" mediators (Smith, 1994). However, low-leverage states do not fit easily into this dichotomy because their position straddles the line between these categories.

The low-leverage state cannot match the influence of a power mediator as it does not have the same level of leverage as the powerful mediators. Despite this, it is still a state with some degree of leverage (economic means) and additional leverage resulting from its ties to the powerful states. Additionally, like other states in mediation, the low-leverage state usually has an interest in the mediation outcomes (Zartman & Touval, 1996), meaning that it may not be entirely a pure mediator (Böhmelt, 2015).

Thus, more research is needed to understand the drive for low-leverage states to act as mediators in conflicts. In addition, the tools and strategies that the low-leverage states employ in their mediation can be better understood. This

is compounded by the fact that success in international mediation is challenging to define and can be considered in many ways (Bercovitch, 2011b; Folberg & Taylor, 1986; Kelman, Bercovitch, & Rubin, 1992; Kleiboer, 1996; Lall, 2014; Mitchell, 1988; Moore, 2014; Vuković, 2014). Therefore, the problem requires a better understanding of low-leverage state motivation for filling the role of mediator in conflicts to appreciate their approach to international mediation.

Intended Research Contribution

The intended contribution of the thesis findings is to present two different approaches of international mediation adopted by low-leverage states: a norm-driven mediation approach and an interest-driven mediation approach, represented by Norway and Qatar, respectively. Few scholars beyond Slim (1992) and Eriksson (2015) have directly studied low-leverage states as mediators, even though low-leverage state mediation has grown in prominence since the end of the Cold War (Goetschel, 2013b). Therefore, this represents a broad gap in the current literature, which this thesis will fill.

However, even beyond serving to fill this gap, this thesis helps answer several calls for research in the literature. Firstly (Kelleher, 2006) calls for further studies on the Norwegian approach to mediation, as he mentions that it is worth studying. This thesis answers the call of both Bercovitch and Fretter for the need to use real-world cases and compare mediation processes to effectively study mediation (Bercovitch, 2011b; Bercovitch & Fretter, 2007). Additionally, this thesis is further based on Svensson's research on pure mediators and

power mediators. Svensson was adamant about the lack of progress in the debate on power mediators and pure mediators (Svensson, 2007b).

Furthermore, this thesis explores an aspect Bercovitch, and Houston (2000) ignored. They mention that communication facilitation strategies led the mediators to be passive in the mediation process and restricted their role to merely channelling information to the disputants due to a lack of leverage or any means to influence the negotiations (Bercovitch & Houston, 2000). Bercovitch and Houston did not mention the case of low-leverage states who possess limited leverage and choose to be either a facilitator, such as Norway or a manipulator, such as Qatar.

The overall structure of the thesis takes the form of eight chapters, including this introductory chapter which includes the research methodology. Chapter two covers the literature review and the conceptual framework. Chapter three offers a comprehensive background on Norway and Qatar in international mediation. Chapter four covers the first case study: Norway and the Oslo Accords 1993. Chapter five discusses the second case study: Norway's role in the Sudanese Comprehensive Peace Agreement (CPA) in 2005. Chapter six presents case study three: The Doha Document for Peace in Darfur (DDPD) in 2011. Chapter seven handles the fourth case study: the Fatah-Hamas Doha Agreement in 2012. The final chapter draws upon the entire thesis. It consists of two parts: research findings and conclusion.

Methodology

The methodology for this thesis is a qualitative comparative case study. Qualitative research is a descriptive, exploratory approach to research that takes a holistic, subjective approach to explore a phenomenon (Cho & Trent, 2006). Rather than taking a phenomenon out of its context and studying it in a purely abstract sense, the qualitative approach allows the researcher to study it in its native context and thereby more fully understand it (Creswell, 1994; Ravitch & Carl, 2015).

The type of research question determines the research method (Thomas, 2017; Yin, 2017). "How and why" questions indicate that the study is qualitative. Based on the main research question, "Why do low-leverage states differently engage in international mediation?" it can be established that this is a qualitative, exploratory case study (Thomas, 2017; Yin, 2017). Therefore, comparing qualitative and quantitative methods is not necessary for this thesis (Ravitch & Carl, 2015). Rather than building a model entirely based on existing theory, qualitative research allows the researcher to ask open-ended questions that elicit participants' fuller range of responses (Merriam & Tisdell, 2015). This makes qualitative research apt for understanding problems about which little is known or poorly understood problems (Merriam & Tisdell, 2015).

case study method involving case analysis and a comparison of a small number of cases (Bennett & George, 2004). A case study is preferred when the researcher examines a contemporary issue for which he/she cannot manipulate

the appropriate behaviour and when "how and why" questions are central to the research (Dooley, 2002; Stake, 1995, 2013; Yin, 2017). Thus, the case study is preferred in this thesis because it handles such a contemporary issue: the engagement of low-leverage states in international mediation. Additionally, in this research, the researcher has no control over behavioural events and the most critical aspects that the research question addressed is "why" questions, such as: Why do low-leverage states engage differently in international mediation?

The case study approach is a robust qualitative design that focuses on looking at the phenomenon being studied in a specific case and drawing useful conclusions about the phenomenon (Yin, 2013). Out of the qualitative methods, the case study approach is especially suited to studying context problems because of its depth. Case studies draw data from multiple sources and comparatively analyse these data to create a more accurate depiction of the case under study (Hartley, 2004). Finally, a case study is a powerful tool to identify new variables and generate new hypotheses (George and Bennett 2005, p. 20).

This thesis employs a multiple-case study (Yin, 2017) or case study method (George & Bennett, 2005). In this approach, a researcher analyses two or more cases individually and then carries out cross-case comparisons to see how the cases are similar and how they differ (George and Bennett, 2005; Yin, 2014). Although a single-case study is more easily carried out, it does not give as much contextual information as a multiple-case study. This type of analysis offers more profound insight into contextual factors so long as the two or more

cases are meaningfully different in context. Additionally, a multiple-case study is more robust than a single-case study, and the evidence from a multiple-case study is powerful and more compelling than from a single-case study (Herriott & Firestone, 1983).

Accordingly, the current thesis used a multiple-case study approach that considered four cases for two low-leverage state mediators representing Qatar and Norway. Therefore, cross-case analysis of these four instances allowed the researcher to draw meaningful conclusions about how the two approaches were different. The case study approach is not without its pitfalls; the main criticism of case study research is the lack of generalization.

It is not easy to generalize from one (single-case study) or two or a few cases, multiple-case study (George & Bennett, 2005; Thomas, 2017; Yin, 2013). To bridge this gap, one should distinguish between statistical generalization and analytical generalization. Statistical generalization is used in quantitative studies which contain many subjects, and it is easy in those cases to generalize the findings to the population. Although analytical generalization is specific to case studies, a case study does not represent a sample; case studies are not "sampling units", and their numbers are too small to represent any larger population, so the goal is to expand and generalize theories (Thomas, 2017; Yin, 2013, 2017).

Research Design

Research design links the research questions to the research conclusions. This linkage goes through data collection and data analysis. The research design should address the research objective, the research question/s, relevant propositions or hypotheses (if available), the unit of analysis, case selection and finally the criteria to interpret the findings (George & Bennett, 2005; Stake, 2013; Yin, 2014; Yin, 2017).

Research design links the research questions to the research conclusions. This linkage goes through data collection and data analysis. The research design should address the research objective, the research question/s, relevant propositions or hypotheses (if available), the unit of analysis, case selection and finally, the criteria to interpret the findings (George & Bennett, 2005; Stake, 2013; Yin, 2014; Yin, 2017).

The research objective can be one of the following: theoretical\ configuration idiographic, disciplined configurative, heuristic, theory testing or building studies (George & Bennett, 2005). The objective of this particular study is a heuristic one. The heuristic case study identifies new variables and hypotheses. Outliers or deviant cases are helpful for the heuristic purpose. The heuristic objective in this thesis presents two different approaches of mediation; the interest-driven mediation approach and the norm-driven mediation approach. It should be kept in mind that this thesis focuses on specific aspects of each case study which are considered parameters of the cases to be employed in the data analysis. This research design includes specific questions to cover those

aspects or parameters that reflect the research objective and apply them to each case to standardise data collection and a systematic comparison (George & Bennett, 2005). The main parameters for this thesis are motivation, leverage (capabilities), impartiality/biased, and finally, the implications of being a pure or power mediator.

The second component of the research design is the research question. As mentioned above, the primary research question is, why do low-leverage states engage differently in international mediation? The sub-research questions are: What are the motivations and the capabilities of interest-based low-leverage mediators? What are the motivations and the capabilities of normbased low-leverage mediators?

A comparative case study research methodology is used to answer the main research question and sub-research questions. A case study has two fundamental parts: the subject and the object. The subject is the case itself, and the object is the analytical frame (Thomas, 2017).

Based on the research question, this exploratory study presents two different approaches to international mediation; interest-driven mediation and norm-driven mediation, represented by Qatar and Norway, respectively. The subjects in this thesis are Norway, and two cases of mediation represent Qatar international mediation each of them. The objects or the analytical frames of this thesis are the interest-driven mediation approach and norm-driven mediation approach.

The case selection is based on the preference of the researcher, who selects the cases based on their relevance to the research objective: a heuristic case study intended to identify new variables or parameters. Usually, there are three reasons to choose a specific case study. Firstly, a case may be chosen based on pre-existing local knowledge. In this instance, the researcher already knew much about the case and needed to understand some specific features better. Secondly, a case may be chosen because it is a key case study, well known, and is a good representative example. Finally, the case may be an interesting outlier case study because it departs from the norm (Thomas, 2017). There are many options of countries to be used as case studies for this thesis. For example, there is the opportunity to use Norway and Qatar and Sweden and Oman. It is known that Sweden and Norway are similar in their international mediation. Both of them are Scandinavian countries that adopt a normdriven mediation approach. In this thesis, Norway is preferred over Sweden because Norway is better known than Sweden in the international mediation field due to its mediation in the most famous conflict, the Palestinian-Israeli conflict. Oman is also known for its role as a Middle East mediator. Oman succeeded in bringing the USA and Iran together for historic talks. However, the researcher chooses Qatar over Oman because Oman adopts a mediation approach similar to the Norwegian approach. Qatar is different from Norway. Qatar uses its financial resources as an incentive in the mediation process. In this thesis, Norway and Qatar are each chosen for different reasons. Nor-

way is chosen because Norway represents the most typical of the low-leverage norm-driven mediation approach. Qatar is chosen because it behaves

differently than Norway, despite sharing the same low-leverage status. Qatar represents a different approach in mediation: the interest-driven mediation approach.

Beyond being low-leverage states, however, marked differences emerge when looking at their mediation experiences, resulting from different approaches adopted by each. On the one hand, Norway's mediation approach is based on a combination of impartiality, confidentiality, consistency, and collaboration between its foreign ministry and nongovernmental organizations (Lieberfeld, 1995). Additionally, the Norwegian model represents international mediation's moral and humanitarian image (Höglund & Svensson, 2009). Norway represents a typical normative ideal of a low-stakes mediator representing and promoting liberal Western values (Wivel, 2013), and its mediation promotes peace globally (Leira, 2013). On the other hand, Qatari officials rely on personal contacts for mediation and the Qatari wealth to pay for the disputants to keep talks going. Qatar is regarded as a prominent mediation authority in the Middle East, conducive to Qatari investment (Barakat, 2014).

It is not only the mediators who matter in the selection of the case studies but also the conflicts. Four cases featuring conflict are selected for this research.

The first case is Norway's mediation between the Palestinians and the Israelis, culminating in the Oslo Accord of 1993. The second case is Norway's mediation which led to the Comprehensive Peace Agreement (Sudan) in 2005. The third case is Qatar's mediation in Sudan (Darfur), culminating in the Doha

Agreements in 2010. The fourth case is Qatar's mediation between Hamas and Fateh (2008).

Those conflicts are selected due to their significance. For example, the Palestinian-Israeli conflict is considered one of the most protracted conflicts in the 20th and 21st centuries. The Sudan-South Sudan conflict is significant because it witnessed the genocide of millions of lives. In the same vein, the Darfur conflict is considered a humanitarian crisis. Many people were killed and displaced. Lastly, the Hamas-Fatah conflict is essential due to its connection to the Israeli-Palestinian peace relations.

Furthermore, the case of Norway's mediation in the Oslo accords is selected because it is different from its mediation between the Government of Sudan and the SPLM/A. While in the case of Oslo Accords, Norway was the sole mediator, Norway was among the Troika in the case of its mediation between the Government of Sudan and the SPLM. In the same vein, the case of Qatar's mediation in Darfur's conflict is selected because Qatar was not a sole mediator – Qatar was among a group of mediators that included the United Nations (UN) and the African Union (AU). The case of the Qatari mediation between Hamas and Fatah is selected to represent an example of Qatari as a sole mediator.

To study low-leverage state mediation, the unit of analysis is the phenomenon of low-leverage states' international mediation. To conduct the case studies described in this thesis, a structured, focused comparison method is used. It is structured because the researcher created questions that reflect the research

objectives and applied those questions to each case to standardise the data collection and make a systematic comparison. It is focused on those only specific aspects of the cases are examined. The cases are undertaken with a specific research objective in mind (George & Bennett, 2005; Stake, 1995, 2013; Yin, 2014 and 2017).

Data Collection

To collect data for any case study research, a researcher should follow the case study protocol, use multiple data sources (evidence), create a case study database, and maintain a chain of evidence (Yin, 2014). For data collection in case studies, there is no statistical sampling where the sample should represent the population. In the case of studies, the data are collected on the basis of theoretical sampling, which aims to deeply understand the cases and facilitate theory development (Stake, 1995, 2013; Yin, 2014 and 2017). The common theoretical sampling approach is to select the extreme case pair (e.g. good vs bad). The case study method is more effective when the researcher writes general questions for each case to standardise the data requirements. As a result, it is easier to compare the results from all cases. Unless the researcher asks the same questions of each case, the results cannot be compared and systematically analysed (George & Bennett, 2005; Yin, 2014). Relevant data for a case study may be collected through interviews, participant observation, direct observation, physical artefact collection, document collection, and archival records (Thomas, 2017; Yin, 2013). At least three sources of data should inform a case study to allow for good triangulation of

data (George & Bennett, 2005; Yin, 2013). This thesis uses the following three data sources: interviews with officials and diplomats representing the disputants and the mediators, document collection, and archival records.

This thesis significantly contributes to our knowledge of these case studies by drawing on more than 30 semi-structured interviews. Those interviews were conducted with elite representatives, including diplomats, political leaders and politicians from different countries like Norway, Israel, Qatar, Sudan, South Sudan, Darfur, the Palestinian Authority and Hamas. For example, Khaled Mishaal, Mousa Abu Marzouk, Hussam Badran (Hamas). From Fatah, Azzam Alahmad and Mohamad Abu-Khoush. From Norway, Jan Egeland, the Norwegian ex-foreign minister and Hilde Johnson, the Norwegian ex-minister of development. From Israel, Yossi Beilin and Yoel Singer. From Qatar, the spokesperson of the Ministry of Foreign Affairs (Lowlowa Al-Khater), and the ex-ambassador of Qatar to Sudan. from Sudan, Altigani Sissi and Hassn Omar. From South Sudan, Pagan Amom. Additionally, General Sumbeiywo, the exhead of IGAD.

According to the snowballing method, these interviewees were found where the interviewees recruited other participants or interviewees. While some interviews took thirty minutes, others took up to two hours. Some interviews were face to face interviews in Qatar. Most of the interviews were techno-based interviews conducted via the phone or Skype. The techno-based interviews were chosen due to either finance limitations or security limitations. For example, interviews in Sudan, South Sudan, the Palestinian Authority, and Israel raised security concerns. It was considered that enough data had been

collected in the interviews when the interviewees started to repeat each other or when the issue was clear. These semi-structured interviews were valuable and greatly enriched the thesis.

Interviews are one of the most important sources of evidence in case study data collection-(Darke, Shanks, & Broadbent, 1998). Interviews can be unstructured or semi-structured (Thomas, 2017). Using unstructured interviews may generate rich data and uncover unexpected evidence by utilizing openended questions (Daymon & Holloway, 2010). However, this thesis utilised semi-structured interviews as the primary source of data collection.

Semi-structured interviews allow the researcher to effectively understand the participants' perspectives (Daymon & Holloway, 2010). Semi-structured interviews represent a compromise between open-endedness and topic orientation (Merriam & Tisdell, 2015). That is to say, they have some degree of structure, being prepared in advance by the researcher so as to ensure that the data collected can answer the research questions. However, they are only semi-structured in that this guide does not serve as a script for the interviews so much as a list of topics to cover. The researcher remains free to include additional probing questions if something is unclear, ask follow-up questions if there seems to be more of interest to be said about a topic or diverge if a relevant digression suggests itself in the course of the interview (Turner 2010).

Thus, these interviews are a flexible approach to collecting data that is assured to answer the research questions while at the same time staying true to the open-ended, exploratory spirit of qualitative inquiry. In order to ensure the

quality of the data collected by the interview, a researcher should follow the case study protocol (Yin, 2014). A copy of the case study protocol for this thesis is included as an appendix. The researcher prepared questions that covered the main aspects reflecting the research objective in the interviews.

Those questions were then applied to each case to ensure standardization and systematic comparison of data collection.

Qualitative research does not aim for specific sample sizes (Mason, 2010). The researcher interviewed the officials and diplomats from each of the mediating nations and each of the disputing nations who participated in the mediation cases. Although in-person interviewing is ideal, budget and security concerns dictated that some interviews had to be carried out over Skype or by phone.

The second source of data collection for this study was document collection. This entailed the compilation of documents relevant to the mediation attempts, such as declassified diplomatic communications between the relevant parties, transcripts of negotiations, etc. These were requested from the relevant parties in photocopy or digital form or found through public resources such as government websites. If these were not made available to the researcher, then what data can be obtained from publicly available sources were included, and the analysis was carried out with a clear caveat in the results noting which sources refused to provide which documentation.

When reviewing policy and process documents, a researcher should be aware that they do not always reflect reality accurately because they are out of date

(Yin, 2014). An example of those documents in this study is the declassified diplomatic communication between Yasir Arafat and the Norwegian government.

The third data source used for this research were archival records such as correspondence letters, diaries, minutes, annual reports, etc. It is claimed that archival records are more reliable than other documents because people use them for record-keeping (Baskarada, 2014). Collecting these data resulted in a careful review of scholarly literature and news coverage of the mediation attempts. These data served to add additional context and grounding for the study and offer an objective metric against which to measure the participants' perceptions.

Data Analysis

Data analysis was carried out in three stages. First, qualitative thematic analysis was carried out on each individual data set. Secondly, case study triangulation was applied within each case. Finally, a cross-case analysis was carried out across the four cases. Thematic analysis followed the six-step approach outlined by Clarke and Braun (2013). This process was applied to each data set and explained below.

First, the researcher carefully read and familiarised herself with the data to ensure a firm grounding. Secondly, the researcher assigned codes to the data. This was done using a codebook. Based on a careful review of the existing literature and the general topics it suggested would occur in a study of international mediation, the researcher developed an initial codebook before starting.

This initial codebook consisted of all the codes the researcher expected to encounter in the data but was expanded to add new codes as necessary. Codes represented unique ideas in the data, such as "motivation "or "leverage". Once codes had been assigned, the researcher moved on to the third stage, searching for themes in the data. An example of a theme might be "the state leverage in mediation". Once themes were found, the fourth stage consisted of exclusively comparing those themes against the data to ensure accurate reflection. Then, in the fifth stage, the researcher ensured all themes were unique and well defined. In the sixth and final stage, the researcher prepared the themes in a report that lists them and their significance, contextualizing the results in the problem's specifics and the general literature (Clarke and Braun, 2013).

Thematic data analysis is the preferred approach to data analysis in qualitative case studies. It provides a comprehensive way to compile the results of qualitative work. Unlike other qualitative methods such as phenomenology, a case study does not have strong philosophical underpinnings (Stake, 2013). Another competing analysis model is process tracing (Collier, 2011). Process tracing focuses on tracing out the causal "threads" in qualitative data through multiple data sources, and in this regard, it is similar to the triangulation approach as described below. However, one common problem the process tracing approach faces, similar to the issues described above concerning quantitative research, is that missing variables in the process tracing can misdirect the research (Collier, 2011). By contrast, the thematic analysis focuses primarily on exploring what information is in the data (Braun and Terry, 2013). Thus,

while it does not have the causal-inferential power of process tracing, it is superior for the not-yet-widely-researched field of small state mediation.

Once the data from each data set had been analysed through thematic analysis, the researcher applied case study triangulation (Yin, 2013). Triangulation represents the process of comparatively analysing the datasets within each case, noting their similarities and differences and reporting these. Places in which the data corroborate one another suggest firm conclusions. At the same time, those in which they differ are of interest due to potential differences in perceptions. Where the datasets disagree, the researcher would theorise potential reasons for this disagreement based on the data and the relevant literature.

For example, suppose the interviews in the current study had suggested that one of the two nations understudies, say Norway, had used a power-based mediation approach. However, the actual documents and transcripts on record suggested that its mediation approach was more like that of a pure mediator. In that case, the researcher might have theorised that this discrepancy resulted from the interviewed participants' desire to project their home country in a more positive light. On the other hand, if the interviews had suggested that Qatar used a leverage-based approach and the final written agreements include provisions for economic rewards to the disputants, then the two sources would be deemed to agree and strongly support that Qatar acted as a power-based mediator.

The final level of analysis was between cases (Yin, 2013). This type of analysis is similar to triangulation but with a greater emphasis on the differences and the reasons. Differences provide essential information if they can be linked to contextual factors. In the current study, understanding how the specific contextual factors in the four cases under study contributed to their differing outcomes provided meaningful insight into the factors determining the success of low-leverage state mediation in international conflicts (George and Bennett, 2005).

Reliability and Validity

Issues of reliability and validity are essential considerations in research to ensure that the effort is not wasted and the results are meaningful. In qualitative research, reliability refers to the credibility or the researcher's measures to ensure that she/he has represented reality accurately (Merriam and Tisdell, 2015). Because qualitative studies are so deeply grounded in the subjective, they will almost be replicable in the same way as quantitative research is. Therefore, this notion of reliability is a poor match. Instead, qualitative research assures reliability by ensuring that the findings reported accurately reflect the actual data (Cho and Trent, 2006).

Toward this end, the researcher took great care in every analysis step to ensure that the data supported all themes and conclusions drawn from themes.

This was proved through the liberal use of direct quotations from the data to support all claims made. For example, if a theme emerged indicating that low-leverage states acted as pure mediators, then this theme was supported by

multiple quotes from the data, allowing the researcher to convincingly illustrate that this theme did emerge from the results rather than being a misrepresented assumption.

Validity is a twofold problem; it must be considered in terms of internal and external validity. Internal validity is not applicable in this study because it is an exploratory case study rather than an explanatory case study. Internal validity is applied in explanatory case studies. An explanatory case study examines how and why the event "x "led to event "y" (Yin, 2014).

External validity deals with how well the results apply to a wider context. This type of validity is called transferability in qualitative research (Cho and Trent, 2006). Although qualitative research does not expressly seek to create transferable findings like quantitative research does (Merriam and Tisdell, 2015), the current research's contextual nature allows for a strong understanding of the similar circumstances to which the results apply. Because case studies, and especially multiple-case studies, seek to capture the underlying context of a phenomenon in addition to the phenomenon itself (Yin, 2013), they provide enough data on the relevant circumstances that a reader—such as a future researcher seeking to apply the results of the study—can reasonably determine whether the two situations are similar enough that the findings can be applied. Additionally, this strong depiction of context will allow future researchers to determine where the findings do not apply and accordingly choose different contexts to explore in future studies. External validity is relevant to this study despite its qualitative nature because the topic has not been given much prior attention in the literature (Yin, 2013). Accordingly, the findings of this thesis are

reported in such a way as to ensure that future researchers can determine how and when they apply when employing or expanding upon the theory developed herein.

In sum, in this thesis, the data were collected following case study protocol, using multiple sources of evidence from at least three sources: semi-structured interviews, document collection and finally, archival records such as correspondence, letters, diaries, minutes and annual reports. The interviews took place with officials and diplomats representing the mediators and the disputants. Semi-structured interviews were the primary source of data. The researcher prepared an interview schedule for with questions that covered the main aspects reflecting the research aims and research questions. Those questions were applied to each case to ensure standardization and systematic comparison of data collection. Data analysis was carried out in three stages: a qualitative thematic analysis was carried out on each case; secondly, case study triangulation was applied within each case; and finally, a cross-case analysis was carried out across the four cases.

CHAPTER TWO: LITERATURE REVIEW AND CONCEPTUAL FRAMEWORK

Literature Review

According to the research questions, the literature review section covers the following subjects: Definitions of international mediation and low-leverage states; secondly, the motivation of the states; thirdly, the behaviour of the states which are engaged in international mediation —whether power mediators or pure mediators; fourthly, the strategies the states usually follow in international mediation —facilitation, formulation, or manipulation strategies; and finally, the necessity of the state to be an impartial or biased mediator.

Definitions of international mediation and low-leverage states

In understanding the role of low-leverage states in international mediation, it is first essential to understand the terms low-leverage states, small states, and international mediation.

There is no agreement among the scholars on the definition of "international mediation" (Bingham, 1986; Kleiboer, 1996; Moore, 2014; L. Singer, 2018; Sisk,1996). There is also no agreement on the definition of "small states" (Archer & Nugent, 2002; Hey, 2004; Maass, 2009). Different scholars define international mediation in different ways. This thesis focuses on the following scholars' definitions of international mediation because those scholars cover different aspects, characteristics and types of international mediation. Some scholars, such as Moore, focus on mediation as a voluntary process. They focus on the voluntary aspect of the mediation process where the disputants voluntarily ask for mediation. For example, Moore (2014) states that mediation is

the process of voluntarily interceding between two parties who cannot resolve their differences on their own.

Other scholars take a different approach and describe mediation as a coercive process. One example of those scholars is Sisk (1996), who defines mediation as either a mediator wanting to stop the violence at all costs or a multitude of would-be mediators pulling at the situation in different or even opposite directions (Sisk, 1996). Other scholars choose to focus on neutrality in the mediation process, such as Bingham (1986).

Some scholars introduce a comprehensive definition of international mediation. An example of those scholars is Bercovitch (1992, p.7), who defines mediation as "related to but distinct from the parties' efforts, where the disputing parties or their representatives seek the assistance, or accept an offer of help, from an individual, group, state or organization to change, affect, or influence their perceptions or behaviour, without resorting to physical force or invoking the authority of law" (Bercovitch,1992, p.7).

This definition is broad and therefore allows for incorporating many other approaches. However, comprehension is not needed to define mediation in this thesis because, in this case, the mediator is a low-leverage state that does not offer mediation but waits for an invitation from the disputants to intermediate (Bercovitch, 2011b).

This thesis also excludes Sisk's (1996) definition, which describes the mediator as a power mediator who uses a manipulative strategy, because this does not match the focus of this thesis, in which the mediators are low-leverage

states who do not have the same level of leverage as a powerful mediator.

Based on this discussion and for this research, which focuses on low-leverage state mediation, this thesis has developed the following definition:

Based on the purpose of this thesis which focuses on low-leverage states as mediators, international mediation is defined as an instrument of conflict resolution where the needy disputants seek the assistance of low-leverage states voluntarily to be a mediator. The low leverage state plays a role of an impartial, mutually accepted mediator to reach a settlement. Due to its lower leverage, the low leverage state is supposed to behave as a pure mediator who cannot manipulate the disputants.

Definition low-leverage states

In international relations, there is no commonly accepted way to distinguish small states from large states among scholars (Archer & Nugent, 2002; Hanf & Soetendorp, 2014; Hey, 2004; Lee & Smith, 2008; Magnette & Nicolaïdis, 2005; Neumann, 2004; Thorhallsson, 2017; Thorhallsson & Wivel, 2006).

Based on their economic power, states could be classified into large and small states. This classification matches the negotiations on distributional effects (Braveboy-Wagner, 2003; Hey, 2004; Katzenstein, 1985; Steinberg, 2002). In the negotiations on armed conflict or security issues, the states may be classified into small states or major powers. This classification is based on the military capacity of the states (Keohane, 1969; Rickli, 2008; Rothstein, 1968), or population (Baillie, 1998).

Indeed, the scholarship offers two approaches to defining small states: the quantitative and the qualitative approaches (Archer & Nugent, 2002; Lee & Smith, 2008). The quantitative approach classifies the states according to absolute criteria such as population, geographical size, gross domestic product (GDP), and military power (Archer & Nugent, 2002). The qualitative approach instead examines descriptive factors such as the state's relations to its environment and the degree to which the state can influence international affairs (Hey,2003, 2004; Keohane,1969), as well as by the power exercised in its external environment (Wivel & Mouritzen, 2004).

Considering only the quantitative definition is problematic. For example, one natural quantitative measure might be a country's GDP because this is a quantitative measure that incorporates an easy measure of size based on wealth. For example, Qatar's GDP is estimated at USD 404.109 billion in terms of purchasing power parity, making it the 47th highest GDP in the world and, at USD 145,894 per capita GDP, the highest per capita in the world (IMF, 2017). However, in terms of actual geographic size, another obvious quantitative measure of state size, Qatar is a mere 11,581 square kilometres, making it 159th out of 199 countries in terms of geographic size (IMF, 2017). Therefore, with two such starkly opposed quantitative size measures, is Qatar a small state or a large state?

However, in international mediation, the actual size of the state is not the appropriate criterion to determine the weight of the state as a mediator, nor is it the proper criterion to distinguish between states as mediators. Moreover, using the quantitative or the qualitative approach to determine the performance

of a state in international mediation may not be appropriate. In order to determine the smallness or largeness of a state in international mediation, Panke (2012) helps resolve this issue by suggesting that one consider size as a relational context. To determine whether a state is small or big in any field, one should characterise one actor relative to another in a specific context (Panke, 2012).

Based on the context of international mediation, states could be classified according to each state's leverage. The leverage is the ability of the mediator to move one or both of the disputants in an intended direction and can be utilised to force them into accepting the proposed agreement (Zartman & Touval, 1996). In other words, low-leverage states in international mediation are the states who do have leverage. However, they have less influence on the disputants than the power mediators. Thus, there is a positive relationship between the state's leverage and its influence or power. The more leverage the state has the more it is able to move one or both of the disputants in an intended direction. Low-leverage states have more difficulties than power states. In this sense, low-leverage states face more difficulties in using bargaining strategies such as concession or threat. They are also supposed to be limited in affecting the mediation process or outcomes.

Low-leverage states in international mediation are the states that have limited leverage compared to the superpower mediators. The superpowers are privileged as major actors with access to unlimited economic resources and hard power capabilities and strategies (Stokke, 2012). In contrast, the low-leverage states do not have the ability to force the disputants to reach an agreement

because they usually lack the stick. In other words, they do not the capacity to exert pressure on the disputants to sign an agreement. However, they have the carrots represented in their economic resources. They may use their economic resources as incentives to encourage the disputants to reach an agreement. In other words, they can reward the disputants to encourage them to reach an agreement but cannot punish or threaten the disputants to force them to an agreement. In addition, they mostly rely on trust-based leverage, which is represented in historical and cultural ties with the disputants. Moreover, in their mediation, they usually need the backing of an international key factor as they have limited influence in the international arena.

Low-leverage states do have economic resources -their motivations and the way they use their economic resources in the mediation process determines whether the low-leverage states are norm-based mediators or interest-based mediators.

They have a limited supply of resources to encourage or discourage certain behaviours and move parties towards an agreement (Beardsley, 2009). Low-leverage states are limited in their leverage, which is often an essential tool in international mediation (Zartman & Touval, 1996). However, this does not necessarily mean that they are ineffective international mediators.

There is a diverse set of approaches to classifying the mediator's sources of leverage. For example, Zartman and Touval (1996) classify the mediator's sources of leverage into five categories: persuasion, extraction, termination, deprivation, and gratification (Zartman and Touval, 1996). Where, (1)

persuasion is the ability of the mediator to convince the disputants that the resolution is necessary. Persuasion refers to a form of facilitation in mediation. (2) Extraction is the ability of the mediator to extract an attractive proposal to present to each disputant. (3) Termination is the ability of the mediator to withdraw from the mediation process or the ability of the mediator to walk away. This source of power depends on the parties' desire for the mediator to engage. The leverage or power of this resource would be undermined if the mediator were to want the mediation process to continue more than the disputants. (4) Deprivation is the ability of the mediator to refrain from granting resources to the disputants. Deprivation typically involves the shifting of resources from one party to the other. (5) Gratification is the ability of the mediator to provide incentives to the disputants to encourage them to reach and accept the agreement (Zartman, Touval, Crocker, Hampson, & Aall, 2007; Zartman & Touval, 1996).

For persuasion, Zartman and Touval (1996) assert that the need of the disputant to reach conflict resolution is considered leverage for the mediator. It is considered leverage because it strengthens the mediator's position (Zartman & Touval, 1996). The ability of the mediator to persuade the disputants would increase when the disputants badly need a solution. The need of the disputants is increased by changes in local, regional and international circumstances. For example, the PLO requested Norwegian mediation after the Iraqi crisis because the PLO had financial and political problems because of its supporting Iraq in Kuwait's invasion. Moreover, by that time, the world had already witnessed the collapse of the Soviet Union, thus producing a unipolar

arena with the USA as the only remaining superpower or hegemon country. The change in the regional and international context weakened the PLO's position and left it more in need of peace with Israel (Lieberfeld, 2008). In the same fashion, the leverage resources could be negative sanctions or positive incentives (Kleiboer, 1996; Touval & Zartman, 1985). Leverage resources could also be material, such as economic aid, or immaterial, such as psychological pressure (Siniver, 2006).

Differently, Reid suggests that leverage consists of two different fundamental components: capability and credibility. By capability leverage, Reid means the extent to which the mediator uses material strength (economic resources) to coerce a conflict settlement. While capability leverage is reliant on coercion, credibility leverage does not. Credibility leverage relies on contextual knowledge, information, historical relations with the disputants, and cultural relations of the disputants (Reid, 2017). In the same vein, Princen (2014) considers information and interest as intangible forms of leverage. Additionally, Beardsley (2008) classifies leverage into tangible and intangible leverage. Where tangible leverage refers to material resources, intangible leverage refers to the ability of the mediator to use the prestige and make promises that will be believed (Beardsley, 2008).

Based on the above discussion, this thesis classifies leverage into two broad groups: trust-based leverage and power-based leverage.

Trust-based leverage is not a coercive tool in mediation. Under the umbrella of trust-based leverage, the credibility of Raid's can be included (Reid, 2017).

The persuasion and extraction categories of Zartman and Touval are considered tools of trust-based leverage (Zartman and Touval, 1996). Additionally, the intangible leverage of Beardsley is considered a type of trust-based leverage (Beardsley, 2008). Trust-based leverage is reliant on the previous relationship of the mediator with the disputants (Bercovitch & Houston, 2000), that is to say, the relationship or alliance between the mediator and the parties, such as bonds, shared history and values and common interest.

These relationships help in establishing familiarity, understanding, trust and acceptability of the mediator (Bercovitch & Houston, 2000; Carnevale & Pegnetter, 1985). This relationship is often represented as historical ties, cultural ties, and personal ties. Historical ties could be a colonial legacy, a previous mediation attempt, or humanitarian aid. Cultural ties could consist of ethnic ties or a shared religious identity. Meanwhile, personal ties are often a byproduct of NGO networking (through humanitarian aid and research) or could refer to high-profile personal ties in personal diplomacy, such as the case of Qatar.

Power-based leverage is represented in usage any pressure or coercion on the disputants to reach an agreement. Under this type of leverage, the three categories of Zartman and Touval, termination, deprivation, and gratification, are included (Zartman and Touval, 1996). The exploitation of the need of the disputants is considered power-based leverage (Zartman and Touval, 1996). Siniver's (2006) classification of leverage, material (economic aid) or immaterial (psychological pressure), is considered power-based leverage. The capability leverage of Reid (2017) is included in power-based leverage. The

negative and positive sanctions (Kleibor,1996) are considered power-based leverage.

In general, the power-based leverage is represented in three types of might: the economic might of the state, its military power and the international political influence. Through the power-based leverage, the mediator can use deprivation and gratification as tools in mediation, in other words, the "stick-and-carrot" policy. In order to measure leverage, these two types of leverage should be operationalised. Power leverage is operationalised in terms of the mediator's economic power, military power, and international influence. The economic power could be measured by using the state's gross domestic product (GDP) as an indicator (Reid, 2017). Trust-based leverage is a function of historical ties, cultural and personal ties.

Based on the type of leverage, the states in the context of international mediation are divided into high-leverage states or power states and low-leverage states. High-leverage states have a high level of economic, military, and political power and do not need to be backed by an international key factor, and they use deprivation and gratification as tools in their mediation.

Compared to high-leverage or power states, low-leverage states have less economic, diplomatic and military leverage to help facilitate the mediation process. Their mediation usually relies on trust-based mediation and the disputants' need for mediation. Finally, low-leverage states use enticement tools and not threat tools in the mediation process. In other words, the low-leverage states lack the stick. However, they may own limited carrots.

Qatar and Norway are examples of low-leverage states. Norway and Qatar have power-based leverage, represented in their economic power, limited international influence, and military might. However, this power-based leverage is not enough to behave as a power mediator such as the USA, for example. As low-leverage states, both Norway and Qatar are supposed to rely on trustbased leverage. In Norway's mediation of the Oslo Accord 1993, and the CPA (2005), Norway exhibited trust-based leverage. The backbone of this trustbased leverage is Norwegian NGOs and Norwegian Humanitarian aid. Those NGOs played an essential role in the Norwegian historical relation between Norway and Palestinians, Israelis, and the Sudanese. Likewise, Qatar's trustbased leverage can be seen in its mediation between Fatah and Hamas and in the Sudanese conflict. Qatar shares a cultural tie with both the Palestinians and the Sudanese. Those cultural ties are based on the religion of Islam, and a shared ethnic group; Palestinians, Sudanese and the Qatari are all Arabs. However, Qatari finance plays a different role in the Qatari International mediation. However, not every Arab country can be a mediator because sharing cultural and religious ties is not enough to be a mediator. The credibility of the mediator is important, the historical experience of the mediator (Egypt), the neutrality of the country (Oman), being a regional hegemon (Saudi Arabia) or strong finance coupled with being under threat (Qatar). However, the religious and cultural ties make the mediator role easier.

In brief, the type of leverage, power-based leverage or trust-based leverage, the mediator used is reflected in the type of mediation strategy the mediator utilizes in the mediation process.

In conclusion, low-leverage states are the states which are small in terms of some aspects such as the size of population or the area and have disproportionate strengths in other aspects such as economic resources. Those disproportionate strengths are considered leverage for those states but not the same leverage level of the power states. By shaping their standing in international relations, they determine their size.

Facilitation, formulation, and manipulation strategies

A mediation strategy is an approach or plan that the mediator uses to resolve a conflict (Bercovitch et al., 1991). International mediation scholars such as Bercovitch, Zartman, Touval, and others (Beardsley *et al.* (2006); Bercovitch (1992) & (2002); Bercovitch and Gartner (2009); Saadia Touval and I. William Zartman (1985); Wilkenfeld *et al.* (2003), classify the mediation strategy into three types: facilitation, formulation and manipulation strategy. This classification is based on the mediator's behaviour in the mediation process. The mediator's behaviour in the mediation process depends upon the resources available to them within the context of the dispute (Bercovitch & Houston, 2000). In other words, the amount and type of leverage a mediator possesses are often reflected in the type of strategy the mediator employs in the mediation process. Mediation strategies can be classified according to the degree of mediator intervention in the mediation process, and this intervention can range from passive to active. Additionally, the classification of mediation strategies is based on the degree of coercion force the mediator practices on the disputants.

Consequently, the mediation strategies range from the least to the most coercive. Thus, the mediation literature has focused on three main styles of mediation strategies: facilitation strategies, formulation strategies, and manipulation strategies (Beardsley et al., 2006; Bercovitch, 1997; Carnevale, 1986; Crocker, Hampson, & Aall, 2007; Quinn, Wilkenfeld, Eralp, Asal, & Mclauchlin, 2013; Touval & Zartman, 1985; Wilkenfeld, Young, Quinn, & Asal, 2007; Zartman, 2007).

Facilitation strategies represent a low level of intervention in which the mediator plays a passive role, akin to a communication channel, and has little control over the mediation process (Bercovitch, 1997, 2011b; Hopmann, 1996; Touval & Zartman, 1985; Zartman & Touval, 1996). The mediator ensures continued discussion between the disputants (Keashly & Fisher, 1996). By contrast, formulation strategies represent a more active mediator role. A mediator using such a strategy has more formal control over the mediation process and can make decisions about issues such as the mediation environment, the meeting type and frequency, the meeting agenda, and the distribution of information and resources (Bercovitch et al., 1991; Bercovitch, 2011). A mediator who acts as a formulator can propose a new solution or alternative to the disputants (Beardsley et al., 2006; Bercovitch, 2011b; Hopmann, 1996; Touval & Zartman, 1985; Zartman & Touval, 1996). Formulators try to convince the disputants that unilateral solutions are less beneficial than negotiation (Zartman & Touval, 2001).

However, the most active mediator and the most coercive mediator is the mediator who uses manipulation strategies. When using these strategies, the

mediator affects the content and the substance of the mediation. The mediator is able to exert different forms of pressure on the disputants, such as sticks and carrots (rewards and punishments). The mediator can even issue new proposals and guide the disputants towards signing an agreement (Beardsley et al., 2006; Bercovitch, 2011b; Bercovitch et al., 1991; Foster, 2011; Kleiboer, 1996). The mediator using the manipulation strategy takes advantage of leverage to influence the mediation process (Touval & Zartman, 1985). Manipulative mediators may offer carrots (compensation) to disputants to incentivise peace. The mediator presents a more attractive option to the disputants by adding benefits or carrots to the proposed solutions (Carnevale, 1986; Zartman & Touval, 1996). Those carrots may include direct compensation, favourable economic policies to the disputants, or diplomatic concessions (Beardsley et al., 2006). Manipulative mediators may use sticks (pressure) to increase the cost of non-agreement. Those sticks may come in several forms, such as economic sanctions, diplomatic sanctions, or even the threat of direct military intervention (Carnevale, 1986).

Using sticks is contradictory to the definitions of mediation for the scholars who consider mediation a voluntary process for the disputants (Bercovitch, 2011b; Bingham, 1986; Kleiboer, 1996; Moore, 2014; L. Singer, 2018; Sisk, 1996). Skjelsbael (1991) considers this to be coercive mediation. The mediator's usage of rewards and punishment is far removed from the true essence of mediation, which promotes persuasion as opposed to coercion (Skjelsbaek, 1991).

Scholarly opinion is divided as to which of these types of strategies is most effective (Bercovitch et al., 1991; Kochan & Jick, 1978; A. Smith & Stam, 2003). For the manipulative mediator who has the leverage to extend rewards and punishments, the mediator can employ the facilitative strategy in order to bring the disputants together and employ the formulation strategy to formulate a proposal. Mediators who lack resources or leverage employ facilitation and formulation strategies (Beardsley et al., 2006).

To reconcile the scholars' divisions, Beardsley et al. (2006) link the crisis outcomes and the mediation strategy employed in the mediation process. They find that facilitative mediation efficiently resolves commitment problems and reduces tension in post-crisis. Mediators focus more on the facilitation strategy to secure a more lasting resolution for the dispute. The most durable agreements are achieved by less intrusion by an external third party. Manipulative mediation is efficient in the case of formal agreements because the mediator uses the carrots and sticks policy (Beardsley et al., 2006).

In sum, there is a linkage between the leverage and the strategy the mediator uses in the mediation process. The manipulative mediator who has leverage is able to create rewards and enforce punishments. This mediator can use the facilitative strategy to bring the disputants together and employ the formulation strategy to formulate a proposal. Meanwhile, the mediator who lacks leverage employs facilitation and formative strategies. Considering the low-leverage state in international mediation and its definition, which strategy does the low-leverage state use as a tool in the mediation process? Is it facilitation,

formulation, or manipulation? This research seeks to fill this gap by answering those questions.

Mediator motivation and the mediation interest

This section addresses the "why" question: "Why does the mediator engage differently in international mediation?" In response, the mediator's motivation or interest in international mediation is now discussed. Being an interest-based mediator or norm-based mediator is a debated issue among scholars. Scholars such as Zartman and Touval strongly believe that mediators have an interest in mediation. Other scholars such as Björkdahl (2013) and Bandarage (2011) believe in norm-based mediation and assert that international mediation is guided by idealism and normative conceptions.

According to Zartman, the mediators do have an interest in the conflict's outcomes; otherwise, they would not willingly engage in mediation. He suggests that this applies to all types of mediators: superpowers, international organizations, and small states or small powers (Zartman & Touval, 1996). Furthermore, he explains that mediators, in general, are involved in mediation because they have their own interests. Consequently, mediators are engaged in mediation for different reasons, sometimes to avoid conflict, which could jeopardise their relationship with the disputants. Other times, states are involved in mediation to maintain their reputation as effective mediators, which is self-interested (Zartman, 2009). Touval (1982) agrees with Zartman when asserting that mediators are motivated by self-interest (Touval, 1982).

Important to note that Touval and Zartman further distinguish between the two types of self-interest motivation: defensive and expansionist motives (Touval, 1992; Touval & Zartman, 1985). On the one hand, the defensive motive occurs when the conflict between two disputants threatens the mediator's interest, such as when the conflict between two states disturbs the regional balance. On the other hand, expansionist motives occur when the mediator aims to increase their resources and power (Spiegel, 1985). By the same token, low-leverage states are also motivated by self-interest. Such interests include the mediator's security and prestige or international standing (Zartman & Touval, 1996). When Zartman and Touval view self-interest as motivation in international mediation, they build their claim on the rational-actor approach, using cost-benefit considerations. Overall, there seems to be some evidence to indicate that actors make decisions based on belief, ideology, and culture (Chai, 2001). Additionally, it is essential to represent the social constructivists' view of international politics, which asserts that international politics is socially built based on social norms and values. (Fierke, 2015). This explains the existence of value-driven interest, or norm-driven motivation, which is the driving factor behind the international mediation of some low-leverage states.

Norms identify and establish the appropriate behaviour for an actor (Björkdahl, 2013). The aforementioned states' international mediation is guided by idealism and normative conceptions (Bandarage, 2011). The motivation of those states who adopt normative-driven mediation is to help the poor around the world, enhance gender equality, promote international peace (Björkdahl, 2013), and make the world a better place (Leira, 2013). States with norm-

driven motivation adopt universal norms and work for the common good of the international community (Björkdahl, 2013; Ingebritsen, 2006). They act as norm agents and share specific characteristics such as domestic democracy, a strong human rights record, provision of development assistance, and promotion of human rights and international peace (Brysk, 2009). They play the role of norm entrepreneurs by constructing and advocating new international norms. Norm entrepreneurship is a diplomatic strategy capable of influencing the world (Björkdahl, 2013).

In like manner, some scholars argue that, in their mediation, low-leverage states are motivated either by self-interest or by norm-driven motivation. Interests are the objectives and the outcomes that a state seeks to achieve or the state's preferences among the possible range of alternative outcomes (Jesse & Dreyer, 2016; Jesse & Williams, 2010). Indeed, Beardsley explains that interest could be public or private. The private interest is represented as a reduction of the spill-over effects which directly affect the mediator, while the public interest is represented by the stability of the international system (Beardsley, 2011).

As can be seen and based on the argument of Beardsley (2011), when the mediator's interest is public and represented by the stability of the international system, it means that the mediation is a norm-driven mediation. A private mediator has self-interested motivation. Notably, mediators are engaged in a mediation process when their private interests are at risk (Greig & Regan, 2008). Obviously, mediators who lack interest in the conflict do not exist, which is not in line with the findings of (Svensson, 2007b), who states that a power

mediator has an interest in the conflict while a pure mediator does not. Duursam (2014) reconciles this gap by distinguishing between two kinds of interest: self-driven interest and value-driven interest.

In sum, the motivations of low-leverage states, in their international mediation, are driven by two types: interest-driven motivation and norms-driven motivation. However, as shown above, it is essential to consider distinguishing between those types of mediation as a spectrum, not a dichotomy division.

Power and pure mediation

While scholars focus on the power mediator who possesses considerable leverage and usually has an interest in the mediation outcome, they ignore the mediators who possess limited leverage and are represented by low-leverage states. A power mediator is a mediator who uses its resources to push the disputants towards its preferred direction in order to ensure its interest when the mediation outcome is reached (Harris & Reilly, 1998; Kleiboer, 2002; Svensson, 2007b; Zartman & Touval, 1996). Meanwhile, the pure mediator usually lacks resources to be used in international mediation and has no specific interest in the conflict (Svensson, 2007b). The pure mediator uses persuasion to facilitate negotiation between the disputants (Fisher & Keashly, 1991) and builds social ties between the disputants while enhancing a communication channel (Svensson, 2007b).

Touval (1982) considers one type of mediator, the power mediator, who makes suggestions to the disputants while seeking to influence them by

offering them incentives and exerting pressure. The mediator needs resources or leverage to offer incentives or exert pressure on the disputants (Smith, 1994).

Leverage or power is the criterion used to distinguish between pure and power mediators. Leverage is the ability of the mediator to put pressure on one or both of the disputants to make them accept the proposed agreement (Fisher, 2016; Kleiboer, 1996; Svensson, 2007b; Zartman & Touval, 2001; Zartman & Touval, 1996).

Scholars debate the effect of leverage on mediation success. Some scholars believe in the necessity of leverage for mediation success (Bercovitch, Anagnoson, & Wille, 1991; Kleiboer, 1996; Smith, 1994; Smith, 1985; Touval, 1992). Other scholars claim that mediators who lack leverage or power may still facilitate mediation success (Slim,1992; Yarrow, 1978). Slim (1992), for example, argues that small states are acceptable mediators because they are not threatening to others. She argues that for the powerful party, a small state can provide a face-saver without threatening the general bargaining position of the powerful party. A smaller state can appear more sympathetic for the weaker party as it understands what it means to negotiate from a weak point (Slim, 1992). One way to reconcile these differences is that power or leverage may be necessary for the powerful mediator, but the impartiality of the less powerful mediator is the main reason he or she is acceptable to disputants (Siniver, 2006).

Although international mediation scholars have debated the impact of leverage, they agree that too much leverage is a real risk when less would suffice. Not only does the burden on the third-party increase, which often follows threats and promises, but too much leverage can prove detrimental to the long-term stability of peace. When third parties are overburdened in their approach, they create artificial incentives for agreements that are unlikely to be sustainable over time (Beardsley, 2008, 2011; Beardsley & Lo, 2014; Böhmelt, 2011; Crocker, Hampson, &Aall, 2001a and 2001b; Gurses, Rost, & McLeod, 2008; Walter, 2002; Werner & Yuen, 2005). Thus, the mediator's promise of compensation for the disputants' concessions cannot be controlled (Kleiboer, 1996). Additionally, the disputants may become dependent on the mediator for future incentives in the long run.

It is probable that mediated settlements that arise from extreme usage of leverage will not last very long because the agreement was built on the mediator's compliance and not on the changing perceptions and attitudes of the disputants (Kelman,1958). In other words, mediators can have a significant impact when they are able to use leverage to stop persistent violence in the short term, especially when they can maintain this leverage over time. Furthermore, mediators can increase their effectiveness by using lighter tactics to help the disputants get over the main barriers to permanent settlements. However, the mediator may sacrifice long-term goals for short-term success (Beardsley, 2013).

Examining the effect of the mediator types on mediation outcomes, mediation scholars disagree on which type of mediator is the most effective peacemaker

(Beardsley et. al. 2006; Carnevale, 2002; Kleiboer, 1996 and 2002). For example, Kleiboer (2002) suggests that a powerful mediator is more effective in conflict settlement than conflict resolution. Conflict settlement implies termination of the conflict without dealing with the underlying causes of the conflict. If the underlying causes still exist, the conflict may arise again. Conflict resolution, instead, addresses the underlying roots of the conflicts, which means that the dispute will not arise again (Kleiboer, 2002). In contrast to Kleiboer, Beardsley et al. (2006) claim that less intrusive tactics in mediation may lead to a more durable peace than more intrusive tactics, such as those used by a powerful mediator.

Svensson (2007) reconciles this gap by suggesting that the power mediator and pure mediator are complementary and not contradictory. The pure mediator and power mediator have different qualities, so they make different contributions to the mediation and the mediation outcomes. Whereas the pure mediator can offer a safe environment for the disputants to address their conflict's central and sensitive issues, the power mediator can use the carrot (rewards) and sticks (punishments) to get the disputants to reach an agreement. In conclusion, combining the pure and the power mediator should enhance the mediation's success (Svensson, 2007a).

However, dividing mediators into power mediators and pure mediators based on leverage may not be accurate. While scholars focus on the power mediator who possesses considerable leverage and the pure mediator who lacks it, they ignore the mediators who possess limited leverage and are represented by low-leverage states. Based on the definition of low-leverage states and the

discussion above, the pure mediator usually lacks leverage to be used in international mediation and has no interest in the conflict (Svensson, 2007b). Additionally, the pure mediator uses persuasion to facilitate negotiation between the disputants (Fisher & Keashly,1991) and builds social ties between the disputants while enhancing a communication channel between them (Svensson, 2007b). While the power mediator makes suggestions to the disputants while seeking to influence them by offering them incentives and exerting pressure by using their leverage (Smith, 1994; Touval, 1982). The mediator needs resources or leverage to offer incentives or exert pressure on the disputants (Smith, 1994). In the end, it is worth mentioning that acting as a power mediator or a pure mediator is connected to the strategy the mediator uses as a tool in the mediation process.

Impartiality and bias in mediation

In international mediation, the mediator could be impartial or biased towards one of the disputants. Impartiality is a question of the perception of the disputants (Touval, 1975b). Impartiality requires the mediator to remain unbiased towards all parties in the dispute and towards the interest and options for settlement (Gerami, 2009). For mediators to be impartial, they must deal with all parties in the same manner, both in terms of procedure and substance (Carnevale & Arad, 1996).

The advantages of impartial mediators or biased mediators have often been debated. Some see impartiality as essential to mediation success (Acland, 1993; Beber, 2012; Foster, 2011; Skjelsbaek, 1991; Young, 2015). From this

perspective, impartiality is the basis for building personal trust and confidence in the mediator, which, in turn, leads to the acceptability of the mediator, which many argue is key for mediation success (Kleiboer, 1996).

However, there is a contrasting belief that mediator impartiality is not needed for mediation success. Those who take this view assert that the mediator's influence and willingness to protect the parties' interests are the main reasons behind the acceptance of the mediator, not impartiality. Conversely, the mediator's bias towards one of the adversaries is the reason behind the acceptability of the mediator by the disputants. From this perspective, it is the mediator's leverage and not impartiality that leads to mediation success (Bercovitch et al., 1991; Kleiboer, 1996, 1998; Siniver, 2006; Smith, 1985; Touval, 1975b, 1982; Touval & Zartman, 1985).

To bridge this gap, a third position recommends distinguishing between two types of mediators: low power or low stakes mediators (pure mediators) and high power or high stakes mediators, power mediators (Smith, 1994). Smith (1994) asserts that the debate over impartiality is a pointless chimera. Both sides use valid arguments to defend their point of view, and each side is based on its definition of mediation, whether it be pure mediation or power mediation. Both power and pure mediators are effective under certain situations, and both face specific obstacles. Therefore, the question of impartiality represents a non-debate. (Smith, 1994).

The pure mediator and the power mediator operate in an entirely different world from each other. The necessity of impartiality depends on whether the

mediator is a pure or power mediator. For the power mediator, impartiality is not necessary. In power mediation, the disputants have no choice but to accept the mediators who are equipped with the resources to induce an agreement, so in this case, Bercovitch and Touval and their team are correct. However, impartiality is crucial for pure mediators who rely on impartiality to build trust between themselves and the disputants. They do not need to boast about the power mediators' resources (Smith, 1994).

Some scholars such as Kleiboer (1996), Assefa (1987), Princen (2014), and Slim (1992) agree with Smith (1994) as they assert that there is a relationship between leverage and the (im)partiality of the mediator. Power mediators do not necessarily need to be impartial to be successful mediators because they compensate for this with leverage. On the other hand, the pure mediator needs impartiality to be accepted by the disputants (Assefa, 1987; Princen, 2014; Slim, 1992).

By linking whether the mediator is impartial or biased with leverage, the literature assesses that the power mediator does not necessarily need to be impartial, and the pure mediator should be impartial in order to be accepted by the disputants. Given the limited leverage that the low-leverage states have, this assessment left a gap in the literature by not answering the question: Should low-leverage states be impartial, or is this not always necessary? This research seeks to bridge this gap.

In conclusion, the literature distinguishes two types of mediators based on leverage: powerful mediators and pure mediators. While the powerful mediator

has leverage, the pure mediator lacks leverage. The sources of leverage may be classified into five sources: persuasion, extraction, termination, deprivation, and gratification. In like manner, the mediation strategies could be classified into three strategies: facilitation, formulation, and manipulation.

By linking the mediator's (im)partiality with leverage, the power mediator does not necessarily need to be impartial, and the pure mediator should be impartial in order to be accepted by the disputants. Reviewing the literature, it is concluded that leverage is the criterion to determine whether a mediator has power or is pure in nature. While the pure mediator usually lacks leverage to be used in international mediation and has no interest in the conflict, the power mediator has leverage and uses it to influence the disputants towards a pacific agreement by offering them incentives and exerting pressure by using their leverage. Being a power or pure mediator reflects on the mediation strategy and impartiality as well. While the power mediator tends to employ a manipulative strategy in the mediation process, the pure mediator employs facilitation or formulation strategy due to their lack of leverage. Moreover, the power mediator does not necessarily need to be impartial mediator. However, impartiality is an essential criterion for the pure mediator to be accepted by the disputants.

In its international mediation engagement, low-leverage states do not fit neatly into the categories of "pure" or "power" mediators because of their unique status as states that often have an interest in the conflict and yet have limited leverage compared to the leverage possessed by many larger states.

Conceptual framework

The conceptual framework consists of two sections. The first section discusses the relationship between power, leverage, and strategy. The second section discusses the difference between norm-based mediation and interest-based mediation and the expected behaviour of the two types of low-leverage states mediators, Qatar and Norway.

Power, leverage, and strategy in international mediation

Power, leverage, and strategy are three components in international mediation which are connected. As mentioned in the literature review, there are three types of strategies used in international mediation: facilitation, formulation, and manipulation strategy (Beardsley et al., 2006; Bercovitch and Rubin, 1994; Touval and Zartman, 1985),

The three strategies are classified based on the degree of the mediator intervention in the negotiations. In the facilitation strategy, the mediator plays a role of a communication host for the two parties, with no substantive contribution to the negotiations. The mediator usually does not attend the negotiations. The second type of strategy is the formulation strategy. In this strategy, the mediator is more involved in the mediation. The formulator can contribute to the negotiations by introducing substantive suggestions or proposals when appropriate or upon the partier's request. The third type of mediation strategy is the manipulation strategy which is the coercive strategy. In this strategy, the mediator uses his power or leverage for guiding the negotiations. The manipulator has a range of tools to manipulate the disputants. In the manipulation strategy,

the mediator can pressure the disputants, promise resources, and even threaten to withdraw from the process. The mediator can reward or punish the disputants. The mediator can offer an incentive to encourage the disputants to reach an agreement. The mediator can practice psychological pressure on the disputants to explain to them the cost of a non-agreement would be (Eriksson, 2019). The choice of strategy is based on how much leverage, the capabilities, and sources of power the mediator has (Bercovitch, 1992). The source of power is represented in the military, economic, and political resources. Those resources are necessary for the manipulator to generate leverage to affect the disputants in the mediation process and the mediation outcomes (Kleiboer, 1996; Heemsbergen and Siniver, 2011). The mediator's type of strategy in the mediation process determines whether the mediator is a pure mediator (Aggestam, 2002) or a power mediator (Aggestam, 2002) or a power mediator (Bercovitch and Houston, 2000).

Norm-based mediation and interest-based mediation

Several components of the international mediation process contribute to determining whether the type of mediation is either norm-based mediation or interest-based mediation. These include motivation or interest, usage of resources, the mediation strategy, the mediator's impartiality and finally, whether the mediator can be classified as either a pure or power mediator.

The motivation of the low-leverage state is the critical factor determining which type of mediation is practised; norm-based mediation or interest-based mediation. Before discussing the motivation or interest of both norm-based

mediators and interest-based mediators, it is necessary to mention that states usually engage themselves in international mediation on the basis of interest; otherwise, they will not be involved in mediation (Zartman & Touval, 1996). This is applied to norm-based mediators such as Norway and Sweden. Such states engaged themselves in international mediation to promote their international standing and influence in international politics (Stokke, 2012).

When norm-based states utilize international mediation as a foreign policy tool, they gear towards survival by seeking status in the international community. However, they adhere to norms as an essential part of their international mediation and balance its interest and the norms. Stokke (2012) argues that Norway involves itself in international mediation to secure itself by increasing its influence in international relations and strengthening its standing in the international community (Stokke, 2012). As a low-leverage state located in Northern Europe, Norway makes it vulnerable. The vulnerability of Norway affects its security and economic development (Stokke, 2012).

Moreover, the mediator is driven by norms and values. The mediation mentioned above is motivated by idealism and normative conceptions. The aim of the low-leverage state, in this case, is to promote international peace and end conflicts. Consequently, the mediators with norm-driven motivation adopt universal norms and work for the common good of the international community. In direct contrast to norm-based mediation is interest based-mediation. Interest-based low-leverage states are engaged in mediations due to interests like investment, security, prestige, or international standing. For Example, Qatar

engages itself in international mediations for security reasons. The geostrate-gic location of Qatar makes Qatar vulnerable. Qatar is a buffer state between two regional hegemon countries, Saudi Arabia and Iran. Based on this, Qatar lives with a degree of looming threat (Fazal, 2011; Jesse & Dreyer, 2016). Consequently, Qatar involves itself in international mediation on a security basis. Qatar believes that being an international mediator would increase its international status and influence in the international community, and by this way, Qatar could secure itself (Khatib, 2013a). Furthermore, Qatar considers its international mediation as a tool for expanding its future investment and protecting its current investment (Khatib, 2013; Barakat, 2014).

Although both Norway and Qatar involve themselves in international mediation based on security reasons, each behaves differently in international mediation. The difference between Qatar and Norway is that the Norwegian engagement in international mediation is motivated by idealism and realism with the absence of significant difference between them (Stokke, 2012). This means that the Norwegians are motivated by both the norms and their interest, which is represented by achieving status in the international arena and securing themselves. In other words, as Stokke(2012) mentions that Norway has achieved a balance between its interest and its norms, idealism and realism. In its engagement in international mediation, Norway adheres to the norms of its international mediation. However, it achieves its interest. In a similar vein, Eriksson confirms that the adherence to international law and encouraging peaceful resolutions for disputes would strengthen the security of small states such as

Sweden, and at the same time, strengthen international peace and security. (Eriksson, 2015).

In a different vein, interest-based mediators fail to balance their interests and the norms of morality. As a result, those mediators mainly engaged themselves in peace to achieve their interests. For example, Qatar has failed to balance morality or norms and its interest, represented in investment in the disputant's territories. However, the Qatari claim that their mediation is based on morality and promoting peace and stability in the Middle East. Qatar is motivated by a combination of survival strategies and international status or prestige (Kamrava, 2011) and in addition to its investment interest (Khatib, 2013). In brief, the point of departure for Norway and Qatar in international mediation is their interest and norms-based motivation. While Norway succeeds in balancing achieving its interest and being a norm entrepreneur, Qatar fails to balance the morality and norms and likely prefers to go with its interest.

The motivation of the low-leverage state is reflected in the way low-leverage states norm-based mediator and low-leverage interest-based mediator employs their resources (usage of resources). Both use those resources in a different ways. For example, norm-based mediators use their economic resources as humanitarian aids to help the disputants. Offering those aids to the disputants contribute to building trust between Norway and the disputants. In contrast, the interest-based mediator uses its economic resources to incentivise the disputants to reach an agreement. How the mediator uses the resources is an essential factor in determining the mediator's type of strategy in the mediation process. For the mediation strategy, the norm-based low-

leverage states are expected to adopt a facilitation strategy in harmony with the balance between the norms and interest motivation. They do not seek to affect the parties or even interfere in the mediation process; they play a facilitation role rather than manipulation. They choose the facilitation strategies because they represent a low level of intervention in which the mediator plays a passive role, akin to a communication channel, and has little control over the mediation process. The norm-based low-leverage states aim to ensure continued discussion between the disputants by using a facilitation strategy. In the same vein, norm-based mediators are expected to adopt impartiality as a tool of equity in their mediation. For a mediator to be impartial, he/she must deal with all parties in a similar manner, both in terms of procedure and in terms of substance. It is expected that the norm-based low-leverage state will remain unbiased towards all parties in the dispute and the interest and options for settlement. For norm-based low-leverage states, impartiality is expected to be the basis for them to build personal trust and confidence in them, which leads to their acceptability, which is crucial for their success.

Finally, the norm-based low-leverage states are expected to be classified as pure mediators who use persuasion to facilitate negotiation between the disputants and build social ties while enhancing a communication channel between them.

As an example of a norm-based low-leverage state mediator, Norway chooses to be a norm entrepreneur, which simultaneously contributes foreign aid to the impoverished on a global scale and seeks to enhance international peace and security. Its foreign policy is one of peace. As a norm-based mediator, Norway

would instead employ the facilitation strategy in its international mediation and would not seek to manipulate the disputants towards an intended direction. It is expected that impartiality is a crucial factor for Norway. In contrast, interest-based low-leverage states' mediators are expected to adopt a strategy of manipulation where the mediators affect the content and the substance of the mediation. Additionally, the mediator is able to offer incentives or carrots (rewards). The interest-based low-leverage state mediators are expected to issue new proposals and guide the disputants towards signing an agreement.

It is expected that impartiality is not necessary for interest-based low-leverage states mediators. It is expected that they believe that their influence and will-ingness to protect the parties' interests are the main reasons behind their acceptance as mediators, not their impartiality. It is expected that they believe that the mediator's leverage and not impartiality leads to mediation success. Finally, interest-based low-leverage states mediators are expected to be classified as power mediators who use their resources to push the disputants towards their preferred direction to ensure their interest in the mediation outcome is reached.

As an interest-based low-leverage state mediator, Qatar's interest is to increase its security and seek to be a leading actor in the Middle East and promote a business environment conducive to Qatari investments. Despite being a low-leverage state, Qatar can incentivise the disputants towards signing an agreement (Kamrava, 2011). Manipulation strategies like this are commonplace in Qatari international mediation. Qatar deployed financial leverage to encourage the conflict parties in Darfur and Yemen to accept a solution, which

means that Qatar brought its pure position in mediation into question as financial leverage is a feature of a power mediator such as the USA (Barakat, 2014). It is expected that it is leverage, not impartiality, which is necessary for the power mediator.

Based on the literature review and the above discussion, it is expected that the norm-based- low-leverage states make a balance between their interest in mediation and their norms, be impartial mediators, adopt facilitation strategy. In addition, they behave as pure mediators, not power ones. Conversely, interest-based low-leverage states mediators are expected to have an interest in the mediation, and they lack the balance between their interests and the norms of morality. Furthermore, they use a manipulation strategy to achieve their interest. For them, impartiality is not a necessary component. Since they are states with low leverage, it is assumed that they cannot play the role of a powerful mediator. At the same time, they cannot be pure mediators because they engage themselves in mediation based on their interests.

However, the actual findings of this thesis are slightly different from the expected ones. It is found that norm-based mediators keep the balance between their interests and norms in the mediation. However, they are not impartial, and they adopt a manipulation strategy, not a facilitation one, as expected. In contrast, the interest-based low-leverage states do have an interest in mediation. They lack the balance between their interest/s and morality or norms in the mediation. While they do behave as impartial mediators, employing manipulation strategy and acting as power mediators.

In general, classifying low-leverage states into norm-based mediators and interest-based mediators should not be considered a rigid dichotomy or binary division. In other words, it is not a binary classification; it should be considered a spectrum classification. In other words, there is always an overlap area between the two types of mediators. Based on the literature, the following table briefs the expected difference between the two cases of low-leverage states in mediation; the norm-based mediator and the interest-based mediator.

Table (1): Norm-based mediators versus interest- based mediators (expected behaviour)

	Low-leverage norm- based	Low-leverage-interest-
	mediators	based mediators
Interest/ motivation	- norm entrepreneur. - Self- interest (security/ gain status in international community/ increase influence in International Rela-	Self-interest (Investment, security/ gain status in international community/ increase influence in International Relations)
	tions).	

Balance between	Make balance between	No balance between their in-
norms and interests	norms and interests.	ity.
Strategy	Facilitation strategy	A spectrum of facilitation, formulation, and manipulation.
Impartiality	Impartial mediator	Not necessary to be impartial
Being power or pure mediators	Pure mediator	Power mediator

CHAPTER THREE: NORWAY VERSUS QATAR IN INTERNATIONAL MEDIATION

Introduction

Different low-leverage states have different motivations and models or approaches to international mediation and do not continuously pursue similar foreign policy goals when using international mediation as a foreign policy tool (Gigleux, 2016). Regarding international mediation motivation and approaches, low-leverage states can be divided into two approaches. The first approach views low-leverage states as interest-based motivation, with their interest in mind. The second approach views them as norm entrepreneurs or norm-based mediators (Hansen, 2013).

Norway and Qatar are examples of low-leverage states in international mediation; each has developed its niche diplomacy, where niche diplomacy refers to a popular approach any state has developed to obtain the best returns and the widest international recognition and attention (Lakatos, 2017). Norway is considered to be an example of a norm entrepreneur because of its foreign policy and international mediation approach, which is norm-driven, wherein Norway plays a role as a facilitator mediator who has no interest in the mediation outcomes except peace promotion all around the world(Bandarage, 2011; Leira, 2013; Waage, 2005) or it succeeds to make a balance between its interest and norms (Stokke, 2012). In contrast, Qatar's form of international mediation is interest-driven mediation (Hansen, 2013; Kamrava, 2011, 2015; Khatib, 2013b; Peterson, 2006b).

This chapter is an essential part of this thesis because it shows the expected behaviour of the two different types of mediation of low-leverage states: norm-based mediation and interest-based mediation. As mentioned in the previous chapter, it is

expected that a norm-based mediator is an impartial mediator who employs facilitation strategy and plays the role of a pure mediator. Besides, an interest-based mediator is expected to be a manipulative mediator who does not pay much attention to impartiality and is expected to be a pure mediator.

Studying four cases of Norway and Qatar mediation proves a difference between expectation and reality. The four cases illustrate that a norm-based state and low-leverage state may act as a manipulator who is not necessarily an impartial mediator. Besides, the interest-based mediator sticks to impartiality, behaving like a power mediator who manipulates the disputants to reach an agreement. To shed light on those two approaches in international mediation, this chapter focuses on the track records of both Norway and Qatar in international mediation.

Norway in international mediation

Despite being low-leverage in international mediation, Norway has its hallmark style on the international stage. The former US president, Barack Obama, described Norway as a country that punches above its weight, referring to its peace-making abilities, when he met with the Norwegian prime minister Jens Stoltenberg in 2011 at the White House (Jakubec, 2015).

The Oslo Accords first established Norway's reputation as a peace-mediator in 1993. Its role in the secret negotiations between the Israelis and the Palestine Liberation Organization (PLO) was a milestone for Norwegian diplomacy and international peace (Jones, 1999). The unprecedented success that Norway achieved through the Oslo Accords has laid the foundations for many requests to assist in conflict resolution, from Kashmir to Kurdistan and from Cyprus to Guatemala (Kelleher &

Taulbee, 2006). The Guatemalan Peace Agreement of 1996, for example, further bolstered Norway's reputation as an international peace mediator and facilitator. Building the roads to peace has since remained the overarching goal of Norwegian foreign policy (Jakubec, 2015). A Norwegian governmental White Paper officially confirmed this, which affirms the priority of peace work in Norwegian foreign policy. This peace work was to be carried out primarily through political commitment and development efforts and affirms that Norway should be a proactive and creative state, making a real difference and impact in the international arena (Jakobsen, 2012). Based on this, Norway has been engaged in many peace processes since 1993, including those of Afghanistan, Colombia, Myanmar, Nepal, the Philippines, Somalia, Sri Lanka and Sudan (Fabra-Mata, 2014).

Norway has several advantages that enable it to function as a trusted mediator in international disputes. Norway lacks a colonial past, which often sources bias and self-interest in international mediation for other countries. As it turns out, Norway has not done much harm to anyone since the days of the Vikings (Ford, 2000). Due to lacking any history of colonialism, Norway has been able to represent itself as a credible neutral mediator in conflict resolution (Kelleher & Taulbee, 2006). Bandarage (2011) claims that Norway's involvement in peace reconciliations is not motivated by any self-interest, political, economic or otherwise, which in turn reinforces Norway's credible position in the international community as an impartial and trusted mediator (Bandarage, 2011). Additionally, Norway's low level of leverage prevents it from pursuing power mediators. Consequently, it is vital for Norway as a mediator to maintain safe and friendly relations with its international counterparts, as Norway has low leverage to reach and implement agreements. Thus, it needs key international actors on

side (Sørbø, Goodhand, Klem, Nissen, & Selbervik, 2011). It is not surprising, therefore, that Norway has good, stable and close relations with many key international actors, such as the USA and EU (Bandarage, 2011).

However, one consideration is Norway's powerhouse economy, built on oil and natural gas. Indeed, it was only after the discovery of oil in 1970 that Norway became one of the richest countries in the world (UNDP, 2016). Today, Norway has the 40th largest Gross Domestic Product (GDP) globally and the 11th largest GDP per capita (The World Bank, 2018). Being an oil-rich country enables Norway to follow its peace initiatives (Kelleher & Taulbee, 2006). Meanwhile, Norway is considered a leading donor of international development assistance, with 1.06% of its Gross National Income (GNI) directed towards development aid, which means it sits as the second-most generous country after Sweden.

Moreover, Norway is the third-largest contributor to the United Nations after the USA and the UK. For example, in 2018, it increased its contribution to the Central Emergency Response Fund (CERF) from NOK 380 million to NOK 420 million (approximately \$52 million) a year for the next four years (UNDP, 2016). Norway also provides bilateral aid via its NGOs to many countries in the Middle East and Africa. Norway's generous development assistance through its NGOs and its international role in promoting international norms, in addition to its prominent engagement in the UN, all play an important role in maintaining Norway's reputation as a committed international peace actor (Kelleher & Taulbee, 2006).

Furthermore, Norway has developed its model in international mediation. This model was invented by Jan Egeland, the Norwegian diplomat who supervised the

backchannel for the Oslo Accords negotiations. Egeland was affected by his work with INGOs like Amnesty International and the Red Cross before joining the Norwegian Ministry of Foreign Affairs (MFA) as an advisor and as state secretary. His NGO background provided him later with the contacts and credibility needed to establish informal cooperation between the MFA and NGOs, which became the institutional basis of the Norwegian model(Jakobsen, 2012).

Norway has succeeded in developing its niche in mediation different from the power politics paradigm of other international actors who have sought to dominate the international political arena, such as the USA. Norway's successful mediation in international disputes proves that a state such as Norway, with a good international record, could assist disputants in resolving their conflicts (Moolakkattu, 2005).

Reviewing the literature shows that seven core features characterise Norway's international mediation model: norm-driven motivation, Track One and a Half Diplomacy, secret diplomacy, impartiality, facilitation strategy, backing by an international key factor and the Norwegian leverage.

Norm-based motivation

The first feature of Norwegian international mediation is the Norwegian motivation in international mediation. The Norwegian motivation is described as a normbased motivation. The literature describes Norway as a moral entrepreneur or norm entrepreneur (Leira, 2013). In order to understand the significance of normbased motivation, it is essential to understand the meaning and the background of the following two concepts: norms and norm entrepreneurs.

By the end of the Cold War, there were calls to evaluate the way people thought about power in international politics and how states should interact in the international community. By that time, many states had already developed practices and standards of acceptable behaviour, or norms, which reflected the states' interest and identity (Ingebritsen, 2002). Norms do not emerge from thin air. They are built over time by agents—individuals or states—who have strong ideas and beliefs regarding appropriate behaviour in their community. Scandinavian countries are good examples of norm entrepreneurs (Ingebritsen, 2002).

Norms are defined as collective thought about proper behaviour for a given identity.

Norms are intersubjective understandings that constitute actors' interests and identities and create expectations as well as prescribe what appropriate behaviour ought to be by expressing values and defining rights and obligations (Björkdahl, 2002).

While norms are the identification and stipulation of the appropriate behaviour for an actor with a certain identity (Björkdahl, 2013), the Norm entrepreneurship is the ability to identify or create opportunities to alter the existing behaviour of others in the direction of the new norm (Björkdahl, 2002). In other words, norm entrepreneurs have strong ideas about appropriate behaviour in their community and set out to alter the behaviour of other communities (Finnemore & Sikkink, 1998). Indeed, a state is considered to be a norm entrepreneur when it has a solid commitment to a particular normative idea or behaviour and is characterised by a strong willingness to change the behaviour of others by promoting this idea (Björkdahl, 2013). Accordingly, Norwegians in their foreign policy and international mediation are guided by idealism and normative concepts. Indeed, this is the reason why many call Norway a moral entrepreneur or norm entrepreneur (Bandarage, 2011). Norway has developed a

national consensus that is built around social-democratic norms that stress international concerns for those around the world who are oppressed (Helgesen, 2003). Consequently, norm entrepreneurship is a diplomatic strategy that involves setting normative standards in the international community in effort to change it (Björkdahl, 2013). According to norm entrepreneurship strategy, low-leverage states such as Norway build their national identity based on universal norms and introduce themselves as good stewards, working for the common good of the international community (Ingebritsen, 2002). These states are agents of normative change around the world. They play this role by providing developmental assistance, offering humanitarian aid and through the promotion of democracy, international peace, and human rights (Björkdahl, 2013; Brysk, 2005).

Acting as a norm entrepreneur, Norway contributes foreign aid to the poor around the world, seeks to enhance gender equality and the role of women, and endeavours to promote peace by pursuing security and mediation in conflicts (Björkdahl, 2013). Thus, Norwegian foreign policy is a foreign policy of peace, designed to make the world a better place (Leira, 2013).

In a different vein, low-leverage states sometimes engage in international mediation for reasons other than humanitarian reasons. The low-leverage states sometimes engage themselves in international mediation for their interest. This interest may be represented in improving their status in the international community by extending their international influence and power. Additionally, international mediation may increase the usefulness of low-leverage states and help them to become more independent concerning their more vital allies (Touval & Zartman, 2007; Zartman & Touval, 1996). Accordingly, It is argued that Norway is engaged in international

mediation not only to promote peace, better organise the international community and increase international justice and international cooperation but also to protect Norway's interests in other areas of the world (Henriksen Waage, 2007). This is truly applied to Norway mediation. However, Norway succeeded in balancing its interest and its norms. For low-leverage states, becoming norm entrepreneurs presents the opportunity to influence world politics and to achieve its interest (Björkdahl, 2002). Their motivation is also usually directed by altruism and ideational commitment (Goertz and Diehl 1992; Jacobsen 1995: 291; Finnemore and Sikkink 1998: 898). Altruistic motivations are meant to promote or defend values, ideas, or norms, and seek to contribute to the realization of these values or norms in practice (Elgström 1982: 33). Ideational commitment, for example, may motivate norm entrepreneurs to promote certain norms because of the moral and ethical values that underpin them (a moral commitment to something that is perceived as ethically right). The norm entrepreneurs create their niche in international politics, and the norms they promote are for the international community's benefit. They usually take the lead in a specific issue and adopt strategies to influence the world and make it a better place to live. Furthermore, they are usually perceived as impartial, and few suspect them of pursuing a self-interest (Björkdahl, 2002). For low-leverage states, their peace policy is affected by their ideas about peace. Their engagement in peace activities is related to the state identity and the values that the state upholds (Wivel, 2013).

Norway's identity is dominated by a so-called "liberal Norwegian peace identity" (Leira, 2013). In order to conduct humanitarian activity and peace work, Norway shares its resources with others. It aims to bolster the liberal approach to the world. The Norwegians view making a difference not as an option but as an obligation.

Additionally, they are convinced that such actions are essential for the creation of a better world and believe that they have an essential role to play in the process (Leira, 2013). However, Norway is a developed country that seeks to adopt the World Bank model of "private sector development" to expand upon its economic opportunities. The profits gained from the Norwegian gas and oil companies working in Africa, for example, amount to more than what Norway pays to Africa in humanitarian aid (Curtis, 2010). In the same fashion, Norway conducts peace activities in Sri Lanka, where some view its role as both a peace facilitator and economic investor as a conflict of interest (NORAD, 2002).

Despite, the Norwegian claim that they are a peaceful nation (Leira, 2013), its leading role in exporting arms contradicts this claim. Norway was the 10th largest exporter of the weapons (The Nordic Pace, 2017). However, Norwegian officials advise that the policy is to refrain from exporting Norwegian weapons to countries at war, or countries where there is a threat of war or any environment with conditions favourable for civil war (Bandarage, 2011). However, Norway has routinely exported its weapons to countries that fit this description, such as Saudi Arabia, Yemen, and the Arab Emirates, all of which are involved in the war in Yemen (The Nordic Pace, 2017). Accordingly, this creates a blatant conflict of interest for Norway's international peace-making initiatives.

Track One and a Half Diplomacy

The second feature of the Norwegian international mediation is the Track One and a Half Diplomacy. The Track One and a Half Diplomacy is a "hybrid diplomacy" because it is a cross-fertilization of Track One and Track Two (Mapendere, 2005). It is

essential to know the meaning of Track One Diplomacy and Track Two Diplomacy. Track One Diplomacy refers to the direct involvement of official representatives from governments in diplomatic activities. It is carried out by diplomats, high-ranking government officials, and heads of state, and it aims at influencing political power structures. Additionally, It is usually considered to be the primary peace-making tool of a state's foreign policy (Mapendere, 2005).

Track Two Diplomacy refers to the involvement of unofficial representatives such as NGOs. It takes the form of informal interaction between members of nations to develop strategies, influences public opinion and organises human and material resources to assist in conflict resolution. It is a complement to Track One Diplomacy and serves as a bridge to negotiations(Jones, 2015; Kelleher & Taulbee, 2006; Mapendere, 2000, 2005; Montville, 1991; Nan, 2005; Richmond, 2000; Volkan, 1991).

Scholars of conflict resolution argue that Track One Diplomacy and Track two Diplomacy do not accurately reflect the full range of peace-making activities carried out by retired politicians, religious leaders, and by organizations such as The Carter Centre and the Norwegian Refugee Council (Nan, 2005). For this reason, those scholars have developed a new category known as Track One and a Half Diplomacy (Mapendere, 2005). Track One and a Half Diplomacy consists of a public or private interaction between official representatives of conflicting governments, or political entities such as popular armed movements and nonofficial facilitators or mediators, to change the attitudes of the parties involved (Mapendere, 2000). In other words, One and a Half Diplomacy is an initiative that involves a combination of unofficial facilitators and official representatives from the conflict in question (Nan, 2005). Therefore,

the main feature that distinguishes Track One and a Half from Track One is that the third party is not an official representative of any government. As mentioned, Track One conflict resolution efforts are facilitated by government representatives or representatives of political institutions. For example, President Clinton's Camp David mediation between Yasir Arafat and Ehud Barak is purely Track One Diplomacy, while Former Finnish Prime Minister Martti Ahtisaari's mediation in Aceh is Track One and a Half Diplomacy.

Involving Track Two Diplomacy gave the disputants a space to freely negotiate and be formal under the pressure of the Track one Diplomacy. It creates the workshop-based approach, and it improves the underlying relationships between the disputants (Jones, 2015).

Similarly, the main feature that distinguishes Track One and a Half from Track Two is the parties involved in the process. In Track Two Diplomacy, the parties involved in the conflict resolution process are not official representatives of the conflicting sides. This is not the case for Track One and a Half, where the parties involved in the conflict resolution process are official representatives of the conflicting parties.

The Norwegian approach to international mediation is conducted by Norwegian officials from the Ministry of Foreign Affairs with the cooperation of the Norwegian NGOs. The Norwegian NGOs personnel help facilitate the communication channel between the disputants and the Norwegian officials. Track One and a Half Diplomacy is represented in the strong relations and cooperation between the government and the Norwegian NGOs (including development and research organizations). There is a mutual dependency between the Norwegian government and the Norwegian NGOs

whereby the Norwegian government depends on NGOs for the execution of its foreign policy, especially in the field of international mediation, and Norwegian NGOs depend on the Norwegian government for funding (Larssen, 2009). Norway's longstanding peace tradition, the close connection between the state and the NGOs, and available funds all enable Norway to conduct its foreign policy of peace and mediation (Leira, 2013).

Thus, the facilitator is not an ordinary citizen, One and a Half Diplomacy. Considering this, it is evident that Norwegian international mediation is a case of Track One and a Half diplomacy. This is confirmed by Kelleher and Taulbee (2006), who suggest that the relation between the Norwegian government and NGOs in international mediation can be defined as Track One and a Half Diplomacy (Kelleher & Taulbee, 2006).

The Norwegian approach to peace-making is based on the Norwegian NGOs or research organisations' field experience and local connections. Examples of those NGOs are: the Institute for Labour and Social Research, FAFO, in the Middle East, Norwegian People's Aid and Norwegian Church Aid in Sudan (Bersagel, 2008; Kelleher & Taulbee, 2006; Nissen & Waage, 2015). Norway integrates its academic and NGO expertise with the knowledge and resources of the Ministry of Foreign Affairs (Kelleher, 2006). The Norwegian government and the Norwegian NGOs, represented in academic and humanitarian organizations, work in a harmonious symbiosis in any mediation process in which Norway is engaged.

The cooperation between the Norwegian NGOs and the Norwegian government may take different forms or roles. First, the Norwegian NGOs can accomplish several things which a government would not attempt, such as using deniability and disguise

to cover up secret negotiations between the disputants (Bandarage, 2011). Second, with the help of the Norwegian NGOs and via their wide network built by work on the ground, the Norwegian Ministry of Foreign Affairs estimates the right time to intervene in any conflict reconciliation (Bandarage, 2011). Third, the Norwegian NGOs have earned a valuable reputation on account of their several decades of experience in various countries. Fourth, the Norwegian officers and peace facilitators tap into the experience and contacts of these NGOs in their peace activities (Hanssen-Bauer, 2005). Fifth, the government depends on NGOs to initiate and facilitate the negotiation processes. The Norwegian initiative, therefore, draws upon the blending of government resources and the experience of the Norwegian NGOs (Egeland, 1999). Finally, these NGOs are active in the area of the conflict and provide and essential channels for trust-building between Norway and the disputants (Bertram, 1995).

Norwegian NGO personnel play an essential role in trust-building for mediation efforts by playing a two-pronged role in mediation. Firstly, they offer vital knowledge and information about the local people in the countries they serve, including crucial contact channels in those countries. Secondly, they contribute to building Norwegian public opinion favouring the targeted initiative.

In brief, the Norwegian government supports NGO peace work as a way to facilitate the mediation process (Egeland, 1999; Walraven, 1999). The NGOs also benefit from Norway's image as a peace mediator (Kelleher & Taulbee, 2006). Thus, NGOs have significant influence because of their close relationship with government officials in Norway and disputant countries. As a result, they are often responsible for bringing disputants together by building trust (Bandarage, 2011).

Going beyond an idealization of the role of Norwegian NGOs in the Norwegian model, how they play a role in the development and representing them as a virtuous force in promoting international peace, Goonatilake (2004, 2006) criticises the role that Norwegian NGOs play in Sri Lanka. Goonatilake accuses Norway of using its NGOs to manipulate politicians and political institutions to promote Norwegian investment at the expense of Sir Lankan sovereignty. Thus, whether Goonatilake's claim is accurate or not, it does not negate the prominent role of Norwegian NGOs in promoting international peace by creating relations between the disputants and the Norwegian government. It also reflects on the desire of disputants to ask Norway to be a mediator. Sri Lanka itself is a strong example. The Prime Minister, Wickremesinghe, wrote to the Norwegian government in 2001, asking it to continue its mediation efforts in Sri Lanka. By involving the Norwegian NGOs in peace efforts as non-state actors, it helped the mediation to proceed in secret (Sørbø et al., 2011).

Secret Diplomacy

The third feature of the Norwegian model of international mediation is secret diplomacy. Secret diplomacy is distinguished by performing mediation or negotiations away from the media and the public (Bjola, 2014; Gilboa, 1998). In secret diplomacy, the public is kept away from the diplomatic events, and its only officials a high level of discretion are involved in the negotiations (Gilboa, 1998; Pruitt, 2008). Indeed, Norwegian peacemakers prefer to work behind the scenes and away from the media and critics (MFA, 2020).

Secret diplomacy has both advantages and disadvantages. Among many advantages is what the Norwegians themselves think about secret diplomacy. The

Norwegians believe that isolation helps disputants focussed on finding common ground between their respective positions (Kelleher & Taulbee, 2006). Thus, Oslo meetings were held privately, which helped build trust and increase the chances of an agreement being reached. Pruitt (1997) argues that Norway's logic is that secret and private mediation encourages informal interpersonal contacts between disputants and eliminates the prying eyes of journalists (Pruitt, 1997).

Secret diplomacy has other key advantages, such as unlocking peace negotiations by providing a suitable environment for constructive negotiations and by preventing the spoilers from ruing the negotiations (Eban, 1983). It is an opportunity to save face and ensure the security of the negotiations (Gilboa, 1998). Without secret diplomacy, there would not have been agreements. This is the case with the Oslo Accord of 1993 since neither Israel nor the Palestine Liberation Organization (PLO) would negotiate openly due to each leadership's internal political constraints (Lieberfeld, 2008). According to Qurei (2006), for Israelis, when the Oslo channel started, it was illegal to contact the PLO according to Israeli law. This was changed later in the Israeli Knesset. He adds that for the Palestinians, the PLO was concerned with the internal reaction among the Palestinians in the Gaza Strip, West Bank, East Jerusalem, and the diaspora (Qurei, 2006).

Bjola suggests that secret diplomacy is beneficial in terms of normalising relationships with former adversaries, especially for protracted disputes (Bjola, 2014). The Palestinian–Israeli conflict was a case of negotiations between two old enemies.

Qurei (2006) confirms that neither of the two parties felt comfortable in the first round of the negotiation. The Norwegian facilitation role helped both sides to contact each other more appropriately. It was secret diplomacy that helped the Oslo negotiation to

continue without the interruption of unwanted political events such as those that had occurred in the Washington negotiations(Qurei, 2006).

Despite the advantages of secret diplomacy, it is still considered a violation of the obligation of elected leaders to represent the public and stand accountable for their decisions. This requires that affairs of domestic concern be determined not by a group of leaders in the secret conclave but rather open to public debate (Bjola, 2014; Lieberfeld, 2008). Despite the claim that secret diplomacy is a factor behind achieving agreement, evidence suggests that such agreements are short-lived. Agreements accumulated under public scrutiny and with public transparency have a better chance of successful implementation (Bjola, 2014). Additionally, secret mediation contributes to the creation of an environment of distrust between the people and their leaders (Bjola, 2014).

The Norwegian Impartiality

The fourth feature of the Norwegian model in international mediation is the impartiality of the Norwegians in their mediation. While Kydd (2003), Zartman and Touval (2007), Svensson (2007) argue that a biased mediator is often effective, Kleiboer asserts that impartiality is the basis for building personal trust and confidence in the mediator, which in turn leads to them being accepted, which is key for mediation success (Kleiboer,1996). Slim (1992) confirms that mediators are more effective when they are impartial and when they lack interest of their own in resolving the dispute (Slim, 1992) Accordingly, Norway, as a low-leverage state, being biased may negatively affect Norway's presence in mediation. Ford (2000) claims that a pivotal

strength of the Norwegian model is that Norwegians do not have stakes or strategic interests that affect their impartiality in mediation (Ford, 2000).

Norway's image as an impartial mediator came from its long commitment and advocacy for addressing the problems of developing countries (Kelleher & Taulbee, 2006). This was especially evident during anti-colonial revolutions, where Norway often supported proposals such as the new international economic order (Egeland, 1988). Additionally, despite being a member of NATO, Norway permits only its own troops within its territory and denies bases and garrisons to other member countries. Naturally, this reinforces its image of independence and neutrality (Holst,1983). However, Norwegian impartiality was brought into question in the Sri Lanka mediation. Höglund & Svensson (2009) claim that Norway was accused of being biased towards the Liberation Tigers of Tamil Eelam, LTTE (Höglund & Svensson, 2009). The accusation of Norway by being biased towards LTTE may refer to the fact that Norway was one of the few who had public contact with LTTE. Norway was playing a facilitator who should see the parties in an equal term. However, the President of Sri- Lanka did not want the LTTE to be in equal status to the Sri- Lankan government (Moolakkattu, 2005).

In sum, Norway behaves as an impartial mediator, and the Norwegian impartiality is considered Norway's point of strength and the key success in Norwegian international mediation.

Facilitation strategy

The fifth feature of the Norwegian international mediation model is using of a facilitation strategy. As a state with low leverage in the international mediation arena,

Norway plays the role of a facilitator rather than a manipulator. While a manipulative mediator uses its resources to push parties in its preferred direction and exercises its leverage over the parties in order to make them comply, the facilitator works on gaining the confidence of the parties, building social ties among them and enhancing communication between them (Svensson, 2007b).

The Norwegian mission in mediation is to promote mutual understanding between parties, acting as a channel of communication that allows the disputants to find common ground between their respective positions. They endeavour to bring the parties together and keep them together, usually taking a more active role only when they approve it. Kelleher & Taulbee (2006) argue that the fact that Norway has the carrots, in this case, economic resources, to use as incentives to reward disputants, but no stick with which to threaten them greatly contributes to its role as an active facilitator (Kelleher & Taulbee, 2006). In the same vein, Ingebritsen (2006) confirms that Norway is an activist facilitator. Ingebritsen states that the Norwegians try to implore, convince, propose, suggest, prod, and ultimately help. However, Kelleher and Taulbee (2006) may be inaccurate in their description of Norway as a mediator equipped with carrots but no stick. Norway does have a stick, which is represented as the withholding of aid. Despite this, Norway chooses not to use this tool due to normative considerations. Norway rarely undertakes a mission where it serves as the sole facilitator. Instead, it usually plays a complementary or support role. Examples include the United Nations involvement in Guatemala. While lacking significant means to make effective threats, Norway cannot guarantee settlement without the participation of others (Kelleher & Taulbee, 2006).

As mentioned above, Norway mediation is suitable for bringing parties to the negotiation table; however, it needs harder leverage to reach and implement the settlement. And to do that, Norway needs to build links with other powerful actors and coalitions (Sørbø et al., 2011). In other words, Norway needs the backing of an international key factor.

Backing by an international key factor

An international key factor backs Norwegian international mediation's sixth feature. This international key factor could be a powerful state such as the USA or an international organization like the UN. Grøn & Wivel (2011) argue that low-leverage states are unable to affect international peace without allying and cooperating with other states who possess more significant influence (Grøn & Wivel, 2011). Sørbø et al. confirm that Norway, as a low-leverage state, cannot accumulate peace agreements without the backing of an international key factor such as the USA or UN. The Norwegians are aware that they would not achieve peace without such support. With the backing of an international key factor, Norway can borrow leverage to compensate for its relatively limited leverage (Sørbø et al., 2011).

Qurei (2006) presents the Oslo backchannel as an example. He explains that

Norway was backed by the USA, which was not informed about the secret channel
until May 1993 at the request of Israel. However, the moment the negotiations took a
formal position, Norway informed the USA about the secret channel between the
PLO and Israel, assuring the USA that the channel was not an alternative to the talks
in Washington but rather complementary support for it(Qurei, 2006). Similarly, Corbin
(1994) and Makovsky (2018) assert that Norway is always concerned about being a

friend to the USA, so in the Oslo secret backchannel, Norway was hesitant about keeping the United States in the dark. The Norwegians believed there could be no peace in the Middle East without the USA's approval. They add that after the negotiations became more official by including Israeli officials, the Norwegians updated the Americans of the Oslo negotiations (Corbin, 1994b; Makovsky, 2018). Corbin mentions the importance of the US in the Oslo agreement; she asserts that American approval of the Oslo agreement would grant international credibility for the agreement and assurance that the Americans would be committed to the agreement and use their leverage to implement the agreement (Corbin, 1994b). Differently, Norway was not alone in the negotiations between the Sudanese government and the rebels in South Sudan. Norway was part of a Troika, which included Norway, the USA and the UK as external observers in support of the mediation efforts of IGAD (Kelleher, 2006; Kelleher & Taulbee, 2006).

The Norwegian leverage

The seventh feature of Norwegian international mediation is Norwegian leverage. Norway's leverage in international mediation can be categorized as trust-based leverage and the disputants' need-based leverage. Thanks to the Norwegian One and Half diplomacy, the Norwegian became a mediator who could gain the trust of the disputants. Regarding trust-based leverage, the Norwegians build trust between themselves and the disputants via Norwegian NGOs that are considered to be the backbone of Norwegian trust-based leverage. The Norwegian non-governmental organizations (NGOs) include development organizations and research-oriented NGOs. As mentioned above, the Norwegian government depends on NGO personnel

who have experience with the local community experiencing conflict. Indeed, NGOs help Norwegian diplomats access appropriate communication channels and help build trust and confidence between Norway as mediators and the disputants. Additionally, the resources Norway has at its disposal for humanitarian aid via its NGOs is considered as leverage that Norway uses in its international mediation (Kelleher & Taulbee, 2006).

FAFO is one example of a Norwegian NGO which played an essential role in the Norwegian historical relations between Norway and the Palestinians and Israelis (Corbin, 1994; Qurei, 2006). Likewise, the Norwegian Church Aida (NCA) and the Norwegian People Aid (NPA) played a distinguished role in the mediation success in the Government of Sudan and the SPLM/A conflict (Nan, 2005). The power of those NGOs lies in their persuasive ability. The principal element of mediation leverage is the mediator's ability to persuade the disputants of the necessity for mediation to reach conflict resolution. According to Zartman and Touval, the mediator's persuasion ability is dependent upon the need of the disputants for a resolution to their conflict which cannot be achieved otherwise (Zartman & Touval, 1996). As for the mediator's persuasion ability, they should be able to convince the disputants of the attractiveness of the proposed conciliation using their communication channels.

In this capacity, Norway fills the role of facilitator. The mediator's persuasion ability depends on the sincere need of the disputants for a solution (Zartman & Touval, 1996). For the disputants to need a solution and seek mediation, they should first recognise that they have reached a mutually destructive stalemate where both of them can no longer stand the situation (Zartman & Touval, 1996), meaning the

conflict should already have reached the escalatory phase in the conflict cycle (Siniver, 2006). The need of the disputants is further encouraged by the local, regional and/or international context.

For example, in the Oslo negotiations, the Gulf war negatively affected the Palestinian Liberation Organization (PLO) position as it was facing severe financial and political difficulties. At the same time, the PLO saw an opportunity to negotiate in the Israeli labour party. Those circumstances helped increase the PLO's willingness to be mediated, and they asked Norway to be a mediator. On the Israeli side, the Palestinian's first intifada and Hamas' domination of Palestine was a concern for Israel (Lieberfeld, 2008).

Qatar in international mediation

Since the mid- 2000s, Qatar has emerged itself as a peace mediator in regional conflicts, notably in the Middle East and Africa. Qatar proves that geographical size poses no barrier. Qatar succeeded in creating a distinct niche for itself in the regional and international community. Qatar has endeavoured to brand itself as an impartial peace maker across the Middle East and Africa and within the larger international arena. The most notable conflicts that Qatar has been involved in as a mediator include Lebanon, Sudan, Yemen, Palestine and finally, the border conflict between Djibouti and Eritrea.

Qatar's success is derived from a carefully combined mixture of diplomacy, marketing, domestic politics, regional diplomacy, and through the strategic use of its sovereign wealth fund (Kamrava, 2015). To better understand Qatar's motivations and

aptitudes, which enable it to be a successful mediator, it is first necessary to study the background of Qatar and the reason behind Qatar's niche in international mediation.

Two essential factors enhance Qatar to be a mediator: the Qatari wealth and the location of Qatar. Qatar is an inordinately wealthy country with an exceedingly small population. It has the highest growth rate in its real gross domestic product compared to any other country in the Gulf Cooperation Council (GCC) or the larger Middle East (Kamrava, 2015; Khatib, 2013a). This has enabled ruling elites to ensure socioeconomic security for their citizenry and has mitigated potential political dissatisfaction. Much of this wealth derives from Qatar having strategically positioned itself as the world's largest supplier of Liquefied Natural Gas (LNG), thus making its per capita income the highest in the world (Kamrava, 2011). Qatar reached these high living standards after enacting many changes since its independence in 1971. Since then, its foreign policy has been directed at keeping a balance between itself and foreign players in order to obtain external protection (Barakat, 2014).

Considering the location of Qatar, Qatar is located between Saudi Arabia and Iran, the region's two competing hegemons, thus giving it the status of a buffer state. Its position between two large, belligerent states naturally creates a lingering threat for Qatar (Fazal, 2011). This explains why Qatar between 1972 and 1995 relied on Saudi Arabia for a security guarantee to protect it from potential spill-over from the Islamic revolution in Iran (Kaussler, 2015b). Theoretically, to protect themselves against hegemony following the Cold War, small states had the choice to either follow the bandwagon approach or the balancing approach. The balancing approach is

represented by the political, economic, and military gathering of a group of states to face the hegemon country's power. The bandwagon approach is a strategy in which the small state realises that the cost of opposing an adversary is greater than the cost of cooperating with them and, therefore, resolves to join the hegemon country.

As mentioned above, these perspectives were dominant immediately following the Cold War and are based on the assumption that small states rely on their neighbours for survival (Ghimire, 2017). As mentioned, this was applicable to Qatari foreign policy in the period between 1972 and 1995, at which time Qatar relied upon Saudi Arabia for protection and stability (Cooper & Momani, 2011).

The Iraqi invasion of Kuwait in 1990, and tension with Saudi Arabia and its opposition to any autonomy in Qatari policy, both exposed the vulnerability of Qatar (Kamrava, 2015). Indeed, the Iraqi invasion of Kuwait put its dependence on Saudi Arabia for security into question. Consequently, this sparked a search for a new defence alliance for Qatar (Barakat, 2014).

In 1995, Sheik Hamad bin Khalifa Al Thani became the new Qatari Emir. He shifted Qatar's foreign policy agenda away from the Saudi hegemony in the region. Furthermore, Qatar's geographical positioning between two competing countries, Saudi Arabia and Iran, has continuously forced Qatar to seek external protection. That is why Qatar in 1991 signed a military cooperation agreement with the USA (Khatib, 2013a). The agreement was developed in 2003 and included the creation of the Al Udeid military base, now considered to be one of the foremost overseas military bases in the Middle East, housing military personnel from the USA, UK, and various other allies (Kamrava, 2015). In making this alliance, Qatar guaranteed its security externally.

Since the year 1995, Emir Hamad has ushered in unprecedented reforms in politics, media and economy, during which he helped Qatar to grow into a regional leader, with the creation of the Al-Jazeera channel and through its mediation in regional conflicts (Barakat, 2014). Qatar's fast economic development has contributed to the emergence of Qatar as a distinct diplomatic power broker (Mohammadzadeh, 2017).

The regional changes in the 1990s also helped Emir Hamad bin Khalifa to make fundamental changes to Qatar and its foreign policy. The ending of the Iran-Iraq war was an opportunity to develop its North Field gas in Gulf waters. This, in turn, paved the way for Qatar to become the world's largest Liquefied Natural Gas (LNG) exporter by 2006, which has reflected upon Qatari foreign policy and its role as an international mediator (Khatib, 2013b). Being the world's largest LNG exporter provides Qatar with vast financial resources available to be used as financial incentives for peace and for hosting mediation talks (Kamrava, 2011).

Internally, the Qatar case exemplifies the importance of internal stability in international mediation, as Qatar did not get caught up in the Arab Spring revolutions that spread across the Middle East and North Africa in 2010-2011 (CIA, 2018). It can be argued that this stability is due in part to its vast wealth, advanced healthcare and educational system and developed infrastructure (Khatib, 2013b). Under Emir Hamad, Qatar succeeded in developing strong state institutions while maintaining strong social cohesion. Qatari domestic stability gives it the time and freedom to focus on and engage in international mediation (Barakat, 2014).

With regard to the connection between internal stability and mediation, the study suggests that the relationship between internal stability and mediation has a strong

positive correlation. In the context of leverage, incentives are one source of leverage to be used in mediation (Zartman and Touval, 1996). These incentives are based mainly on the economic resources of the state mediator and thus are logically tied to economic growth, which in turn requires political or internal stability (Alesina & Rodrik, 1994; Lindner & Strulik, 2004). Political instability, on the other hand, reduces economic growth (Alesina, Özler, Roubini, & Swagel, 1996; Huntington, 2006). Therefore, Qatari internal stability helps Qatar develop its economy and enjoy abundant economic resources and economic surpluses, which can be used as leverage for international mediation.

For Qatar, international mediation and the promotion of international peace is a core element of Qatari foreign policy. Qatari prioritises international peace and security and states that its foreign policy aims are to strengthen international peace and security by using peaceful resolutions (Nuruzzaman, 2015). International mediation was even officially included in the Qatari national constitution in 2003 and part of its foreign policy. In the Qatari constitution of 2003, article 7 states that" The foreign policy of the State is based on the principle of maintaining international peace and security by encouraging the settlement of international disputes by peaceful means, and supporting the people's right to self-determination and non-interference in internal affairs of the State, and cooperation with peace-loving nations" (The Qatari constitution, 2003, p. 2). The constitution suggests that its motivation for mediation is to strengthen international peace and security, which can be seen as a moral motivation. This is in line with Barakat's (2014) perspective which points to the moral and religious motivation of Qatar's mediation ambitions and cites the usage of verses from the Holy Quran to justify the claim (Barakat, 2014). In contrast, Khatib (2013) and Kamrava

(2011) assert that Qatari mediation is guided by self-interested motivation in order to protect Qatar from regional rivals.

Qatar mediation is often heavily influenced by Qatar's regional rivals. For example, throughout Qatari mediation in Yemen, Saudi Arabia was able to undermine Qatari's mediation strategy (Hansen, 2013), which is one of several concerns related to Qatari's efforts in international mediation. Saudi Arabia shares a long and porous border with Yemen. The country has followed the Houthi rebellion in Yemen with great interest and has, on several occasions, launched bloody assaults against the Houthis to defeat them. So naturally, Saudi Arabia disapproved of Qatari mediation in Yemen. Instead, they went to great lengths to undermine Qatari mediation efforts by pouring money into the Yemeni military and their allied tribes while simultaneously portraying Qatari mediation efforts as a proxy initiative of Iran. Additionally, in August 2010, during the Qatari mediation in Yemen, Saudi King Abdullah asked the Yemini president to add a condition granting the Saudis the right to be against any agreement with the Houthis(Hansen, 2013).

Kamrava (2015) asserts that Qatar, in its international mediation, bucks the trend of a low-leverage state by succeeding diplomatically and emerging as a regional powerhouse, going well beyond merely ensuring its survival and security (Kamrava, 2015). Qatar is one of the Gulf Cooperation Council (GCC) states which had the courage to involve themselves as mediators. Thus, the mediation role was restricted to the major regional powers such as Saudi Arabia and Egypt (Kamrava, 2011). Some scholars claim that Qatar, through its mediation in Sudan, has challenged

Egypt, which considers Sudan to be its backyard (Sudan Tribune, 2011b). The same

is true for Saudi Arabia's role in Yemen, during which it was wary of Qatari violation of Saudi influence there. Indeed, that is why the Saudi government undermined Qatari's attempts at mediation in Yemen (Hansen, 2013). Jasem (2012) confirms that throughout its international mediation, Qatar has never attempted to challenge the traditional leadership roles in the Middle East, including those held by Egypt and Saudi Arabia (Jasem, 2012).

The following section reviews the Qatari approach to international mediation. This approach consists of seven features represented in interest-based motivation, business diplomacy, public Diplomacy, Qatar impartiality, Personal Diplomacy, manipulation strategy, and Qatari leverage.

Interest-based motivation

Qatar's international mediation is driven by interest-based motivation. The Qatari policy seeks to ensure the survival of the state, Qatar's security and the achievement of international prestige (Kamrava, 2011, 2015). Survival strategy is essential in the case of Qatar, seeing as it is a buffer state located between two larger belligerent states, Saudi Arabia and Iran. Based on its location, Qatar is more likely than other states to experience state extinction and lives with a constant degree of the looming threat (Fazal, 2011; Jesse & Dreyer, 2016). Qatar is determined to maintain its security and stability, but its location is challenging for a small state with such unlimited ambition in its foreign policy. It finds itself on the Arabian Peninsula, defined by different political and military rivalries. By aiming to increase its international profile, Qatar aims to protect itself from the dangers of small state vulnerability (Cooper & Momani, 2011) and avoid what happened to Kuwait in 1990 (Rickli, 2016; Roberts, 2012).

Based on the need for security, some countries may involve themselves in international mediation either to avoid the danger of spill-over of nearby conflicts or to protect their own interests (Kamrava, 2011). Blanford (2008) shows that Qatar engages in mediation between disputants who are allies of Iran, such as Hezbollah and the Houthi. Qatar has attempted to mediate between the Houthis and the Yemeni government and between Hezbollah and the Lebanese 14 March Movement (Blanford, 2008). Khatib (2013) illustrates that Iran plays an essential role in those conflicts, as it is a prominent supporter of Hezbollah and maintains strong relations with the Houthis in Yemen, all of which affects Gulf security. Qatar is doing its best to prevent the conflict from spilling over while at the same time maintaining a cordial relationship with Iran, with whom it shares the world's largest oilfield (Khatib, 2013a).

Qatari security is connected to the branding of Qatar as an international mediator.

Qatari international mediation efforts are usually based on branding Qatar. Qatar aims to portray its image as a regional power, introducing itself as an impartial mediator seeking peace and stability in the Middle East and the international community.

Qatar in international mediation is exercising leverage on the international community to establish itself as a recognised brand and introduce itself as an international ally of the West (Peterson, 2006b; Roberts, 2012; Urichsen, 2014).

Qatar gains unique benefits by representing itself as an international ally of the West.

Qatar has achieved the security it needs to protect itself in the unstable region by hosting the al-Udeied US airbase (Khatib, 2013b; Peterson, 2006b; Roberts, 2012).

By being an international ally of the West, Qatar presents itself as a modern,

business-oriented state with the ability to compete in the international community (Roberts, 2012).

However, the Qatari mediation is not without fanfare, as this fanfare is important to Qatar in branding itself as a peace broker (Kamrava, 2011). Dickinson (2012) claims that the short-term result is that Qatar is able to fulfil its goals of branding itself as a peacemaker rather than focusing on achieving lasting peace in the countries it has targeted for international mediation. Dickson states that this explains Qatar's inability to ensure that agreements are implemented, as it lacks the diplomatic corps needed to sustain peace efforts after the agreement is reached. Dickinson bases his analysis on several cases representing the Qatari mediation experience to prove this. For example, Qatari mediation between the Yemeni government and Houthis in 2007 did not lead to a truce between the two sides. Likewise, the Doha agreements in 2008 involving Lebanon did not reduce violence and instability or even decrease tension between the rivals and did not offer peace to the country. Additionally, Qatari mediation between Hamas and Fatah in 2012 also did not lead to a long-run change or conciliation between the two parties(Dickinson, 2012).

Based on those examples, Dickinson (2012) claims that Qatari mediation did not change much in the conflicts where it was engaged. However, Dickinson's argument can be refuted by the fact that Qatar has objectively had significant success in its conflict mediation, though those successes have not necessarily brought closure in the form of conflict resolution. Mediation in conflicts, whereby disputants pledge to resolve the conflict, in practice does not necessarily lead to the resolution of the conflict (Kamrava, 2011). Moreover, international mediation contributes to the strengthening of the soft power of Qatar, which in turn contributes to the legitimacy and status of

Qatar in the international arena. That, in turn, furthers Qatari influence and security (Minich, 2015).

Generally, the Qatari foreign policy should not be understood only, in terms, of the Qataris' quest for security; rather, Qatari foreign policy should also be understood as status-driven. Mohammadzadeh (2017) confirms that Qatar is seeking to increase its influence in the international community. Qatar's ambition is to initiate the shaping of political change. That is seen demonstrated in Qatari foreign policy towards the Arab Spring as Qatar financially supported the rebels in Egypt, Libya and Syria (Mohammadzadeh, 2017). Ennis & Momani (2013) explain that the Qataris aim to extend their influence and become the leading player in Middle East politics, competing in mediation with Saudi Arabia, which has traditionally played the lead role in regional conflicts. Lacking neutrality in its intervention and mediation of regional conflicts and being biased towards the Lebanese March 14 movement in the Lebanese civil war, Saudi Arabia has opened the door for Qatar to fill the gap. Ennis & Momani (2013) claim that Qatar has seized this opportunity to successfully situate itself as an alternative to Saudi mediation in the Middle East and regional politics(Ennis & Momani, 2013). However, Al Qassemi (2011) confirms that Qatar has been careful not to cross the line regarding its relationship with Saudi Arabia and is careful not to oppose Saudi foreign policy, even regarding its domestic policy. Accordingly, Qatar supported the Gulf Cooperation Council (GCC) under Saudi leadership in the Arab Spring when the decision was made to guell the rebellion in the Bahraini uprising of 2011. Qatar also lent support during the Yemini uprising when the GCC suggested moderating the transition trajectory in Yemen and seeking negotiation instead of overthrowing the regime of its president, Ali Abdullah Saleh. He adds that in 2008

Qatari-Saudi relations were settled, and a rapprochement was reached. Al Qassemi argues that the settled relation was encouraged by the realpolitik of Qatar and an awareness of its limited influence in the Gulf area (Al Qassemi, 2011). In the same manner, Khatib (2013) illustrates that the relationship between Saudi Arabia and Qatar continued to oscillate between tension and cooperation until the Arab Spring in 2011 when the two countries shared the same instability concerns and feared chaos spreading into their own territories, which has been pushing them towards cooperation, although in some cases they still confront one another (Khatib, 2013a).

However, during the Arab Spring, Qatar moved away from its traditional foreign policy role as a mediator to embrace change in the Middle East and North Africa and support transitioning states. The Arab Spring era put Qatar's foreign policy into question, viewing it as an overreached policy. Qatar's new leadership, which came to power in June 2013, adapted by reverting to a more pragmatic foreign policy and addressing the fallout from its support for the Islamist movements in the region (Urichsen, 2014). The recent outbreak of diplomatic and political turmoil in June 2017 between key Gulf countries—namely Qatar on the one side and Saudi Arabia, the UAE and Bahrain supported by Egypt on the other—is the most serious since the formation of the GCC in 1981. One of the key accusations that have been levelled against Qatar by the other three fellow GCC countries is its alleged support for terrorism and its support for Islamist groups in the region, including in Libya (El-Gamaty, 2017).

In sum, all of the motivation sources discussed above fall under the umbrella of interest-based motivation. Through its international mediation, Qatar seeks to secure itself and brand itself as a peace broker.

Business Diplomacy

The second feature of the Qatari international mediation approach is business diplomacy. Business diplomacy helps to establish sustained, positive relations between the top executives or their representatives in one country and the foreign government representatives and non-governmental stakeholders in other countries, aiming to build and sustain legitimacy in a foreign business environment (Wolters, 2012; Kesteleyn et al., 2014).

In Qatar, business diplomacy plays a significant role in Qatari international mediation. Qatar uses business to drive diplomacy and, conversely, uses its diplomacy to protect its business interests (Gulbrandsen, 2010). Qatar uses its financial capacity to drive its diplomacy, which can be seen reflected in its international mediation policy (Group, 2009). In Qatar, there is no significant distinction between the income of the state and the income of the rulers in Qatar (Rathmell & Schulze, 2000). Indeed, business in Qatar is politics. Qatari international investment operates through its sovereign wealth fund (SWF), also known as a sovereign investment fund, a state-owned investment fund that invests in assets such as real estate (Kern, 2007). Furthermore, the Qatari economic system is characterised by state capitalism, where the state undertakes economic activities and deals with the means of production through the management of state-owned business enterprises. State capitalism helps the reshaping of international politics and the international economy by transferring large economic power to the central authority of the state (Bremmer, 2009).

For example, in addition to his ministerial positions, Sheikh Hamad is a member of the Supreme Council for the Investment of the Reserves of the State and the Ruling

Family Council. He is also vice-chairman and CEO of the sovereign-wealth fund Qatar Investment Authority (QIA), and chair its direct investment arm, Qatar Holding.

Furthermore, he is the chairman of QIA's wholly-owned real estate arm, Qatari Diar, while also sitting on the board of Qatar Energy and Water Company (Kerr, 2008).

Through business diplomacy, Qatar uses foreign policy to protect its investments. For example, Qatar started investment in Yemen's real estate sector in 2000 based on cooperation between a Yemeni government company (Shibam Holding) and Qatari Diar (a real estate company controlled by Shaik Hamad). At the time of Qatari mediation in Yemen, Qatari Diar announced a \$600 million project. In addition to this, several Qatari banks are active in Yemen (Group, 2009).

The same is true in Sudan, where Sheikh Hamad embraced his real estate ambitions to secure Qatar's business interests. Qatari Diar has \$400 million worth of residential and hotel projects in Khartoum. In addition, the banks have a very central presence. In Sudan, for example, one can point to the Qatar National Bank Al-Islami, which is entirely owned by Qatar National Bank (QNB). QNB opened its first branches outside Qatar in Khartoum around the time when Doha became involved with the Darfur peace process (Group, 2009). Qatar pledged a USD 2 billion investment to neutralise potential Libyan spoilers in Libya. Likewise, in South Lebanon, Qatar granted USD 300 million for reconstruction (Minich, 2015).

Finally, Qatar is concerned about energy security as the world's leading LNG exporter. With a large and increasing portion of its exports going to Europe through the Suez Canal, Qatar is particularly concerned about the security situation in the Gulf of Aden, Bab Al-Mandab, and the Red Sea, which may explain Doha's activist foreign

policy around the Horn of Africa, mediation in Sudan and Yemen, cultivation of good relations with Eritrea, and its long-standing interest in Somali stability (Gulbrandsen, 2010). Indeed, financial investment is a Qatari tool for international mediation (Kamrava, 2011). Additionally, success in international mediation is significant for Qatari investment (Kesteleyn, Riordan, & Ruël, 2014). For example, the success of Qatari mediation in Darfur offers the chance for Qatar to invest in Darfur and Sudan in general.

Qatar, with its enormous financial resources, can finance its mediation efforts both during the talks and after reaching agreements (Kamrava, 2011). It is essential to mention that Qatari business diplomacy in Qatari international mediation is coupled with their public diplomacy. Qatar uses investment to facilitate international mediation and vice versa; Qatar uses international mediation to protect its existing investment.

Public Diplomacy

The third feature of the Qatari mediation approach is public diplomacy. Public diplomacy, also called people diplomacy, refers to an international actor's attempt to manage the international environment through engagement with a foreign public (Cull, 2009; Nye, 2010). In other words, public diplomacy is the ability to affect others to obtain the desired outcome. Furthermore, it is a tool government uses to mobilize and attract the public to other countries rather than the institutions that govern them.

Broadcasting, subsidizing cultural exports and arranging exchanges are tools of public diplomacy used to attract the attention of the public in other countries (Nye Jr, 2008). Public diplomacy includes any government-sponsored effort that aims to communicate directly to a foreign public.

Munro (2019) suggests two types of public diplomacy. The first one refers to branding where the government seeks to improve its image at the international level.

Branding is intended to affect the foreign public's long-term perception. The second type of public diplomacy refers to political advocacy to facilitate rapid results. It is used to provide foreign support for immediate policy objectives (Munro, 2019). Public diplomacy is essential for small states. Peterson (2006) confirms that a small state under threat of absorption from its neighbours may develop a unique niche in any field that benefits the regional and international community by using public diplomacy. The small state can demonstrate to foreigners that it is more valuable or useful as an independent state rather than an absorbed one (Peterson, 2006a).

Qatar is an example; Qatar succeeded in developing its niche in international mediation to promote international peace and security and utilised public diplomacy to brand its niche. Qatar utilised its Aljazeera satellite channel as a tool of its public diplomacy to brand its niche in international mediation, which was established in 1996. Technically it is privately owned; however, the royal family financially support it.

Aljazeera has traditionally offered 24-hour broadcasting news in Arabic throughout the Middle East and since 2006, through the Aljazeera English Channel, has also offered the same service to the international community (Cooper & Momani, 2011).

Thus, Aljazeera has adopted two internal and external roles in nature. The internal aspect focuses on discussing controversial topics in the Arab and Muslim public environment. Meanwhile, the external role focuses on representing the Arab and Muslim perspectives on regional and international events to the world (Powers & Gilboa, 2007). Aljazeera plays a vital role in Qatari international mediation, where it helps Qatar to maintain lines of communication with different actors and serves as a bridge

between parties in mediation (Barakat, 2014). For example, when Israel attacked Lebanon in the war of 2006, Aljazeera championed Hezbollah, then Qatar went in to mediate in Lebanon (Hammond, 2014). This sort of media coverage is a staple feature of Qatari international mediation, despite some scholars' view that secrecy is one of the vital elements of any mediation effort.

Pulliam (2013) asserts that the most successful mediation is the one that is conducted in secret (Pulliam, 2013). Furthermore, Barston (2014) and Lieberfeld (2008) argue that the leaking of elements of secret negotiations creates difficulties because concessions made by one party are exposed, thus weakening its position. They add that in such instances, the credibility of the mediator may be put into question, or the media may present a misunderstanding of the negotiations (Barston, 2014; Lieberfeld, 2008). While publicity of the mediation increases Qatar's international profile, too much media focus on the mediation process may draw attention away from the developing mechanisms and long-term involvement in the conflict (Barakat, 2015).

While secrecy is the central pillar of the Norwegian international mediation, the Qatari international mediation approach is based on media coverage through the Aljazeera satellite channel. Qatar's mediation efforts have taken place in the spotlight of the media, where the local and regional media outlets are involved (Barakat, 2014). The Qatari diplomats often offer interviews to the media while the mediation process is still underway. Those diplomats usually portray the country's role in incredibly favourable, glowing terms. After the end of the mediation process, keenly recognizing the value of the advertising of their mediation efforts, Qatari leaders continue to refer to the Qatari mediation efforts as significant accomplishments, mentioning their

importance for regional stability. There is plenty available on Aljazeera and Qatar in the context of regional security/politics resources on mediation cases; this is Qatar's way of portraying what is referred to as niche diplomacy (Kamrava, 2011).

Qatar's impartiality

The fourth core feature of the Qatari mediation is Qatar's impartiality in mediation.

Qatar's preferred foreign policy method in international mediation is a tactic that avoids it having to take sides and thus allows it to maintain its position of "neutrality" and its status as "everyone's friend" (Khatib, 2013b).

Barakat (2014) and Kamrava (2011) confirms that Qatar is an impartial mediator with stability and experience in international mediation and attaining peace (Barakat, 2014; Kamrava, 2011). In 1995, the Emir Hamad Bin Khalifa Al Thani founded Qatari modern foreign policy. Three principal elements mark this foreign policy: pulling Qatar from being dependent on Saudi Arabia, making Qatar a peace mediator in the Middle East region, and finally, adopting high-profile diplomatic relations with all allies and avoiding conflict with adversaries. The Emir adopted those three elements in order to keep Qatar safe and secure (Nuruzzaman, 2015).

Regarding international mediation, Nuruzzaman (2015) assures that Qatar from the first moment adopted impartiality by implementing a strategy of "get to know all and act" (Nuruzzaman, 2015). Mitchell (2008) asserts that impartiality has always been a crucial component of Qatari international mediation (Mitchell 2008). Qatar builds relationships with all parties and invests those relations in mediation.

Considering the Qatari impartiality, one should consider two periods of time. From the year 2003 to the year 2011, when the Arab Spring began, Qatar proved its

impartiality as a mediator. However, throughout the years 2011 and 2012, amid the Arab Spring, Qatar took an aggressive stance against the violent oppression of protestors in Libya and Syria and supported the protestors against these two political regimes. The tools they used to support the rebels in these countries ranged from campaign contributions to the direct provision of weapons. It was a swift departure from Qatar's traditional role in international mediation as an impartial mediator (Barakat, 2012). In June 2013, the Qatari Sheikh Hamad announced the immediate transition of power to his son Emir Tamim. This may have been an admissions of error in Qatar's biased intervention in Syria and Libya and a return to Qatar's previous role as an impartial mediator (Hammond, 2014). The new Emir, Sheikh Tamim, seeks to restore the Qatari reputation as an impartial mediator (Minich, 2015).

This impartiality is upheld as Qatar's strong point. However, Qatari impartiality is not without critics, namely regional power such as Saudi Arabia. For example, Saudi Arabia has routinely criticised Qatar for its mediation in Yemen and has accused the country of bias towards Iran (Barakat, 2012). Kamrava (2011) presents an example of Qatari impartiality. He shows that in its mediation of the Darfur conflict, Qatar invited both disputants to negotiations in Doha, and before this step, the country's Minister of State for Foreign Affairs himself went to Darfur to witness the reality of the situation there. Those behaviours helped establish Qatar's unbiased image (Kamrava, 2011).

Barakat (2012) and Mitchell (2008) each have considered Qatar to be a neutral and impartial mediator (Barakat, 2012; C. Mitchell, 2003). Moreover, Barakat (2012) suggests that the impartiality of Qatar attracted the disputants to accept Qatar as a mediator in the first place. He adds that Qatar has diverse relations with many different

actors in the Middle East and the wider international community. Those relations led the disputants to trust Qatar and prefer it as an impartial mediator. He presents Lebanon as an example. He confirms that Qatar succeeded as a mediator in Lebanon and was a better choice for Hezbollah than Saudi Arabia and Egypt, neither of whom had any previous relationship with Hezbollah and would have almost certainly not been an impartial mediator (Barakat, 2012).

Some may claim that Qatar was not an impartial mediator due to hosting people at Aljazeera channel or Qatar hosting people, such as Muslim brothers affiliates, on the Qatari state. For example, on September 15th, 2004, the Aljazeera channel hosted Mohamed Al-Awa, the Muslim Ulama Secretary-General, on "bila hudod" (without borders). In the interview, which Ahmad Mansour conducted, Al-Away denied the Darfur Holocaust and blamed the rebels for the events taking place in Darfur. Many understood from this interview that Qatar, in this case, represented by the Aljazeera channel, was biased towards the AL Bashir government (Mahmoud, 2004). This argument could easily be refuted by reviewing Aljazeera satellite channel during and before the Qatari mediation in Sudan. One can see that Aljazeera conducted numerous programs and interviewed many people from Darfur, focusing on their suffering. Those people included the rebel leaders as well as the citizens of Darfur. That indicates that Aljazeera was eager to show both opinions. That helps to reject the allegation that Qatar was biased towards the Government of Sudan.

For the Muslim brotherhood affiliations, some may claim that Qatar is known for its support of the Muslim brotherhood. They present a key representative of the brotherhood, Yousuf Al-Qaradawi, as an example. Roberts (2012) claims that Yousuf Al-

Qaradawi has been settled in Qatar for many years and is an influential factor in Qatari personal politics (Roberts, 2012). Pulliam (2013) illustrates that Qatar's hosting of several Muslim Brotherhood leaders made it liable to accusations of bias in any mediation involving the Muslim brotherhood (Pulliam, 2013). However, the Qatari biased toward Muslim brotherhood may be not accurate. Qatar hosted a variety of exiles in addition to the Muslim brotherhood. Qatar hosted the Hamas leader, Khaled Mishaal, the Indian artist, M.F. Husain, and former Iraqi Foreign Minister Naji Sabri Al Hadithi. In addition, they hosted many of Saddam Hussain's family, the former president of Mauritania, Abbasi Madani, former FIS leaders in Algeria, prominent Libyan cleric Ali Al Sallabi, and former Knesset member Azmi Bishara. This diverse collection indicates a Qatari tolerance for exiles of all backgrounds (Roberts, 2014).

In sum, impartiality is a significant component in Qatari mediation. Qatar works hard to keep equal relations with all parties and invests this relation in any future mediation. Those relations are a by-product of Qatari personal diplomacy.

Personal diplomacy

The fifth main feature of Qatari mediation is personal Diplomacy. Personal Diplomacy means that the head of state/president or Foreign Minister decides to embark on diplomatic visits or meetings alone, rather than using an ambassador (Larres, 2002; Marsh, 2017). Qatar's impartiality combined with Qatari personal diplomacy is a potent combination that constitutes the country's trust-based leverage. Thanks to Personal Diplomacy in which the Emir and the Qatari foreign minister play the main role, Qatar usually succeeds in bringing disputants to the negotiation table and is often credited with the signing of a deal (Kamrava, 2011).

The Qatari mediation is based on the personal Diplomacy. For example, before the Qatari mediation in Darfur, the Qatari Minister of State for Foreign Affairs, Ahmad bin Abdullah al-Mahmoud, contacted conflict stakeholders, such as the USA (Blanchard, 2011). In the Qatari mediation in Lebanon, Hezbollah agreed on the terms of the agreement only after the Qatari Emir telephoned the Syrian president (Minich, 2015).

Understanding the role of personal diplomacy is helpful to understand The Qatari tribal monarchy's structure and history. Qatar has been led by the Al Thani family since 1825, with a few rival families such as the Al Attiyah. The Qatari tribal monarchy's structure is divided into rulers, movers, spacers, and tellers. The rulers are the Emirs, the central and sovereign person behind Qatari political decisions. The second group is the movers. The movers consist of individuals capable of strongly and directly influencing the Emir's political decisions. A mover example is Mozah bint Nasser Al Missned, the second wife of the previous Emir Hamad bin Khalifa and the current Emir Tamim bin Hamad. The third group is spacers, members of the large, powerful families which are given space for relevant foreign policy decision making within a quite narrow subfield of foreign policy, an example being the first chairperson of Aljazeera Sheikh Hamad bin Thamer Al Thani. The last group is the tellers, individuals with access to the movers or the Emir. They do this via informal networks, such as the so-called Wasta, or through business associations, sport and media organizations and friends and families. This group can focus on issues of interest to the movers and rulers. The Department of Foreign Affairs in Qatar is considered to be no more than a logistics institution with a limited influence on politics. Its main role is as a tool to enhance the ruler's efforts (Hansen, 2013).

Considering Qatar's international mediation, it is based on an elite with intense personal engagement. This elite consists broadly of the Emir, the prime minister and the foreign minister (Barakat, 2014). Those three persons are responsible and are mainly involved in the mediation process. The Emir is the crucial person in this structure who advances the mediation process. The Emir and the foreign policy officials have often acted as well-intentioned and well-informed mediators intending to turn an intractable dispute into a win-win scenario (Ulrichsen, 2014). They both rely on a combination of instinct, charisma and wealth to push through an agreement (Barakat, 2012). They also use this combination to form new networks. They depend heavily on these networks and know many officials in Western and Eastern countries on a very personal level (Dickinson, 2012a). This network is unique in that it is based on a form of nepotism, that is, the Wasta network, the religious organizations and family ties that exist inside the decision making elites and structures (Urichsen, 2014).

Considering Norway's international mediation, it is based on the combined efforts of the Norwegian Ministry of Foreign Affairs and Norwegian NGOs, used primarily for networking (Kelleher & Taulbee, 2006). Unlike Norwegian international mediation, Qatari international mediation efforts are personal and based on the elite's decision-making structure. Qatar's role in international mediation is based on a combination of its financial resources and its personal involvement with the world's heads of state (Kamrava, 2011).

However, Qatari personal diplomacy is a source of strength and weakness in Qatari mediation. The strength can be seen in focusing on the elite; Qatari international mediation is characterised by the ability to act quickly, the availability of money, and the modest size of the decision-maker structure (Kamrava, 2011). Additionally, the

strength can also be seen in the Emir and the well-informed elite of Qatari, which has been remarkably effective in international mediation and has often succeeded in bringing the disputants to the negotiation table to move forward with peace talks. Qatari personal diplomacy helps in establishing trust, managing partners, facilitating negotiations, managing the potential spoilers, as in the case of Lebanon and Syria (Roberts, 2011) However, the weakness of this personal diplomacy and its personalization of the mediation process lies in having this decision maker structure that might not necessarily contribute to international mediation success. Based on this structure, Qatar might succeed at bringing disputants to the negotiation table and even get them to sign a deal, but they may fail to sustain the process to completion (Kamrava, 2011).

Additionally, the weakness of this personal diplomacy and its personalization of the mediation process lies in the lack of formally trained personnel. This reflects negatively on the mediation process and is seen reflected as a decrease in the contextual knowledge of the conflict. Qatar has been criticized for its lack of knowledge regarding best-practice strategies in mediation, post-settlement implementation and cease-fire monitoring (Roberts, 2011). Personal diplomacy, in some instances, hindered the cultivation efforts of deeper institutional expertise in mediation strategies (Roberts, 2011).

The centralization of the decision making in the hands of the Emir and his elite negatively affects Qatari mediation. It leads to quick reactions and decisions regarding Qatari international mediation and, additionally, an absence of a foreign policy infrastructure and a lack of professional capacity to follow up mediation efforts and implementation(Khatib, 2013b).

Personal diplomacy is not without its shortfalls, especially on the implementation level and the follow-up mechanism (Kamrava, 2011). The lack of an elective follow-up mechanism for monitoring implementation is a key concern because it leads to a breakdown of Qatari efforts in international mediation.

For example, Yemeni members of the implementation committee met with rebel leaders but made little progress because they operated in a vacuum. There were no regular contacts between signatories and Qatari officials and no formal mechanism to address agreements (Group, 2011). Qatar needs professional personnel who can translate the Qatari success in mediation to long term policies that support Qatari mediation on the ground (Kamrava, 2011). Minich (2015) recommends that personal diplomacy be preserved but enhanced by a trained diplomatic apparatus experienced and knowledgeable administrators who can provide the resources needed for conflict resolution (Minich, 2015).

Roberts (2011) explains that the transition of power to Sheikh Tamim created a large vacuum: namely, the two critical roles in Qatari mediation that were previously filled by Emir Hamad and the Qatari ex-foreign minister, Bin Jassim Al Thani. Qatari mediation had heavily relied on them (Roberts, 2011).

In sum, Qatari personal diplomacy is a source of both strength and weakness in Qatari mediation. However, it is considered a distinct and essential feature of Qatari mediation and is the key to success for Qatar in international mediation.

A manipulation strategy

Another prominent characteristic of Qatari mediation is the usage of a manipulation strategy. For the manipulative mediator to make a promise or threat, the mediator

must first bring the disputants together using a facilitation strategy and /or structuring a proposal using a formulation strategy. By using the facilitation strategy, the mediator contacts the disputants, gains their confidence, transmits messages between the disputants and allows the interest of all disputants to be discussed. While in formulation strategy, the mediator helps formulate a framework for an acceptable outcome and makes sustentative suggestions and proposals. On the other hand, the manipulative mediator holds the disputants at the negotiation table, presses the parties, threatens the withdrawal of resources, and finally threatens with punishment and mediation withdrawal. Additionally, employing the manipulation strategy, the mediator has the ability to use a vast arsenal of incentives and ultimatums to change the way the issues are framed and affect the perspective of the disputants towards those issues. The choice of strategy is dependent upon the resources the mediator possesses (Bercovitch & Houston, 2000).

Qatar utilises its oil and gas financial power in international mediation. For example, Qatar used its wealth in Darfur negotiations to encourage the disputants to sign an agreement (Roberts, 2015). Qatar has thus far practised power-based mediation. Its interventions heavily depend on Qatari's capacity to wield its vast financial resources to offer incentives for conflicting parties to come to an agreement, which indicates that Qatari mediation is more advanced than pure mediation (Moran, 2009). Indeed, the promise of future investment and development is a genuine component in the Qatari mediation approach.

By introducing incentives to the disputants to incentivise peace, the mediator presents a more attractive option (Carnevale, 1986; Zartman and Touval, 1996; Beardsley et al., 2006). According to the theory of international mediation, Qatar is

employing the manipulation strategy in international mediation by introducing those incentives to the disputants. It is worth mentioning that the manipulation strategy allows the mediator to employ the carrots (incentives) to promote the disputants to reach an agreement (Beardsley et al., 2006) and the stick (pressing) in order to increase the costs of non-agreement (Carnevale, 1986).

From its mediation in Lebanon, Sudan, and Yemen, it can be concluded that Qatar seems to have developed its strategy based around facilitation, formulation and manipulation strategy. The Qatari manipulation takes the form of incentives. In other words, Qatar's strategy is manipulative in the sense of offering rewards and promising financial benefits to disputants in return for resolving the conflict (Kamrava, 2011; Urichsen, 2014). For example, Qatar pledged USD 300 million for construction in Lebanon to promote a power-sharing agreement between Hezbollah and the government of Fouad Siniora. For this reason, some scholars have labelled Qatari engagement in international mediation as chequebook diplomacy (Rabi, 2009).

However, using incentives as carrots helps bring the parties to the negotiation table and reach an initial agreement. However, this effect is typically limited to the short run. In granting incentives, the mediator causes the disputants to focus on short-term gains rather than addressing the underlying roots of the conflict (Barakat, 2012). Indeed, the Qatari mediators work based on their framework by adopting a strategy that involves manipulating financial leverage as incentives for the disputants to encourage them to agree.

The Qatari Leverage

One of the main components of the Qatari mediation approach is its leverage. Qatari leverage in international mediation can be broken into three categories: the Qatari economic resources or incentives, personal diplomacy, and the cultural and religious ties between Qatar and the disputants.

Qatar's wealth is considered power-based leverage and is heavily used in Qatar's international mediation (Kamrava, 2011). Indeed, the primary source of Qatari power-based leverage is represented in its material incentives to the disputants if they reach an agreement (Kamrava, 2011). Qatar has conducted huge investments in infrastructural development projects in the countries where it has played the role of mediator. For example, Qatar is one of Southern Lebanon and Sudan (Kamrava, 2011). In practice, the Qatari experience in mediation proves that the power of rewards or incentives plays a significant role in its mediation success (Minich, 2015). Unlike most low-leverage state mediators, Qatar is able to invest its financial power to bring parties to an agreement (Minich, 2015).

Given its position as the state with the highest GDP per capita in the world, its massive financial resources help Qatar to finance its peace talks and peace agreements (Ulrichsen, 2012; Urichsen, 2014). Due to its enormous wealth, Qatar as a mediator focuses on providing material incentives in its mediation. Incentives are among five types of leverage resources distinguished by Zartman and Touval. As mentioned above, those five sources are represented in persuasion, extraction, termination, deprivation, and gratification (Bien et al., 2000).

The material incentives that the disputants receive from the mediator can result in different outcomes (Böhmelt, 2010; Schrodt & Gerner, 2004). Those material incentives could take one of two forms: either direct cash payments or investments in infrastructure projects, both of which are associated with conflict settlement (Kamrava, 2011). Qatar has heavily invested in the countries in which it has played the role of mediator. For example, Qatar made 2 billion USD investments for the Darfur region and pledged 500 million USD in reconstruction aid in the Saada Province. Additionally, it provided 300 million USD for reconstruction in Lebanon (Minich, 2015). Additionally, the Qatari security arrangement with the United States provides Qatar with greater autonomy in its foreign policy, which is considered another source of Qatari power-based leverage (Krane & Wright, 2014).

Qatar deploys its immense wealth or economic resources combined with its Personal Diplomacy. Qatar's Personal diplomacy is also considered a source of leverage, creating a potent combination. Personal diplomacy, represented in the Emir and the Qatari foreign minister, usually plays a significant role in Qatari mediation. By using personal diplomacy, Qatar usually brings the disputants to the negotiation table, and personal diplomacy encourages the disputants to sign a deal (Kamrava, 2011). However, Qatari wealth and Personal Diplomacy are not the only sources of Qatari leverage in international mediation. Sharing cultural and religious ties is a type of leverage that Qatar has in its international mediation arsenal. The cultural and religious ties are considered leverage, which helps build trust between Qatar as a mediator and the disputants. Notably, most Qatari mediation attempts have dealt with regional, Muslim and Arab societies. Qatar's leverage is based on Qatar's cultural and religious traditions, and the disputants share. In terms of mediation in Arab and Muslim

societies, Qatar has an advantage over any western mediator. In its mediation in regional conflicts, Qatar is able to use a culturally appropriate framework and avoid using an imported" one-size-fits-all" approach that may not match the culture and norms of the disputants (Mac Ginty, 2008a). Cultural ties are considered leverage for Qatar because the disputants are sure that Qatar understands their needs well. Those cultural ties strengthen the position of Qatar as a mediator. For example, Qatar used those cultural ties to be a mediator in the Darfur conflict. Qatar proved to the Darfuris that Qatari would be an honest and credible mediator who would not oppose them. Leverage from the cultural and religious ties can be seen in the mediation between Fatah and Hamas and the Sudanese conflict. Qatar shares a cultural tie with both the Palestinians and the Sudanese. Those cultural ties are based primarily on Islam's religion and a shared ethnic group; Palestinians, Sudanese and the Qatari are all predominantly Arabs.

In summary, Norway and Qatar are wealthy countries with low leverage for international mediation. There is, however, a distinct difference between the Norwegian model of international mediation and the Qatari approach in international mediation. Unlike Norwegian international mediation, which is normative, Qatari mediation is interest-motivated and directed at gaining international prestige to enhance the country's regional standing. Additionally, Qatar uses its international meditation attempts to strengthen its regional position. Equally important, the building block of the Norwegian mediation is Track One and a Half Diplomacy. One and a Half Diplomacy is represented in the strong relations between its NGOs and the Ministry of Foreign Affairs and uses its NGOs as a tool to build trust with disputants, make connections among the local community and initiate the international mediation process. In contrast, the

building blocks of Qatari international mediation are its finance and personal diplomacy. Qatari diplomacy in international mediation is sometimes referred to as the "cheque book" approach because it relies on its considerable financial resources to obtain a final peace agreement (Kamrava, 2011; Rabi, 2009). Qatari networking is based on the "Wasta-network, and family ties within the decision making elite (Urichsen, 2014).

Based on the above discussion, it was expected that the research would show that Norway is a norm-based mediator acting as a pure mediator, employing a facilitation strategy, and known for its impartiality in international mediation. On the other hand, the research was expected to show that Qatar is an interest-based mediator that acts as a power mediator employing a manipulation strategy and not paying attention to impartiality. This study challenged those expectations and found different facts about Norway's and Qatar's behaviour in international mediation. The following four chapters illustrate the actual findings of the Qatari approach and the Norwegian approach in international mediation.

CHAPTER FOUR: CASE STUDY ONE: NORWAY AND THE OSLO ACCORD 1993

Introduction

On 13 September 1993, the World witnessed the signing of the Oslo Accord, the product of a series of secret negotiation rounds in Oslo between Israeli representatives and the Palestinian Liberation Organization (PLO) representatives, with the presence of Norway as a mediator. Both sides had managed to agree on a Declaration of Principles calling. It includes mutual recognition of Israel as a state and the PLO as a representative of the Palestinian people (Waage, 2005). The need of both the Israelis and Palestinians for a solution prompted Norway's mission to bring the parties together at the negotiating table. The international, regional, and local changes determined the specific time for Norway to become a mediator. On the Israeli side, they desired to find an alternative to suppress the Palestinian Intifada and help Israel to avoid further criticism by the international community for its deterrent and bone-cracking policy against Palestinians in the West Bank and Gaza Strip (Waage, 2005). On the Palestinian side, the political, economic and financial weakness of Yasir Arafat and the PLO helped Norway in particular to become a mediator. Additionally, the Norwegian Track One and a Half diplomacy was leveraged to build trust between Norway and the two parties and became the main reason for the disputants to accept Norway as a mediator.

Specifically, Norway's leverage in mediation between the Israelis and the PLO resulted from the trust and confidence built on the personal relations between FAFO and the disputants. In the past, scholars have argued that Norway had neither a carrot nor a stick to use as leverage and push the disputants towards an agreement.

Also, the literature shows that Norway usually employs a facilitation strategy in its mediation that is coupled with its impartiality (Ingebritsen, 2006; Kelleher& Tabulbee, 2006; Ford, 2000). However, this new research has revealed a different profile from the known image of Norway as an impartial mediator who employed the facilitation strategy.

Overall, the main argument in this case study is that the Norwegian mediation between the Israelis and the PLO, two disputants who have power disparity, brings both the Norwegian facilitation role and the Norwegian impartiality into question. This case study shows that Norway used a facilitation strategy and a manipulation strategy. Norway played the role of facilitator until May 1993 and from that time on has acted as a manipulator and a power mediator by pressing the weaker party (the Palestinians) towards achieving agreement, exploiting the desperate need of the PLO for any solution. . In its mediation between the strong Israeli side and the weak PLO, Norway played a manipulative role, taking advantage of Arafat and the PLO's weakness and practising psychological pressure on Yasir Arafat and PLO. Accordingly, Norway acted as a power mediator, practising a manipulation strategy on the PLO. At the same time, Norway itself was manipulated by the stronger disputant, the Israelis. Israel took advantage of the eagerness of Norway to reach an agreement and achieve the status of peacemaker on the international political map. Additionally, the power disparity between the Israelis and the PLO did not help Norway be an impartial mediator. As a lower-leverage state, Norway failed to properly manage the power disparity between the Israelis and the PLO.

In sum, the role of Norway in the Oslo-backed channel was a complicated one. In its mediation between the strong Israeli side and the weak PLO, Norway played a

manipulative role, taking advantage of the weakness of Arafat and the PLO. Additionally, Norway was not an impartial mediator in its role in Oslo secret channel. Those arguments are explored in this chapter, divided into three main parts: a brief about the Palestinian-Israeli conflict, the Oslo backchannel, and the Norwegian international mediation approach in the Israeli-Palestinian conflict.

The Israeli-Palestinian conflict

This section is not a detailed historical analysis of the Israeli-Palestinian conflict because the tale of Palestinian-Israeli hostilities is beyond the scope of this thesis. For this thesis, only a brief description of the conflict is provided, focusing mainly on the circumstances that led to the Oslo Accord. The purpose of this section is to direct the reader's attention to those situations which put the two disputants in need of a solution and encourage both of them to accept Norway, a low–leverage state, to act as mediator. Without a doubt, the Israeli-Palestinian conflict has been one of the most complex conflicts in the world for several decades (Tessler, 2009). The conflict is a struggle between two distinct ethnic groups with competing claims on the same land and a conflict of national identity and self-determination (Gerner, 2018). Both the Palestinians and the Israelis put forth strong arguments based on history and religion to support their claims as both have deep roots in the region (Pfeffer, 2014). However, each of them denies the other's right to the land and has been caught in a cycle of denial for many years (Elon, 2015; Halevi, 2018).

The conflict's roots date back to the late 19th century and the Jewish migration to Palestine. This immigration was strengthened in 1917 when the British Foreign Minister, Belfour, announced the Belfour Declaration, in which he pledged British support

for the Jews in establishing their Jewish national home in Palestine, a territory which would remain under the British Mandate until 1922. The declaration would later support and increase the Jewish immigration to Palestine after the extermination of six million Jews by the Nazis during the Second World War (Helmreich, 2017).

Then, on 29 November 1947, the General Assembly of the United Nations adopted a resolution to partition Palestine into two states: a Jewish state and an Arab state. Straightaway, the Arab leaders opposed the resolution. Eventually, Israel was announced as a state on 15 May 1948 (Farsakh, 2017). However, the Jewish entity in Palestine on 15 May 1948 was not a fragile infant but rather a modern, western industrial society. Israel was a state in every meaning of the word, even before declaring itself as a state in 1948 (Alkhaldi, 2010). Nearly two decades later, in the Arab-Israeli War of 1967, Israel took over the West Bank, Jerusalem and the Gaza Strip (Schulze, 2013).

For this reason, the Palestinians have found themselves in a never-ending struggle to return to their occupied land. Accordingly, and to serve this purpose more effectively, the Palestinian Liberation Organization (PLO), and National Council (PNC) as the PLO's supreme body, were formed in 1964 to lead the Palestinian struggle against Israel. The largest faction in PLO emerged as Fatah, which Yasir Arafat headed (Shemesh, 2012). Ever since the Camp David agreement between the Egyptians and the Israelis in 1978, there have been several efforts from the Palestinian side to create a Palestinian state. However, the Israelis would never give a state to the Palestinians on the occupied land of 1967(Anziska, 2018).

Towards Oslo back channel

By the late 1980s and leading into the early 90s, there occurred important changes in the international context, the regional context, the Israeli national context, and the Palestinian national context (Pruitt, 1997). Firstly, on the international level, the Soviet Union collapsed in 1991, resulting in a new unipolar world order with the USA as a hegemon country. This allowed the United States to be the sole player in the world, including the Middle East (Dougherty, 2001). The collapse of the Soviet Union in 1991 and the 1991 Gulf war strengthened US influence in the Middle East and deeply affected the regional players (Aruri, 2003). In the meantime, Yitzhak Shamir, the Israeli prime minister, understood the importance of US support to absorb immigrants who came from the former Soviet Union. The Americans were aware of the importance of timing in launching a peace conference. Given the collapse of the Soviet Union, the USA planned to find a solution to the Israeli-Palestinian conflict by launching the Madrid peace conference at the end of October 1991, which later was moved to Washington. The negotiations in the Madrid conference took two tracks: a bilateral track and a multilateral track. The bilateral track was between Israel, the Palestinian-Jordanian (joint delegation), Syria and Lebanon. Meanwhile, the multilateral track dealt with five common issues that concerned all parties: water, the environment, arms control, refugees, and economic development (Bercovitch, 2011b).

The PLO was excluded from those talks, and the Palestinian delegation was represented by residents of the Gaza strip and West Bank. Moreover, the Palestinian delegation was not independent; it was under the Jordanian umbrella. However, the Palestinian delegation was in direct contact with PLO in Tunisia. The failure of the Madrid peace conference and Washington talks was a motivation for the Israelis and the

Palestinians to initiate a secret channel of talks and was the inspiration for Norway to sponsor those secret negotiations (Bercovitch, 2011a; Shlaim, 2005).

The second event, which paved the way to the Oslo secret channel, was on the regional level in the Middle East. It was the Iraqi invasion of Kuwait and the Gulf war in 1990-1991. The invasion of Kuwait by Iraq and PLO's support of Saddam Hussain weakened the PLO's position and severed its access to financial resources which it is used to receive from Arab countries such as Kuwait and Saudi Arabia (Bruck, 1996; Lundberg, 1996; Makovsky, 2018; Peres & Landau, 1995; Pruitt, 1997). Yasir Arafat was weakened by lending support to Saddam Hussein against the United States-led coalition. The Gulf War weakened the PLO and made it ready to make peace through a third party who could easily approach the Israelis (Egeland, 2008). Arafat's support for Saddam Husain had left the PLO without the economic and diplomatic leverage that it used to enjoy from Arab countries. This support deprived the PLO of its political backing and cost the PLO much of its financial resources. Moreover, it marginalised the PLO and isolated it (Qurei, 2006).

The Gulf war demolished the Arab consensus on Palestine; consequently, the Palestinian question was removed from the top of the official Arab agenda (Aruri, 2003). Furthermore, Kuwait, the Gulf countries, and Egypt joined USA and Israel in a campaign aiming to demonize Arafat and delegitimize the PLO. Conversely, the Gulf War strengthened Israel and imposed its diplomatic framework in any expected negotiations. Moreover, after the Gulf War, Israel was convinced that the Arab recognition of Israel as a state was a precondition to any negotiations between the Israelis and the Palestinians (Aruri, 2003).

Indeed, the collapse of the Soviet Union and the Iraqi war were a window of opportunity for peace in the Middle East (Aruri, 2003). On the national level, the first Palestinian intifada has widened this window. The Palestinian first intifada (Arabic for uprising) took place in December 1987 across the West Bank and the Gaza Strip. This intifada witnessed a growing influence of the Hamas movement, which posed a threat to both Israel and Yasir Arafat. Likewise, Israel was concerned that Hamas would cooperate with militant Iran and create а serious military threat Israel (Makovsky,2018;Rabin,1993). Similarly, Arafat's concern was that Hamas would become an alternative leadership to the PLO (Corbin, 1994b). The PLO lost much support to Hamas in the Gaza strip and West Bank during the first Intifada (Agha, Feldman, Khalidi, & Schiff, 2003; Qurei, 2006).

Indeed, the first Intifada in the Gaza Strip and the West Bank returned the Palestinian question to the forefront of the Arab and international agenda (Qurei, 2006). It is considered a significant leap in paving the way to the Oslo agreement. Additionally, it provided prestige and political influence on the internal leadership in Gaza Strip and West Bank, such as Haider Abdel Shafi and Hanan Ashrawi. However, the founding of Hamas during the Intifada created a challenge for the PLO, who was politically and financially suffering in the wake of the Gulf war, because of support to the toppled Saddam Hussain by Yasir Arafat.

More importantly, Intifada boosted the need for the Palestinian leadership for negotiations. Additionally, the Jordanian departure from the Palestinian scene in July 1988 forced the Americans and the Israelis to deal directly with the Palestinian leadership (Hirschfeld, 2014). It was clear that any negotiations should be coordinated with PLO,

whether through the Palestinians in the Gaza Strip and West Bank, through the US-PLO dialogue, or the Egyptians (Hirschfeld, 2014).

Similarly, during the First Palestinian Intifada, Israel faced international condemnation for its crackdown on the Palestinians. As a result, Israel was under pressure, realizing that it was time to initiate secret peace talks with the PLO. Additionally, Israeli leadership thought that the PLO leaders would control the Palestinian intifada better than the Israeli army and could relieve Israel of the occupation burden, which would pave the way to normalizing the relations with Arab countries, which could not materialize without the agreement of the Palestinians (Brynjar,1998; Waage,2000). Moreover, Israel realised the extent to which Arafat was weakened in the wake of the Gulf War, knowing that Arafat would enter the negotiations with very few options and less influence. Israel wanted to capitalise on this weakness. This may help explain why Israel avoided talks with Arafat in 1979 (Brynjar, 1998; Waage, 2000).

As has been noted, these substantial changes in circumstances on three levels— national, regional, and international— created an environment in which a low-leverage state such as Norway could take advantage and fill the role of mediator in this complex conflict. According to the Palestinian perspective, the Norwegian setting would bring the marginalised PLO back to centre stage and would give Arafat complete and direct control over the Palestinian side in the negotiations (Ashrawi, 1996; Mahmud'Abbas, 1995; Qurei, 2006). It is worth mentioning that Arafat asked the Norwegian government several times during the 1980s to convey a message to Israel's foreign minister, suggesting Norway as a negotiation place, but Israel refused due to its conviction to not talk directly to the PLO (Pruitt, 2008).

In hopes of paving the way to negotiations with Israel, Yasir Arafat announced in 1988 that the PLO accepted the UN Resolutions 242 and 338. This announcement was understood as an indirect recognition of Israel because it granted secure and recognised boundaries to Israel (Pruitt, 2008). Simultaneously, a new Israeli government was formed in November 1988. It was a national unity government headed by Yizhak Shamir. Peres filled the role of finance minister, and Rabin served as defence minister (Hirschfeld, 2014). Rabin's previous experience had taught him that treating the Palestinian people with violence led nowhere, and he recognised that the time was ready for peace. His immediate priority was to put an end to the intifada. Rabin's thought was strengthened by the end of the war between Iraq and Iran in 1988 as he was afraid that Iran and Pakistan would develop a nuclear weapon capability, which could undermine the Israeli deterrence power. For this reason, he was determined to make peace with both the PLO and Syria (Hirschfeld, 2014). Making peace was not only in the mind of Rabin but was also a dream for Shemon Peres (Hirschfeld, 2014). Indeed, peace in the Middle East was a dream of Shimon Peres and his small team. Indeed, peace in the Middle East was a dream of Shimon Peres and his small team, Yossi Beilin (the deputy of Shimon Peres), Uri Savir (Director General of the Israeli foreign ministry), and Avi Gil (Director General of the Israeli foreign ministry). The defeat of the Likud government in the Israeli Knesset in the summer of 1992 gave a chance for Peres's dream to come true (Peres & Landau, 1995). The Israeli Labour party had come to power in 1992, and its leaders were aware of the PLO's severe political and financial weakness, especially after losing the support of Saddam Hussein and the Gulf funding (Ashrawi, 1996; Bentsur, 2001; Corbin, 1994b; Haykal, 1996; Murphy, 1999; Peres & Landau, 1995; Savir, 2010; Stein, 2005).

Contributing to the situation, the PLO broke off the Washington talks in mid-December 1992. This action was in response to Israel's expulsion of more than 400 Hamas affiliates to Lebanon (Hirschfeld, 2014). The PLO was feeling the pressure, as they were excluded from the bilateral negotiation in Washington, and they were afraid of losing their primacy over Hamas (Beilin & Simpson, 1999).

Rabin and Peres strained to find any way to convince Arafat to support the negotiations in Washington fully. The Israelis were aware that their recognition of the PLO would help the PLO gain acceptance at an international level and help it to regain its legitimacy. Israel was also aware of the importance of this card and was sure that it could play it to reach the desired conclusion in an agreement (Hirschfeld, 2014).

It should be noted that most American initiatives were based on obligating the participants to accept the UN Security Council Resolutions 242 and 338 (Hirschfeld, 2014), which was considered a concession Israel wanted to avoid. This was the reason for Israel's preference for a low-leverage state such as Norway to be a mediator. In the same direction, the PLO had chosen Norway as neither possessing political ambition nor significant influence and therefore not likely to arouse American sensibility (Abbas, 1995). Additionally, Yasir Arafat's suggestion of Norway as a mediator was due to the strong relationship between Norway and Israel, the strong ties between Norway and the US, Norway's ties to the EU (Waage, 2007).

In brief, From the Palestinian perspective, the Norwegian mediation would bring the marginalised PLO back to centre stage and would give Arafat complete and direct control over the Palestinian side in the negotiations (Ashrawi, 1996; Lieberfeld, 2008; Abbas, 1995). From the Israeli perspective, Israeli engagement in the negotiation with the PLO would prevent Hamas domination of the Palestinian state (Beilin &

Simpson, 1999). Under those circumstances, early in December 1992, Ashrawi and Hussaini, with the help of the Norwegians, arranged a meeting between Hirschfeld and the head of the Palestinian delegation, Abu Alaa, in London, where the negotiations took place that day (Hirschfeld, 2014).

In sum, the changes of the circumstances on the international level, regional level, Israeli national level and the Palestinian national level paved the way to Oslo's secret channel.

The Norwegian role in the Oslo secret negotiations

First, it is essential to mention the development of the relationship between Norway and the Israelis on one side and between Norway and the Palestinians on the other. It is equally important to consider that Norway has been a religious country rooted in conservative Christianity. Norway has a state church, the Evangelical Luther Church. The Norwegian generations were raised on tales of the Bible which portrayed Palestine in their mind the land of the Jews, not the land of the Arabs (Waage, 2000). The Norwegian Christian community used to see Israel through the eyes of religion and marked its existence as fulfilling the prophecies of old. Moreover, on the political level and in the eyes of the Norwegian labour party, Israel used to be seen as evidence that the socialist paradise had come true. This may explain the Norwegian's initial support of Israel when it was established in 1948.

Indeed, the relationship between the Israelis and the Norwegians began with establishing the Israeli state. Norway felt sympathy for the Jewish people who had suffered so terribly during the German Nazi era. At that time, the UN's Norwegian delegation-supported creating a new Jewish state. Accordingly, the

Norwegian labour government established strong ties with their Israeli counterparts (Waage, 2000).

The Norwegians had operated in this capacity even before the founding of the state of Israel in 1948. Norwegian politicians, particularly from the labour movement and Christian circles, fostered friendly relations with Israeli colleagues and counterparts. The Norwegian Confederation of Trade Unions had also cooperated closely with the Israeli trade union movement, Histadrut (Yossi Beilin & Simpson, 1999; Corbin, 1994b; A. Elon, 1993; Pruitt, 1997). This support of Israel lasted throughout the 1960s. However, these amiable relations were not the same after 1967 when Israel occupied the West Bank and the Gaza Strip (Egeland, 2008). In the war of 1967, the Norwegians began to view Israel differently, as a strong country, not the victim of Arab aggression. Indeed, two main factors made the Norwegians change their image about Israel and feel sympathy towards the Palestinians: the war of 1967 and the Norwegian peacekeeping forces serving in Lebanon in 1978(Egeland, 2008). After the war of 1967, Norway decided to take a balanced position to the Middle East conflict (Qurei, 2006).

The image of Israel in Norway began to change in 1978 as thousands of Norwegian peacekeeping forces served in Lebanon, and Norwegian journalists and politicians began to form negative views on Israel (Damen, 2013; Pace, 2018). They all contributed to a gradual shift in the position of Norway towards one of Palestinian sympathy (Egeland, 2008). Since that time, Norwegian-Israeli relations have fluctuated continuously (Pace, 2018). Notably, the Norwegian sympathy towards the Palestinians began to increase. In that period, the

Norwegian-Palestinian relationship began to take root (Corbin, 1994a; Mahmud'Abbas, 1995).

Through the 1970s and the 1980s, the Norwegian labour governments started to build a relationship with the PLO (Egeland, 2008). Indeed, the relations between the Norwegians and the Palestinians started in the 1970s, when many Norwegian solidarity groups were established to support the Palestinian cause. In this atmosphere, Foreign Minister Knut Frydenlund met unofficially with the chairperson of the Palestinian Liberation Organization, Yasir Arafat, in Tunis in December 1982. Consequently, the PLO opened a representative office in Oslo (Corbin, 1994a; Mahmud'Abbas, 1995). Additionally, for the first time, Norway started to consider the Palestinian refugee question as a political question and not a humanitarian one. Additionally, the Norwegian Ministry of Foreign Affairs started to recognise refugees as Palestinians, as opposed to Arab refugees (Waage, 2000). This indicates that the Palestinian issue was considered independently and not as an Arab issue. Accordingly, Norway supported the UN resolution in 1974 that enabled Yasir Arafat to address the UN General Assembly (Yossi Beilin & Simpson, 1999; Qurei, 2006). In 1987, Stoltenberg, the Foreign Minister of Norway, established a Norwegian group whose aim was to stimulate peace mediation between the Palestinians and the Israelis. This group included Jan Egeland, Mona Juul, Terje Larsen, and Marianne Heiberg, the wife of Johan Holst (Yossi Beilin & Simpson, 1999; Qurei, 2006). These individuals contributed to building confidence between the two opposing parties. This confidence was the building block in the Norwegian mediation between the Israelis and the Palestinians. Norway's

close relations with the Israeli side made Norway an intriguing mediation prospect for the PLO. Along the same lines, the Norwegian's direct contact with Yasir Arafat helped to offer Israel a backchannel to negotiate with the PLO in the Oslo secret channel (Mahmud Abbas, 1995; Makovsky, 2018).

The story of the Oslo secret channel began in 1992 when Terje Rød Larsen, the director of the Norwegian Institute for Applied Social Science, FAFO, and his wife Mona Jull, a Norwegian diplomat, welcomed a delegation of Palestinians from Tunis led by Abu Ala, Arafat's senior economic adviser. Abu Ala raised the issue of a possible dialogue with Terje Larsen (Waage, 2000). In the meantime and after talking for hours with Abu Ala, Larsen was convinced that FAFO might facilitate an Israeli–Palestinian meeting (Kelleher & Taulbee, 2006).

In the same year, Larsen met Yossi Beilin, who knew him through Larsen's research project in Gaza and the West Bank to examine Palestinian living conditions. In that meeting, Larsen raised the possibility of a Norwegian mediation role between the Israelis and the Palestinians. Beilin, in his turn, promised to be an operator for peace in case the labour party won the election in June 1992 (Aggestam, 1996; Corbin, 1994a; A. Elon, 1993; Mahmud'Abbas, 1995; Makovsky, 2018; Peres & Landau, 1995).

On 19 June 1992, there was a meeting in East Jerusalem between Larsen, Faisal Husseini, Yossi Belein and Yair Hirschfeld. However, this meeting led nowhere (Qurei, 2006). Later, on September 10, 1992, by permission of the Norwegian Prime Minister Stoltenberg, Egeland met with Yossi Beilin, who became the deputy foreign minister in the new Israeli government. In the

presence of Larsen, Mona Jull, Shlomo Gur and Hirschfeld, they discussed the possibility of secret talks in Oslo and assured the Israelis that the Norwegian Ministry of Foreign Affairs would stand behind FAFO in sponsoring the secret talks. The meeting was so secretive that even the Norwegian ambassador did not know about it (Egeland, 2008).

Beilin was not ready for direct contact with the PLO at that time. As a result, he suggested his friend, professor Yair Hirschfeld. Accordingly, Jull and Larsen arranged appointments between Abu-ala and Hirschfeld, while Beilin had already met Faisal Alhussaini in Jerusalem (Beilin & Simpson, 1999). Larsen arranged a meeting between Abu Ala and Yair Hirschfeld in London. In that meeting, Hirschfeld assured Abu Ala that he only represented himself and not the Israeli government (Qurei, 2006). Having encouraged Beilein, two weeks later, Larsen Travelled to Tunis. Beilin was hoping to know Larsen's evaluation of the people in Tunis. Larsen met Yasir Arafat, who was disappointed after the defeat of the Swedish socialist government in the poll, a government that would have sympathised with the PLO. Arafat himself told Larsen that Norway should take over the Swedish role. By this, he was referring to the role of gobetween. Moreover, Arafat asked Larsen to convey this message to the Norwegian government. Arafat was aware that the Norwegian socialist government had run a balanced policy in the Middle East and was trusted by both sides (Corbin, 1994b; Qurei, 2006).

Larsen concluded that the PLO was the only body that could negotiate on behalf of the Palestinians. When he went back to Oslo he, Jull and Egeland phoned Beilin to advise him to go further with the London meetings and to create a secret channel in Oslo (Corbin, 1994b). They feared that direct official contact with the PLO would contravene Israeli law, which at that time prevented any official contact with the PLO (Egeland, 2008). For this reason, Yossi Beilin suggested that two Israeli academics initiate the negotiation with PLO. Those academics, Yair Hirschfeld and Ron Pundak, were already in contact with Abu Ala and Hassan Asfour (Beilin & Simpson, 1999; Mahmud'Abbas, 1995; Peres & Landau, 1995; Waage, 2005). Consequently, a secret channel between Israel and the PLO was agreed upon. It would be initiated in Norway under FAFO's auspices, while FAFO, the Norwegian Institute for Applied Social Science, offered a cover of plausible deniability (Qurei, 2006). The FAFO project in Gaza and West Bank was to serve as a cover for the secret talks. This suggestion was supported by Foreign Minister Stoltenberg (Egeland, 2008). As a result, the Oslo back channel opened in January 1993, with the two Israeli academics, Yair Hirschfeld and Ron Pundak, on the Israeli side and the PLO's representatives, Abu Ala', Hassan Asfour, Maher Kurd and Mohammed Abu Koush, on the Palestinian side. Though the Palestinian participants were all PLO officials, the talks at this stage were entirely informal, aiming to build a bridge, create informal political contacts, and find out if anything could be done to break the stalemate in Washington's negotiations. The only Israeli official aware of the meetings was Beilin, who was deeply involved from the beginning and watched progress closely, with Peres and Prime Minister Yitzhak Rabin informed only after the first round. The Norwegian government had approved and eagerly backed the

talks. Deputy Foreign Minister Jan Egeland and Mona Juul briefly attended the first meeting (Waage, 2005).

Ultimately, the culmination of the first phase of the secret meetings in Norway was drafting the Sarpsborg Declaration of Principles (DoP), completed in March 1993. Eventually, by the end of August 1993, the Accord formulated the principles of peace. Later, the Declaration of Principles on Interim Self-Government Arrangements (DoP) was signed at the White House in the presence of leaders from all around the world (Beilin & Simpson, 1999).

The Norwegian approach in the mediation of the Palestinian-Israeli conflict

The Norwegian motivation

In its mediation between the Israelis and the Palestinians, Norway succeeded in building its international mediation model's fundamental pillars or features, which paved the way for further attempts at international mediation. The following section will discuss this case further by looking at Norwegian motivation, Norwegian leverage, Norwegian strategy, Norwegian impartiality, and finally, whether Norway was a pure or power mediator.

As mentioned in chapter three, the Norwegian aim in international mediation has been to set normative standards in the international community (Björkdahl, 2013). AS mentioned in the previous chapters, Norway, like any other state-mediators, Norway does have an interest in international mediation. This interest is presented in having influential status in the international community or international relations. At the same time, Norway was seeking to

promote peace in the international arena is one of the normative standards that Norway has developed to make the world a better place. For those reasons, Norway has been engaged in international mediation. However, Norway succeeded in balancing its interest and its norms. According to this vision, it has been supposed that the Norwegians have no direct and specific self-interest in their attempts at international mediation. Generally, Norway has an interest in the mediation itself, which is presenting in increasing its status and being influential in the international arena, using its norms as a tool in international mediation. Norway is distinguished by its norms, which is why it uses them in international mediation. Accordingly, Norwegian mediation has been characterised as norm-based. According to Hellmüller et al. (2015), the mission of a norm-based mediator is not only to bring violent conflicts to an end but also to integrate values, human rights, justice, and other norms into their overall strategy (Hellmüller, Federer, & Zeller, 2015).

Norway was willing to promote its image as an international peacemaker, taking advantage of having a positive reputation with no colonial past (Waage, 2007). In their mediation between the Palestinians and the Israelis, the Norwegians seized the opportunity when the PLO asked them to mediate and make peace between the two disputants (Qurei, 2006). However, it is important to confirm that Norway's engagement in international mediation has always been important for Norway's self-perception, as its engagements in international mediation promote the Norwegian national image of moral power and norm entrepreneur. Indeed most Norwegians believe in the necessity of the Norwegian role to promote international peace (Kelleher & Taulbee, 2006). Through

its international mediation, Norway can promote peace and gain a reputation as a peacemaker (Höglund & Svensson, 2009).

Opinions differ on the issue of Norwegian motivation in its mediation in Oslo's secret channel. There are three arguments related to their motivation at that time. The first argument asserts that Norway had no interest in its mediation beyond promoting peace between the Israelis and the Palestinians. Mahmoud Abbas (1995) asserts that the Norwegians' negotiation process cannot be explained by a desire for political credit or even international reputation. He adds that their hard work and worries reflect an inner motivation and genuine desire to bring about peace in the Middle East. Therefore, he says, they feel disappointed when the negotiations reach a deadlock (Mahmud Abbas,1995). In the same vein, Singer suggests that Norway has no interest in the mediation between the Israelis and the PLO. He explains that Norway is a small country far away from the Middle East. Additionally, Norway has no Jewish lobby. The Norwegians are neither Muslims, nor Jewish nor Arabs (interview with Singer, 2019).

The second argument is represented in what Hassan Asfour says. Hassan Asfour, one of the principal Palestinian negotiators in Oslo, has a different opinion than both Abbas and Singer. Asfour asserts that Norway had interests in the mediation between the Israelis and the Palestinians in the negotiations that led to the Oslo Accord in 1993. He argues that the Norwegian status in the international community changed after the Oslo agreement. Its position at the international, Arab and Palestinian levels was not the same before and

after the Oslo agreement. After the Oslo agreement, many Norwegians occupied high-ranked positions in international organizations such as the UN, thanks to the Oslo agreement. Most of the Norwegians who participated in the mediations occupied high-ranking positions at the international level. "Anyone who says something different than this, he is wrong", Asfour says (interview with Asfour, 2018).

Hassan Asfour did not mean that Norway had received direct benefits at the economic level. However, he claims that Norway did receive benefits on the political level, which reflects on the economic level. He says, "On the political level, the world witnessed how a small state like Norway was able to resolve one of the world's most complicated conflicts in political history" (interview with Asfour, 2018). Asfour assures that Norway was rewarded for being a mediator between the Palestinians and the Israelis" based on this, Norway was rewarded for its relations with the Arab world. All of this reflects on its economic interests as it helps Norway to open up new economic horizons", Asfour says (interview with Asfour, 2018).

Any country in the world, especially the size of Norway, is generally looking for ways to be visible in the international arena, reflecting on its relations with other countries. This might help the country build stronger economic relations. It is not only Hassan Asfour who confirms that Norway benefitted from its mediation in the Oslo secret negotiation, but also the Norwegian professor Hilde Waage. Waage agrees with Hassan Asfour by asserting that Norway was seeking a visible place in the international arena. Waage says," for Norway,

and it was very essential to gain visibility on the international stage and show that small Norway could do something" (interview with Waage, 2019).

The third argument stands between the first argument and the second one. This argument is contradicted. An example of this argument is represented in what Mohammad Abu-Koush says. Mohammad Abu-Koush is one of the principal Palestinian negotiators in the Oslo secret channel. Unintentionally, he puts himself in a contradictory position when describing the Norwegian motivation in the Oslo secret channel. On the one hand, he asserts that "Norway was eager to reach a peace agreement between the Israelis and the PLO because this would put Norway on the international map as a peacemaker" (interview with Abu-Koush, 2019). On the other hand, he was assured that the Norwegians have no interest in their involvement in the Oslo secret channel. He says, "the Norwegians were interested in nothing" (interview with Abu-Koush, 2019).

Based on the three arguments, it is concluded that the results of the interviews for this case study agree with what this thesis finds. As mentioned, Norway does have an interest in international mediation "increase international status or influence" while it plays the role of norm entrepreneur. This thesis agrees with Abbas when he describes the Norwegians hard work in their negotiation as not a desire for political credit but a desire to bring peace in the Middle East. It is expected that the Norwegians do their best to achieve agreement between the Palestinians and the Israeli to achieve their ultimate interest, "Increasing international status". There is no contradiction between what Abbas says and what this thesis finds.

In the same vein, what Singer says contradicts what this thesis finds. Singer justifies his claim that "Norway has no interest in mediation between the Palestinians and the Israelis". He justifies his claim by presenting some important facts such as Norway is away from the Middle East, no Jewish lobby in Norway, and finally, Norway is not a Muslim country. Unfortunately, Singer's justifications are not enough to prove that Norway has no interest in the mediation between the Palestinians and the Israelis. On the one hand, Singer is right that Norway does not have a specific interest when in specific mediation this conflict. However, Norway ambition was to be influential status in the international arena. No other conflict can serve Norway ambition more than the protracted conflict between the Israelis and the Palestinians.

Hassan Asfour and Hilde Waage were so straight when asserting that Norway has an interest in international mediation between the Palestinians and the Israelis. However, this does not mean that Norwegian mediation was not a norm-based mediation. The Norwegians involved themselves in the mediation process to promote peace between the two parties and norm entrepreneurs in the international community.

In brief, Norway is a state that is actively working to promote peace in the international community and that by doing so, it is making the world a better place to live. However, as a low-leverage state, Norway was eager to find its place in the international arena. By mediation, a dispute as famous as the Israeli-Palestinian conflict, one of the world's most famous conflicts, especially after signing the Oslo agreement, Norway proved itself on the international political stage as a peacemaker.

The Norwegian leverage

The Norwegian leverage is represented in four elements: the Norwegian Track One and a Half diplomacy, the extensive personal contacts on both the Palestinian and the Israeli side that FAFO, the Norwegian economic resources have developed, and finally, the needs of the disputants. Those elements are overlapped connected, and it is not easy to study them separately. The Norwegian model in international mediation is characterised by Track One and a Half Diplomacy. This is displayed clearly in Norway's mediation in the Oslo secret channel. Norwegian NGOs played a unique role, fronted by the Institute for Applied Science (FAFO) in cooperation with the Norwegian government, represented by the Norwegian Ministry of Foreign Affairs. The Track One and a Half Diplomacy was the key to success in the Oslo secret channel. The Oslo secret channel was initiated through a project. This project focused on the economic and social conditions of the Palestinians in the occupied territories. The project was conducted by FAFO and was funded by the Norwegian ministry of foreign affairs (Qurei, 2006). Thus, FAFO came to play an essential role in the Oslo Accords. The mediation between the Palestinians and Israelis was facilitated by the Norwegian Ministry of Foreign Affairs and FAFO. It was a combination of formal and informal mediation processes (Aggestam, 2002). Formally, the Norwegian Ministry of Foreign Affairs provided the financial resources and the logistical aid for the meetings and, informally, FAFO provided the flexibility and the needed environment of secrecy for the talks to occur and helped in building trust and confidence between the two negotiated teams (Egeland, 1999). More importantly, the purpose of the participation of FAFO was that it provided the

deniability element, which was necessary for the early stages of the negotiations (Siniver, 2006).

The Norwegian NGO's role was not restricted to initiating the Oslo secret channel. Through Track One and a Half diplomacy, Terje Larsen, with the cooperation of the Norwegian ministry, facilitated the negotiations process between the Israelis and the Palestinians (Larsen, 2019).

FAFO and the Ministry of Foreign Affairs extended until signing the Oslo Accords in 1993. Both FAFO and the Ministry played an essential role in upgrading the level of the negotiations from non-official (academic) to an official. Through April and May 1993, Terje and Mona Jull made hundreds of telephone calls to Jerusalem and Tunis informing them of the other sides' questions, worriers and propositions until Israel finally Israel agreed to upgrade the talks to the official level with a guarantee from Norway to keep the talks secret (Egeland, 2008). By using Track One and a Half Diplomacy, the Norwegians succeeded in convincing the Israelis to upgrade their representation in the Oslo secret negotiations. In June 1993, the official negotiations kicked off with the joining of Joel Singer, who was appointed as a legal advisor for the Israeli foreign ministry and was Rabin's man in the Oslo negotiations (interview with Gur, 2019 and interview with Singer, 2019).

Through Track One and a Half diplomacy, Terje Larsen with the cooperation of the Norwegian ministry facilitated the negotiations process between the Israelis and the Palestinians

The second source of the Norwegian leverage in the Oslo secret channel is the extensive personal contacts on the Palestinian and the Israeli sides that FAFO has developed. FAFO was instrumental in building strong relations with the Israelis on one side and the PLO on the other. FAFO effectively facilitated the mediation process between both sides, led by Terje Larsen, who worked as director of FAFO and initiated the Oslo backchannel. Larsen developed extensive personal contacts on both the Palestinian and the Israeli side (Bandarage, 2011).

Through the aforementioned project, Larsen succeeded in setting up a web of communication both with the Israelis and the Palestinians, which resulted in the creation of the Oslo channel (Qurei, 2006). Singer mentions the important role of Terje Larsen: "Terje Larsen was the one who attracted the Norwegian government to the possibility of mediation between the Palestinians and the Israelis. Terje Larsen got the Norwegian foreign minister involved indirectly in the secret talks", Singer says (interview with Singer, 2019).

Terje Larsen asserts that he was not representing Norway when he started contacting the Palestinian and Israeli leaders. While representing FAFO, he worked closely with Palestinians such as Faisal Hussaini Fathi Arafat and Israeli leaders such as Yossi Beilin. He worked with both sides via FAFO's project to improve living conditions in the Palestinian territories. FAFO, led by Terje Larsen, succeeded to develop extensive personal contacts on both the Palestinian and the Israeli sides (Larsen, 2019; Singer, 2019; Gur, 2019). Terje Larsen started the secret talks between Yossi Beilin and Faisal Hussaini, and they were facilitated by FAFO, according to Larsen (interview with Larsen, 2019). He adds, "at the outset it was not the foreign ministry which initiated

the talks; it was basically myself and my capacity as the director of FAFO"(interview with Larsen, 2019).

While the extent of the personal contacts on both the Palestinian and the Israeli sides was considered a source of leverage for Norway in its mediation, the Oslo secret channel. However, there is no consensus on the positive impact of those relations. The Israelis consider those relations as a strong point. Yoel Singer, the Israeli negotiator, states that "the Norwegian closed relation with both sides, the Palestinians and the Israelis, is considered a very strong contribution to the Norwegian mediations", Singer says. He adds, "they were neither considered to be pro- Israelis nor pro-Palestinians, they were friendly to both sides, and they were able to simply serve in a way that was quite objectively sympathetic to both sides" (interview with Singer, 2019).

Singer explains that "based on those personal relations developed by FAFO, a trust was created between the Norwegians with Palestinians and the Israelis. The trust is considered part of Norway's leverage in its mediation between the Israelis and the PLO (interview with Singer, 2019). Using similar words, Yossi Beilin, the deputy of the Israeli foreign minister, describes Norwegian relations with both sides. Beilin confirms that the Norwegian leverage is reflected in their relations with both sides, the Israeli and the Palestinian. The Norwegian readiness and the Norwegian initiative to help in peace-making are the main factors that enabled Norway to succeed in mediation between the Israelis and the PLO in the Oslo secret channel (interview with Beilin, 2019).

In contrast, a senior Palestinian officer considers the close relationship between the Israelis and the Norwegians to be the main reason Norway was biased toward Israel (interview with a senior Palestinian official, 2019).

The third kind of leverage that the Norwegians possess is their economic resources. Norway has little influence on world politics and has limited military power, being a loyal member of NATO with close relations with the USA (Waage, 2007). However, Norway is a rich country with a strong economy based on fish, oil, and a budget surplus, and it is known for the generous donations and humanitarian aid that it grants as economic assistance to developing countries (Waage, 2007). Norway chose to utilise its economic resources to promote peace in the international arena by two ways. The first by funding project conducted by the Norwegian NGOs in the disputants' countries. Second, by funding the mediation cost such as meetings of the disputants.

Norwegian finance is considered leverage for Norway's international mediation in the Oslo secret negotiations. Norway promised Yaser Arafat and the PLO to contribute the money needed to build the Palestinian entity (interview with Waage, 2019). Likewise, in the Oslo back channel, Norway covered all the expenses and costs of hosting the Oslo secret negotiations; the flights, the hotels and even paid salaries for the two Israeli academics, listing their names on the FAFO payroll for deniability.

The fourth type of leverage is need-based leverage. As Zartman asserts, the need of the disputant can become leverage to the mediator (Zartman,1996).

Though Israel rejected Norwegian mediation when Arafat originally suggested it in 1979, they agreed in 1992, when Abu Alaa raised the issue of possible

Norwegian involvement with Terje Larsen (Waage, 2002, 2005). Israelis and Palestinians changed their minds and agreed to negotiate and seek a solution for their conflict as both sides needed an end to the hostilities. Both sides needed a solution, which explains their acceptance of Norway as a mediator. This desperate need for a solution is considered leverage for Norway, a low-leverage state.

For the Palestinians, the political and economic weakness of Yasser Arafat and the PLO after supporting Sadam Hussain and the PLO urgently needed a solution that would help it regain its legitimacy in the international political stage. The year 1992 was the right time to clinch a deal with the PLO because of the political and financial difficulties it faced at the time (Peres & Landau, 1995).

Arafat found out that the Norwegian mediation would bring the marginalised PLO back to centre stage and give him complete and direct control over the Palestinian side in the negotiations—(Ashrawi, 1996; Lieberfeld, 2008; Mahmud'Abbas, 1995).

Israel needed the Norwegian mediation to avoid the pressure from Washington negotiations based on the UN Resolutions 242 and 338. Additionally, engaging the PLO in the negotiations would prevent Hamas domination of the Palestinian cause (Beilin & Simpson, 1999). Israel also needed Yasir Arafat to save the face of Israel in front of the international community by controlling the Intifada and resistance movement, including Hamas and Islamic Jihad. The changes in the local, regional, and international contexts proved that the

window of opportunity was there and created the right environment for the Oslo channel at that time, the time was ripe for clinching a deal.

Mediation Strategy

Norway introduces itself as a facilitator mediator, employing a facilitation strategy in international mediation. As a low-leverage norm-based mediator, it is expected that Norway would employ a facilitation strategy. The Norwegian role was built on procedural and communicative strategies and not on manipulation due to limited Norwegian leverage over the two parties (Siniver, 2006). The Norwegian role in the secret channel was to play the role of bringing the parties together, promoting trust, and explaining difficulties the parties faced, but not taking a position on the substance of the talks or suggesting a route that the talks should follow (Corbin, 1994b).

While there is evidence supporting the claim that the Norwegian role is a facilitation one, there is enough evidence to prove that Norway was a manipulator and used a manipulation strategy. It is essential to assert that Norway used a spectrum of the three mediation strategies, facilitation, formulation, manipulation, in different stages of the mediation processes. Indeed, Norway played the role of a facilitator at the first phase of the negotiations. However, Norway acted as a manipulator in the second phase of the negotiations.

The turning point in the Norwegian position began in May 1993, when Johan Holst was appointed as a new Norwegian Foreign Minister and decided to play an active role in the negotiations (Waage, 2005). According to the senior Palestinian official, it is seldom highlighted in writings about the Oslo talks is that

the original 'deal' that the Palestinians had negotiated throughout the first five months had been an entirely different arrangement of what ended up in August as the Declaration of Principles. During the first five months, the Oslo track The negotiations were supported by Shimon Peres, the foreign minister. The prime minister, Isaac Rabin, was informed but refrained from extending his support, as he had continued to prefer the ongoing bilateral negotiations in Washington. Only after Rabin lost confidence in the possibility of achieving a deal with the Washington Palestinian team did he join Peres in supporting the Oslo talk and sent his representative, Mr Joel Singer, to Oslo in early June.

During the months that preceded June, the deal had been the creation of a PISGA: the Palestinian Interim Self-Governing Authority, a one-level council of 33 members representing the West Bank and Gaza Strip, that with run the PISGA throughout the five-year interim period, the PLO leadership and institutions and organs would remain situated and active in Tunisia and other Arab countries throughout the interim period and up until the conclusion of the negotiations on the permanent status issues. When Singer, Rabin's representative, joined the talks in early June, he practically took over the Israeli negotiating team, scraped and cancelled the various documents on the PISGA, and presented a new deal: that Arafat, the leadership, the PLO and its institutions would enter the oPt and run the autonomy, with new institutional and security arrangement for this new deal. Singer gave the Palestinian team dozens of questions on which he requested answers from Arafat interview with a Palestinian Senior official, 2020).

Based on these changes in the Norwegian position, the secret negotiations could be divided into two stages. The first stage extended from the beginning of the negotiations to May 1993, when the Norwegians employed the facilitation strategy. The second stage took place in May 1993, when Norway used a manipulation strategy. This change in the Norwegian position was confirmed by Shlomo Gur, the Shimon Peres office manager. According to Gur, the Norwegian role changed in May 1993 and shifted to one of active mediation when the Norwegian Foreign Minister, Holst, joined the negotiations as a direct participant (interview with Gur, 2019).

The Norwegian used the facilitation strategy to relieve the tension between the two sides. They did this notably when they reached a deadlock, leaving them to talk together, while the Norwegians would trail behind without interfering in their talks. The Norwegian's role in the negotiations was to offer a friendly environment between the two disputants and to reconcile the different points of view for both sides (Mahmud'Abbas, 1995). Additionally, the Norwegians worked hard to clear the obstacles and misunderstandings that existed between the two parties (Jones, 1999). The meetings were informal to permit the exchange of information between the disputants (Corbin, 1994a; Makovsky, 2018; Pruitt, 1997).

The Norwegian facilitation strategy is also presented in Norway's role consisted in getting the parties together, booking flights and places, offering a friendly environment for the negotiations, and keeping the negotiations secret (Waage, 2007). At the beginning of the negotiations, the Norwegian primary mission was to facilitate the travel and meetings of the two parties to bring

them together (Beilin & Simpson, 1999). The Norwegians were keen to develop confidence between the Palestinian and the Israeli teams and encourage personal relationships. For example, the Norwegians encouraged the parties to eat together at the same table with the Norwegian team. The Norwegians did their best to create a human relationship between the two parties by offering opportunities for personal and non-official talks outside the formal context. As a facilitator, the Norwegian team did not directly engage in the negotiations or talks between the disputants most of the time. Mostly the team's role was restricted to initiating the meetings, leaving the two parties to talk in private, giving advice and offering the best atmosphere for the meetings (Qurei, 2006). To emphasise the facilitation role of Norway, some interviewees even avoided the usage of the word "mediation" when describing the Norwegian role in the Oslo back channel and insisted on instead calling it "facilitation" (interview with Abu-Koush, 2019; Y. Singer, 2019).

Joel Singer claims that the Norwegian role was that of a facilitator and no more than a facilitator. He explains that the Norwegians were not even in the room during negotiations; their role was limited to providing accommodation, finding venues for the meetings, and providing security for the negotiators. Additionally, they did all they could to offer a stress-free environment for the negotiators: "they were not involved in any real mediation in the sense of ok, this is the draft we propose to you, we will go now from one side to the other, we will come back with other's views. I did not even give them a copy of the draft. They were not able to participate in any issue related to drafting" (interview with Singer, 2019). Singer explains that their role was to encourage the two

teams not to give up with phrases like "do not let it break down, we are almost there". The Norwegians offered atmospheric support, but not really mediation in the normal sense, Singer says. Singer explains that the Norwegian style of mediation was different from the mediation of the USA, who usually presented the draft for an agreement and asked the negotiators to give their comments. Singer asserts that "nothing like this happened with the Norwegians" (interview with Singer, 2019).

Singer shows that Norway had no way of exerting pressure on either side. He says "unlike superpowers like the United States or even regional powers like Egypt which also mediated between the Israelis and the PLO in various phases of the negotiations, Norway is a small nation far away; they have no stick, no carrots", Singer says, (interview with Singer, 2019).

Shlomo Gur stated that Norway's role was not mediation but facilitation (interview with Gur, 2019). Gur meant that Norway's role was to bring the two parties together, facilitate discussions, and finance the meetings (interview with Gur, 2019). Gur built up his claim that Norway was a facilitator, not a mediator (manipulator), on the fact that the Norwegians, Mona Juul and Terje Larsen, were not in the meeting room when the two parties met (interview with Gur, 2019). Gur talks about the first stage of the negotiations before Holst joined the negotiations in May 1993. At that stage, Norway was using a facilitation strategy.

Most people who have been interviewed in this study defend Norway's role in the Oslo secret channel as being that of a facilitator, not a mediator. The interviewees might not have been aware that facilitation is one strategy out of the three mediation strategies. Reviewing what they said, it is clear that they refer to an approach in which the mediator deploys manipulation strategy in his mediation. In literary terms, facilitation is a type of mediation.

In the same vein, Yossi Beilin asserts that "the Norwegians did not interfere in the content of the negotiations. However, they did whatever they could do in order to assure that the negotiations continued by telling each party do not give up, you have to talk again, you have to get back to the negotiation table etc", Beilin says (interview with Beilin, 2019).

Beilin assures that The Norwegians did whatever they could to ensure that the atmosphere was friendly and conducive to an agreement. Beilin explains that the Norwegian role was represented in financial terms and preparing the agreements' technicalities that could not be dismissed. Beilin adds, "one should not get caught in a limbo between the role of a facilitator and the role of a mediator. Carter was a mediator in the talks between Israel and Egypt, but Norway was not a mediator between Israel and the PLO" (interview with Beilin, 2019).

Mohammad Abu Kush, a Palestinian negotiation member in Oslo secret channel, agrees that the Norwegian role in Oslo secret channel was that of a facilitator: "Norway was an excellent facilitator", he says (interview with Abu-Koush, 2019). With those words, Abu-Koush describes the Norwegian role in Oslo secret channel. He built his claim on the fact that there were no Norwegians in the meeting rooms, and the role of Norway was to convey messages between the Israelis and the PLO representatives. Indeed, this is the role of facilitation which Norway role at the first stage of the negotiations.

There is no doubt that those interviewees are right when they claim that the Norwegians played the role of facilitator. However, it is important to mention that this role changed after May 1993. Since May 1993, it started the second phase of Norwegian mediation where Norway started to play the role of manipulator.

In the second phase, Johan Holst, the Norwegian new foreign minister had the clear intention of playing an active and direct role in the negotiations between the Israelis and the PLO. He ,himself, attended the fifth round of talks from the 8th to 9th of May 1993, even before the Israelis upgraded their representation in the negotiations (Waage, 2005). In this phase, Holst decided to make the Oslo channel the main official channel of negotiations between the Israelis and the PLO (interview with Waaage, 2019). In the second stage, Norway gave away its facilitation role and played a new role that exceeded facilitation.

Shlomo Gur, an Israeli official, mentioned Norway's pressure in the secret negotiations to reach an agreement. However, he asserts that "Norway was pushing both sides to reach an agreement, and this was in the later stages of the negotiations" Gur says (interview with Gur, 2019).

Norway psychologically manipulated Yasir Arafat because Norway was aware of the weakness of Yasir Arafat, because of his huge mistake in supporting Saddam Hussain. Arafat felt isolated in the Arab World and many Palestinians had been expelled from the Arab States, mainly the Gulf. This is affirmed by

Beilin who says, "Arafat lost his financial support. In order to get back to the Arab world and in order to open the door to the United States, he agreed to think on what he had not agreed to before" (interview with Beilin, 2019).

It is worth mentioning two things about the Norwegian manipulation. Firstly, Norway manipulated Yasir Arafat (the weak side), not Israel (The strong side). Secondly, Norwegian manipulation had two forms; material manipulation and immaterial manipulation. The material manipulation is represented in Norway's financial support to help the PLO collect after reaching an agreement. The immaterial manipulation is represented in the psychological manipulation of Yasir Arafat.

Some examples could prove the Norwegian manipulation of Yasir Arafat. One of the critical issues was the extra corridor. Yasir Arafat demanded an extra corridor to connect Gaza with the West Bank. However, Holst had a different opinion.

Waage (2019) assures that in the second phase, the Norwegians psychologically manipulated the weakness of Yasir Arafat, putting pressure on him to accept things that he could not have accepted in a normal situation, such as taking the idea of the extra corridor off the table and being content only with the idea of safe passage. Waage explains that in that meeting, Holst told Arafat that Israel would never accept it, even if it were simply a promise of safe passage (interview with Waage, 2019). The trip that Holst made by visiting Tunisia to meet Yasir Arafat was an exception to the facilitation role of the Norwegian mediation. Their goal was to create peace between the Israelis and the

Palestinians and to create a new international role for Norway. Waage asserts that Holst warned Yasir Arafat that the talks might collapse in that meeting. Holst stressed the danger of the talks collapsing, warning Yasir Arafat that the PLO could never hope to achieve a better agreement than the one on the table (interview Waage, 2019). While the Norwegians manipulated Yasir Arafat, they were extremely helpful to the Israelis by putting pressure on Arafat (Waage, 2019).

Singer, who claims that Norway was a facilitator, mentions "the extra corridor issue" as an exceptional case of the Norwegian facilitation role. He clarifies that Norway at that time exceeded the role of facilitator. He adds that, at that time, the trip of the Norwegian foreign minister, Holst, to convince Yasir Arafat to forget this demand because it would spoil the negotiations. Singer asserts that the Norwegian minister of foreign affairs clearly said to Yasir Arafat: "You cannot raise this demand; it does not belong to you" (interview with Singer, 2019).

Holst succeeded in convincing Arafat to give up his demand for the "extra corridor or "kissing point" as Arafat called it. Holst was fully aware of the fact that this idea was unacceptable to Israel as such a condition would effectively cut Israel into two parts (interview with Gur, 2019).

Holst himself admitted that he pressed Yasir Arafat. This is found in. alter from Holst to the Israeli Foreign Minister Shimon Peres on 16 July 1993. This document is one of many documents related to Oslo negotiations that were lost from the Norwegian Ministry of Foreign Affairs archive. In the letter, Holst

wrote, "I challenged him in this connection to explain his approach of the Jericho-issue and the question of links and access between Gaza and the Jericho-pocket... when I pressed him on the issue, that he was not proposing any corridor controlled by the PLO or an international authority through Israel" (Holst, 1993).

Moreover, Yasir Arafat was threatened by Holst, according to a senior Palestinian official who was very closed to Arafat at that time. The senior Palestinian official explains that the Norwegian foreign minister, Holst, conveyed a threat message to Yasir Arafat: "the PLO's continued presence in Tunisia is not sustainable due to international pressure if the deal is not concluded, the senior Palestinian official says. He adds that Holst's messages were balanced with promises of support. However, the threat of evicting the PLO from Tunisia was an important part of the objective of Holst's meeting with Arafat, according to the senior Palestinian official. He adds Arafat answered that Cairo would be a reasonable substitute as a host of the PLO headquarters, and Holst's answer was that Arafat should feel certain that even that possibility had been taken care of, according to the senior Palestinian official (interview with a senior Palestinian official, 2020). The senior Palestinian official assures that Norway could not do that without backing from the USA (interview with a Palestinian Senior official, 2020).

In this interview, the senior Palestinian official mentions the second type of Norwegian manipulation of Yasir Arafat, the material manipulation; the second kind of manipulation the weak Yasir Arafat by promising him a large amount of money from Norway and the International community in case an agreement would be achieved.

Norway manipulated the bankrupt Yasir Arafat by promising to raise large amounts of money for development and aid, only if he would accept. Waage, the Norwegian professor, assures this. According to Waage, "the main goal of the Norwegians' promise to help the Palestinians build and develop their institutions was to build what could become a state in the future. She adds Norway was able to pump in significant amounts of Norwegian kroner to fund such projects ... money from Norway and help to obtain more money from other funders "(interview with Waage, 2019). Indeed, this is what happened in reality. With the aid of donors, Norway secured money for the new Palestinian entity.

The way the Norwegians dealt with Arafat is inconsistent with the role of a mere facilitator. The Norwegian Minister of Foreign Affairs, Holst, was very much aware of the PLO and Arafat's need for a solution, and he used this need. Holst manipulated Yasir Arafat and the PLO. This fact would never be admitted by the Norwegians, the Palestinians, or the Israelis. The Norwegians would deny practising any manipulation on the Palestinian side.

This would ruin the Norwegian reputation as a norm-based mediator an honest peace broker. Norway would avoid any criticism of its peace-building efforts, on which lots of money were spent.

In sum, the Norwegian intervention between the Israelis and the PLO could be divided into two phases. Phase one extends from the beginning of the

negotiations in December 1992 until May 1993. In the first phase, the Norwegians played the role of facilitator between the two disputants. The second phase started at the beginning of May 1993 and led up to the initial agreement in August 1993. In this phase, the Norwegian role shifted from facilitation to manipulation. It would appear that Norway first took a facilitation strategy approach, and later switched to a manipulation strategy after the Foreign Minister, Holst, joined the mediation process. The same thing can be said for Norwegian impartiality.

Impartiality/ bias

Being an impartial mediator or biased mediator is a debatable issue. While some argue that Norway was biased toward Israel, others argue that Norway was an impartial mediator. Both sides use the historical relationship between Norway and Israel as evidence to support their claim.

The historical relationship between Norway and Israel, the power disparity between the Israelis and the Palestinians, and how Norway managed it brings into question Norwegian impartiality. However, the historical relations between Norway and Israel was one of the many reasons that encouraged Arafat to suggest for the first time in 1979 (Waage, 2007).

Based on those relations, Norway was accused of being biased toward Israel in the Oslo secret negotiations. However, it should also be mentioned that the Israeli-Norwegians relations changed much in the 1990s due to the Israeli security concerns and the way they dealt with the Palestinians during the First Intifada (Waage, 2007).

Indeed, the Norwegian mediation between the Israelis and the Palestinians can be divided into two phases regarding impartiality. The first phase was from December 1992 until May 1993, when the Norwegians were an impartial mediator. The second phase was from May 1993 until signing the agreement, when the Norwegians switched their role and gave away their impartiality by becoming biased toward the Israeli side.

Despite of those historical relations between the Israelis and the Norwegians, the Israelis deny that those relations made Norway biased towards Israel, Singer claims that "The Norwegians were not being biased towards Israel" (interview with Singer, 2019). Singer considers those relationships as being a strong point in the Norwegian mediations. "They were neither considered to be pro-Israelis nor pro-Palestinians, they were friendly to both sides, and they were able to simply serve in a way that was quite objectively sympathetic to both sides", Singer says (interview with Singer, 2019). Corbin supports the Israeli claim and provides assurances that the Norwegians in the Oslo secret channel used their unbiased stance to stay afloat in their difficult position (Corbin, 1994b).

Similarly, Beilin denies that Norway was biased to either the Israeli or the Palestinian side: "I do not think that the Norwegians were involved in supporting the demands of either the Palestinians or the Israelis. They were not in the room at all." He adds that the Norwegians had warm amiable feelings for both the Israelis and the Palestinians and did not approach the two disputants with different attitudes (interview with Beilin, 2019). Beilin was right when he said that Norway was not biased toward Israel. However, this is only true in the

second phase before the Joining of Holst. Holst was in the meeting room. According to Waage, Holst had to embrace the Israeli position; otherwise, it would be no deal (Waage, 2019).

On the Palestinian side, there is no consensus on the issue of Norwegian impartiality. Mohammad Abu-Koush, one of the principal Palestinian negotiators in Oslo, claims that Norway was an impartial facilitator; he says, "The Norwegians were not biased to Israel." Abu-Koush built his argument on the fact that the Norwegians did have a good relationship with Israel but did not support the occupation (interview with Abu-Koush, 2019). However, his justification needs to be more validated. Not supporting occupation is one issue, and being impartial in mediation is different.

In a different vein, a senior Palestinian official who was closed to Yasir Arafat, confirms that Norway was biased towards Israel: "Norway's close relations with Israel did play a significant role in the biased attitude it took. The senior officer expected that Norway would play a role of an unbiased mediator. However, he offers assurances that throughout the process, Norway acted as a support team for the Israeli negotiating team (interview with a senior Palestinian official, 2019). The senior official confirms the biased of Norway towards Israel by asserting that the Palestinians were negotiating two teams on the same issue. He says, "In fact, it turned out that we were negotiating with two teams, the Israeli and the Norwegian, selling us the same goods". The Palestinian officer means that the Norwegians were biased toward the Israelis, and both have the same tongue (interview with a senior Palestinian officer, 2019).

As expected, the Norwegian representatives describe themselves as impartial mediators on the Norwegian side. They claim that they were impartial to both parties. The Norwegians built their claim on how Norway dealt with the two parties, treating them equally by providing them with the same services, the same accommodation etc. Larsen says, "I picked up the Palestinian and the Israeli delegation members at the airport personally, and some of my staff picked up the others. We started with the Palestinians and then the Israelis and then the Palestinians again and we accommodated them...we always used exactly the same hotel rooms, exactly the same food, exactly the same services etc..." (interview with Larsen, 2019).

In a different vein, a Norwegian scholar, Hilde Waage, claims that the Israeli foreign minister used the Norwegian foreign minister as a tool in the negotiations, assuming with good reason that the Norwegians would present the Israeli point of view to Arafat (interview with Waage, 2019).

Waage goes further on the claims of Norwegian partiality and explains that the Norwegians agreed with everything that Israel did and that without doing so, there would have been no agreement. "Norway very strongly wanted to play a role, and in order to do so, Norway had to be completely 100 percent sided with Israel" (interview with Waage, 2019). She explains that this is not an issue of being biased towards Israel. She says that it is a cynical calculation of the best interests of Norway, and this is how to be a weak mediator like Norway. She adds that the Norwegians could have told the Israelis: "Sorry, this is not the way we do it as a moral country". She explains that the Norwegians could have explored other choices. She asserts that Arafat was in a hurry and he

wanted peace, and the Norwegians were more willing to accept compromises and take up a biased role based on Israeli demands (interview with Waage, 2019).

However, Waage asserts that Norway was an impartial mediator when Norway hosted the two disputants in the same hotel and dealt with each in the same way and without discrimination (Waage, 2005). Waage Explains that the turning point in Norwegian impartiality happened when Johan Holst became the Foreign Minister of Norway in May 1993. Holst decided to take a direct and active role in the mediation from the first moment. He was eager to reach an agreement between the Palestinians and the Israelis (Waage, 2019). The way that Holst worked with the Palestinian side to give on specific issues may bring Norway's impartiality into question (Waage, 2005).

In the second phase of the negotiation rounds, when the Norwegian Foreign Minister, Johan Holst, joined the negotiations, he played the role of a critical person with complete political responsibility. He travelled to see Arafat in Tunis and carried the message of Arafat to the Israeli party, showing them the degree to which Arafat needed any solution to avoid being marginalised. Holst spoke to Arafat in a firm voice, advising him to take the chance or lose the opportunity forever. He succeeded in convincing him that the PLO would never achieve a better deal than what was on the table (Waage, 2007).

Waage argues that Norway was pursuing the agenda of Israel and thus was a biased and partial mediator. She explains that superficially, Norway treated

the two parties in equal ways. "However, this was limited to food, cars, and hotels," Waage says.

It sounds as though Hilde Waage is not the only person who asserts that Norway took the position of Israel. Shlomo Gur, the Israeli official who worked as Shimon Peres's office head, also shares Waage's opinion. Shlomo Gur demonstrates that the Norwegians not only agreed with Israel but were ready to do everything to reach an agreement. Gur represents the mutual recognition agreement as an example that Norway agrees with the Israeli position. Israel suggested that the PLO must recognize Israel as a state, so Norway took this position because, without the mutual recognition, there would be no agreement (interview with Gur, 2019). Waage refers to the Norwegian bias towards Israel to the power disparity between the disputants. She asserts that the two parties are in a power asymmetry. Norway did its best to let the Palestinian party feel equal to the Israelis. They did what they could to ensure a symmetrical process by offering the two disputants the same hotels, same cars, same time, and even the same food (Waage, 2007). However, the power disparity encouraged Norway to play the role of partial mediator favouring the stronger party, Israel. Norway was a low-leverage mediator, mediating between two parties with unequal power. Beilin asserts that Israel was the stronger state, while the PLO was still politically and financially weak from the aftermath of the First Gulf war in 1991 (interview with Beilin, 2019). The strength of Israel forced Norway to play the role of partial mediator towards Israel unwillingly. Waage argues that if the mediator is weak in a situation

where one disputant is significantly weaker than the other, then the mediator must stand with the stronger party (interview with Waage, 2019).

Waage adds that the Norwegians were well aware of the power disparity between the two disputants—the Israelis and the Palestinians. The Norwegian Foreign Minister, Johan Holst, was very discerning and perfectly understood that to reach an agreement, they would need to be biased towards the Israelis as they were the stronger party. She assures Holst went all-in on the Israeli side to reach an agreement; Waage explains this was not because Holst likes Israel but simply because this is the approach to use when you are a small mediator dealing with disputants power disparity. Thus, the Norwegians were extremely cooperative with the Israelis, and the whole game became focused on persuading the weaker Palestinians (interview with Waage, 2019).

Norway was biased toward the stronger party, the Israelis. Norway is aware that the one who controls the process is Israel, the one who can stop and end the negotiation is Israel because it is the strong party. Norway wanted to reach an agreement. The only way to reach an agreement is to do what Israel asked and adopt the Israeli position. In other words, to be biased towards Israel.

Norway adopted the position of Israel and worked according to the Israeli interest. Indeed, Norway is not the only side that worked for the interests of Israel. Arafat himself worked for the interests of Israel. According to the "memo" from Johan Holst to Shimon Peres on 16 July 1993, Arafat and Norway were working according to the interests of Israel. Arafat chose Jericho because it did not compromise recent settlements and because it would enable Arafat to

add a major Palestinian camp, making Arafat's mission to control the Intifada easier. Arafat considered the Israeli interests and sensitivities regarding the Intifada and the settlement (Hølst,1993).

Indeed, it is not always correct that the mediator must stand with the stronger party where one disputant is significantly weaker than the other. It could go either way. It depends on the situation. In the Oslo secret negotiations, Norway supported the strong party (Israel) to reach an agreement. However, the Norwegian mediation in the conflict between Sudan and South Sudan shows that Norway supported the weak party, the SPLM, who represented South Sudan to reach an agreement.

In sum, Norway played the role of impartial mediator in the first phase. However, when Holst became the Norwegian Foreign Minister in the second phase, Norway gave away its impartiality. In this second phase, Norway adopted Israeli's vision and built agreements based on what Israel would allow. By adopting the Israeli point of view, Norway was not only biased towards Israel but was utterly supportive of Israel. The Norwegian behaviour in this instance could be attributed primarily to the power disparity between Israel as a strong state and the PLO with weak Arafat.

CHAPTER FIVE: CASE STUDY TWO: NORWAY'S ROLE IN THE COMPREHEN-SIVE PEACE AGREEMENT (CPA) IN SUDAN 2005

Introduction

Since its independence in 1956 from Anglo-Egyptian colonial rule, Sudan has witnessed different internal conflicts and civil wars (Johnson & Matthews, 2005). One of those wars was the civil war in southern Sudan and was essentially a civil war between the North (the government of Sudan) and South Sudan (1983–2005). There has been much debate around the reasons behind this war. However, the first commonly accepted reason is the development disparity between the North and the south (Johnson & Matthews, 2005; Kaldor, 2013; Natsios, 2012). As seen by the southern Sudanese, the second reason for civil war is ethnicity and religion (Deng, 2011; Natsios, 2012).

The North-South civil war ended in 2005 with the accumulation of the Comprehensive Peace Agreement (CPA), which is the case study for Norwegian mediation and the discussion topic of this chapter. On 9 January 2005, the civil war between the Government of Sudan (GOS) and the Sudanese People's Liberation Movement/Army (SPLM/A) was ended by signing the CPA in Kenya. Some variant actors played a role in the mediation process that led to the signing of the CPA, including both regional and international mediators. The regional mediators were from the neighbouring countries of Kenya and Ethiopia, who together constituted the Intergovernmental Authority on Development (IGAD). On the other hand, the international mediators included the USA, the UK, and Norway (Rolandsen, 2005; Taulbee, Kelleher, & Grosvenor, 2014; Zambakari, 2013). Unlike its role in the Palestinian-Israeli conflict,

where Norway was a sole mediator, Norway in the Sudanese conflict acted not alone but alongside the UK and USA, who together constituted the Troika (Katete, 2010; Kelleher & Taulbee, 2006).

Norway is one of the several countries which contributed to South Sudan's birth.

Norway's role was an important one, as Norway acted not only as a facilitator in the peace process but also Norway was an active mediator who funded the peace process and actively participated and supervised separate talks while negotiating the disputants. The Norwegian facilitation role in the mediation process was a by-product of the Norwegian Track One and a Half Diplomacy, in which Norway's mediation efforts are based on the unique cooperation between the Norwegian government and the Norwegian NGOs.

At the time of the Sudan negotiations, the Norwegian government was represented by the Norwegian development minister, Hilde Johnson, who was the primary mediator of the CPA of 2005 (DW, 2014). Two prominent Norwegian NGOs represented the Norwegian NGOs dedicated to the peace efforts: the Norwegian Church Aid (NCA) and Norwegian People's Aid (NPA). Both operated as development NGOs and worked in both Sudan and South Sudan, offering humanitarian aid to the people in need in those areas. Via these two organizations, the Norwegians succeeded in developing personal relations with the government of Sudan and the rebels in South Sudan. Norway then used those relations to mediate between Sudan and South Sudan. Norway worked secretly behind the scenes, making efforts to talk with officials and unofficial stakeholders from different layers in Sudan and South Sudan, and

contributed to the facilitation of negotiations between the disputants in support of US, UK and Kenya-led IGAD efforts.

In its mediation between the government of Sudan and the Sudan People's Liberation Movement (SPLM), Norway had not been fair in the sense of impartiality. The key argument in this chapter is that Norway manipulated both sides in its mediation between the government of Sudan and the SPLM. While in Oslo secret negotiations, Norway adopted the position of the strong party, Israel, in the mediation of the Sudanese conflict, Norway adopted the position of the weak party, the SPLM, and in doing so, forfeited its impartiality. As mentioned, in its mediation in the Sudanese conflict, Norway was among the Troika and so was able to adopt a mediation policy that borrowed leverage from the USA and knowledge from the UK. This stronger, more informed policy was the definitive reason behind the success of the CPA. The three countries constituted a perfect mediation combination of the US carrots and sticks, the Norwegian trust and confidence and the British knowledge of the conflict.

This chapter discusses key subjects to shed light on the Norwegian mediation role in the Sudanese conflict. It includes a brief on the conflict between South Sudan and Northern Sudan, looks at Norway's role in the mediation of said conflict and finally analyses the Norwegian mediation approach.

The conflict between the Government of Sudan and the SPLM (in South Sudan)

South Sudan was announced as an independent state in 2011, and this independence came after many years of civil war and several attempts at mediation which culminated with the signing of the Comprehensive Peace Agreement 2005 (Johnson &

Matthews, 2005). It is important to mention some facts about these two regions. First, the people of northern Sudan are Arab and Muslim, while the people of South Sudan are non-Arab and predominantly Christian. For this reason, the outside world often perceived the conflict as one between the religious views of Christianity and Islam. The southern tribes were decidedly against the policies of forced Arabization and Islamization that the Khartoum government applied to them (Natsios, 2012). Secondly, Sudan has vast natural resources, most of them in South Sudan. Those natural resources include oil and valuable minerals (Natsios, 2012). Indeed, 75% of Sudanese production oil is in South Sudan. Copious quantities of gold, diamond, uranium, copper, and coltan are found in South Sudan. Additionally, South Sudanese wealth is represented in its fertile soils and plentiful rainfall. Due to these ideal agricultural conditions, it is argued that all of Africa and even the Arab world could be fed by farming Southern Sudan (Johnson & Matthews, 2005; Natsios, 2012).

Third, during the British-Egypt rule in Sudan, Britain isolated the south to protect the region from slave traders. Unfortunately, this isolation also resulted in sluggish development in the south and created development disparity between the north and the south (Natsios, 2012). Since that time, the south has largely been ignored by the north. Fourth, the North-South civil war can be divided into two civil wars: from 1956 to 1972 and 1983 to 2005. Fifth, the southern tribes struggled to form a multi-religious and multi-ethnic southern independent state (Natsios, 2012).

To better understand the civil war in South Sudan, it is essential first to look at the history of Sudan and South Sudan in particular. Towards the end of the 19th century, the British occupied Egypt and Sudan. In 1899, the British agreed with Egypt called

the Anglo- Egyptian Condominium, in which it was agreed that Britain would control Sudan (Hanssen, 2017). The country's southern region suffered neglect and marginalisation with minimal infrastructure and investment under the Anglo-Egyptian regime. Moreover, the people of the South were seen as second class citizens (Johnson, 2016). Much to their delight, Egypt and Britain promised to grant Southerners self-determination at the end of World War II. However, this promise was broken by the Northern Sudanese with Egyptian connivance (Johnson, 2016). It was not until 1956 that Sudan gained its independence from British occupation. However, the Southern Sudanese wanted self-determination (Hanssen, 2017).

The outbreak of civil war between the northern and southern Sudanese took place before the independence of the Republic of Sudan in 1956. The outbreak took place in 1954. Between October 18th and the 21st, 1954, southern leaders organised a conference in Juba to vote for independence from Egypt. The conference proposed an autonomous South within Sudan or a form of self-determination leading to possible independence (Natsios, 2012). However, the conference failed (Johnson, 2016). This was followed by northerner domination of the posts left vacant by the departing British in South Sudan. This action resulted in an outburst of anger among the southerners. In July 1955, riots erupted in Yambio, which required police intervention in the chaos -fired into the crowd, killing eight people (Natsios, 2012).

One month later, on 18 August 1955, another riot broke out in Torit, in which southern protestors killed northern officers. Administrators and merchants, along with their families, were also counted among the victims (Johnson, 2003). More than 300

persons were killed (Johnson & Matthews, 2005; Natsios, 2012). Matters quickly escalated into a full-on civil war (Rolandsen & Leonardi, 2014).

On January 1, 1956, the British left Sudan, leaving an independent but unstable Sudanese state with an inexperienced government. However, the British departure and ensuing independence of Sudan did not bring with it the independence of South Sudan but rather substituted one coloniser with another (Lagu, 2006; Natsios, 2012). In 1962, the Sudanese military government led by Abbud restricted the activities of Christian missionaries in South Sudan. From 1969 to 1985, power in Sudan shifted to the government of Colonel al-Numeri. At the beginning of his rule, he promised to resolve the southern Sudan Civil war through political negotiations and not military intervention as, in any case, the military solution would not go well for him (Natsios, 2012).

As Joseph Lagu in South Sudan had succeeded in forming an armed force that matched the North, the outcome of the continued conflict was unsure. Lagu had unified military command under his leadership with 13,000 southern troops equipped with Israeli training and weapons. In January 1971, from Addis Ababa, al-Numeri declared a unilateral cease-fire in Southern Sudan. After this declaration, both parties reached an agreement in February 1972 in Addis Ababa. which was signed by Lagu for the South and Mansur Khalid for the North (Natsios, 2012). The agreement provided for a Sudanese federal state where a legislative assembly and a council of ministers from the south would govern South Sudan. However, the southern government would be appointed by president Numeri (Natsios, 2012).

The Addis Ababa Peace Agreement granted self-government but not self-determination, and, in that respect, it was a fragile agreement in many Southerners' eyes.

Moreover, it was a superficial document with no guarantors or international mechanisms to ensure its implementation (Johnson, 2016). Despite of the weakness of the Addis Ababa agreement, it did lead to stability in the region until 1983 (Natsios, 2012). The defects of the Addis Ababa agreement led to its collapse and contributed to a resumption of the civil war in 1983 (Johnson, 2016). On June 5, 1983, considering many attempts to overthrow him, president Numeri abandoned his position as a socialist to embrace Islamic thought. Accordingly, he announced the death of the Addis Ababa agreement by breaking the south into three provinces with different capitals, replacing the southern Regional Assembly in Juba with three much weaker legislative bodies with no independent authority, eliminating the southern army units, and cancelling the status of English as the official language by substituting it with Arabic. Moreover, he imposed the Islamic Sharia law on all Sudanese people, including those in the south (Johnson, 2016).

This set the second civil war (Natsios, 2012). In reaction to Numayri's violation of the Addis Ababa agreement, on 31 July 1983, Garang, a prominent Marxist southern leader, created both the unified Sudanese People's Liberation Army (SPLA) and the Sudanese People's Liberation Movement (SPLM) as its civilian arm (Natsios, 2012).

Through the SPLM/A, Garang defended the demand and vision of justice and equality for all the Sudanese without discrimination based on religion, language, or ethnic group. He advocated a "New Sudan" where all Sudanese, including the marginalised, would live in a safe country characterised by rightful representation in government, a

country based on multi-religious and multi-ethnic traditions, a country that would respect diversity and serve more than just the elite and privileged (Johnson, 2016). The ultimate goal of Garang was "the right to self-determination." He believed that there should be a referendum in which the South should decide for themselves whether to be independent or to be a part of a united Sudan (Johnson, 2016).

SPLA was trained and funded by Ethiopia and the Soviet Union. Garang was in control of most of the South by the end of 1985 (Natsios, 2012). On 6 April 1985, the Sudanese Minister of Defence, Major General Siwar al-Dhahab, announced that the military had taken control of the government, ousting Numeri. This was followed by a new democratically elected coalition government, led by Sadiq al-Mahdi and his Ummah Party, which had received the largest number of seats in the National Assembly. Unfortunately, Sadiq al-Mahdi's government proved to be no better to the Southerners than Numayri's government had been. The new government moved to destroy the Southern culture and pursued the policy of forced Arabization and Islamization (Johnson & Matthews, 2005; Natsios, 2012; Rolandsen, 2005; Rolandsen, 2011).

On June 30, 1989, a new military coup led by Brigadier Omar al-Bashir successfully ousted Sadiq al-Mahdi's government (Johnson & Matthews, 2005; Natsios, 2012; Rolandsen, 2005). At the beginning of the year 2002, the civil war between the Muslim-dominated government and the rebels of Southern Sudan reached its peak. The Sudan People's Liberation (SPLM) led and represented the rebel groups (Johnson & Matthews, 2005; Natsios, 2012).

In April 2002, the Intergovernmental Authority on Development (IGAD), which had been created in 1996 to strengthen regional cooperation, initiated a mediation to

resolve the conflict of North/South Sudan. IGAD's initiative was led by Kenya and was diplomatically and financially supported by the Troika: USA, UK, and Norway. As a result of these joint efforts, a framework for conflict resolution was formulated, which suggested self-determination in South Sudan. The framework culminated in the signing of the Machakos Protocol in July 2002. The Sudanese government in the North and the Sudan People's Liberation (SPLM) signed the protocol. It formed the basis for talks which led to the comprehensive peace agreement (CPA) of 2005, which covered several issues such as power-sharing, security issues, and wealth sharing (Johnson & Matthews, 2005).

Towards the Comprehensive Peace Agreement (CPA) and Norway's role

It is essential to mention that the peace negotiations between the Sudanese government and the representatives of South Sudan (SPLM/A) were mediated by the IGAD with the support of a Troika (Norway, UK, USA), which was to be its driving force. The role of the Troika included funding the negotiation secretariat, providing mediation experts to assist in the talks, and supporting the primary mediator, the IGAD Special Envoy, General Lazarus Sumbeiywo (Johnson, 2016). However, Norway has played an essential role in the mediation between the government of Sudan and the rebels in South Sudan. The Norwegian role will be discussed in two sections; the first is the Norwegian NGOs' role in the mediation and Norway's role in the mediation process.

The Norwegian NGOs pave the way to the Norwegian international mediation:

Norway has a historical link to the people of South Sudan, thanks to two of the Norwegian NGOs: the Norwegian People Aid (NPA) and the Norwegian Church Aid (NCA). Those two NGOs had worked in South Sudan and Sudan since the 1970s. In both cases, the organizations had close contact with governmental authorities— the NPA with the Southern Sudan Authority, the NPA was only working actively in the SPLMA controlled area, and the NCA was working with the Sudanese government (Johnson, 2019). Therefore, the relationship between the NPA and the government of Sudan suffered because the Government of Sudan thought that the NPA supported the rebels in South Sudan, the SPLM (Amum, 2019). Indeed, the NPA supported the people behind the SPLM line in the SPLM-controlled areas, constituting 30% of the South Sudan territory (Amum, 2019).

Norway's funding for South Sudan goes as aid through two kinds of organizations; Norwegian NGOs such as the NCA and NPA, and international organizations such as Save the Children, Red Cross, United Nations, and the World Bank (Tvedt, 1994; Tvedt & Badal, 1994). However, most Norwegian funding to South Sudan went through NCA and NPA. The NCA has developed many sectors in South Sudan, such as education, health, agriculture, and construction. It repaired and built new roads, established primary health centres and created primary and secondary schools. Additionally, the organization drilled many wells in South Sudan (Tvedt, 1994; Tvedt & Badal, 1994). In the same manner, the NPA provided humanitarian assistance to the Sudanese people in parts of South Sudan under the SPLM/A (Kaiya, 2016; Shanmugaratnam, 2008; Shanmugaratnam et al., 2002). Furthermore, the NPA developed different sectors in South Sudan, such as the health sector, food security

sector, and educational sector (Copnall, 2014; Ø. Rolandsen, 2005; Shanmugaratnam, Mamer, & Kenyi, 2002).

For the Norwegians to be mediators, they first had to make two essential connections: the SPLM and the Government of Sudan. The Norwegians succeeded to make those two connections through NPA and NCA. The NPA had been working with the humanitarian arm of the SPLM, the Sudan Relief and Rehabilitation Agency. NPA worked only in the areas which SPLM controlled. Consequently, it succeeded in developing credibility with the southerners. The second connection was the NCA which had worked in the government-controlled area and the rebel-controlled area. Furthermore, it had access to the SPLM/A leadership and the Sudanese government (Johnson, 2011).

Because Norway's contacts originated in their historical links to these NGOs, the relationship between Norway and the leadership of SPLM gained significant credibility.

This credibility was the basis for Norway's role; there would not have been a Norwegian mediation (Johnson, 2019). Additionally, the NPA and the NCA personnel knew the Sudanese people at all levels from top to bottom. They were in touch with the local people in Sudan and South Sudan. The Norwegians, via their NGOs, succeeded in building trust and strong relations with the Sudanese government, the opposition, the rebels, and the local communities (Kelleher, 2006). This trust enabled the Sudanese to perceive Norway as a mediator who had no interest in the conflict, thus easing the process of mediator acceptance. Additionally, Norwegian humanitarian aid and development work in Sudan enabled the Sudanese to see Norway as credible (Kelleher, 2006).

The Norwegian NGOs do things the Norwegian government cannot do alone. For example, in many conflicts, the Norwegian government cannot alone offer humanitarian aid, the promotion of human rights and peace prevention. These are considered sensitive issues, and thus, the Norwegian government opts to conduct these activities via its NGOs (Bandarage, 2011).

However, Norway is dependent on its NGOs for peace-making and has often been criticised by those who view its NGOs to disguise the interest of the state. In other words, the Norwegian government involves its NGOs to achieve a specific interest. For example, Goonatilake claims that Norway created a new form of colonialism called "NGOs colonialism" by establishing a new independent class of local people capable of making a stand against the sovereignty and serving the external donor as opposed to the local actors (Goonatilake, 2005).

The argument goes further, accusing certain church-based Norwegian NGOs such as NCA of focusing on areas where Christians are a minority to change the cultural domination, much as it has in South Sudan (Bandarage, 2011). The Norwegian NCA facilitated the secessionism of those minorities from society (Bandarage, 2011). Often in doing so, the Norwegians, the "potent small power", serve the interest of the US, "the impotent superpower" (White, 2005).

Indeed, with the cooperation of the Norwegian government, NCA made continuous efforts to influence the two parties (Bandarage, 2011). For example, Halvor Aschjem from NCA, using his personal relationship with the Sudanese government and SPLM, invited both representatives of the Sudanese government and rebel representatives to his farm in Norway (Kelleher, 2006). Additionally, the Norwegian Minister of

Development, Hilde Johnson, was also directly involved in the negotiations between the Sudanese government and rebel leadership (Kelleher, 2006; Kelleher & Taulbee, 2006).

The Norwegian government and Norwegian NGOs in Sudan and South Sudan work together to end the conflict. The role of the NPA and the NCA in the development of Sudan and South Sudan was the basis of Norway's positioning with the Norwegian government in Sudan and South Sudan (Johnson, 2016).

Norway, Troika, IGAD in the mediation process

Despite being a part of the Troika and not the primary mediator, Norway has played a prominent role in the mediation between the government of Sudan and South Sudan.

The Norwegian mediation idea was laid out in the first meeting held by Hilde F. Johnson, the Norwegian minister for Human Rights and International Development, with Ali Osman Mohamed Taha, Sudan's Foreign Minister. The meeting took place in Rome in the autumn of 1997. Both ministers were on a mission to attend the meeting of IGAD and the international support group in the context of the IGAD partner's forum. During the meeting, Ali Osman Taha emphasised the necessity of direct talks with SPLM/A (Johnson, 2011).

Hilde Johnson chaired the Troika team and managed the support efforts. She used her relations with all parties; the Sudanese government, the rebels in South Sudan (or freedom fighters as she preferred to call them) and the primary mediator and IGAD representative, General Sumbeiywo (Johnson, 2016). She also used her close relations with the British Minister for International Development and the US envoy to

Sudan. Leveraging this impressive network was instrumental in achieving the CPA (Johnson, 2016).

In 1989 Bashir and Garang met in the state House Nakoura, refusing to shake hands. They were convinced that peace was not on the agenda at that time. Then, in 1994, the Sudanese government and SPLM/A signed a declaration of principles (DOP) which constituted the basis of the conflict resolution between the two parties. The DOP asserted that the law must guarantee social and political life equality for all Sudanese. While SPLM/A appended its signature on 20 May 1994, the government of Sudan did not append its signature until 1997.

Later, in July 1997, the two parties met again at the statehouse in Nairobi to discuss the possibility of peace in Sudan. This time the two parties were convinced that fighting would lead nowhere and that only negotiations could bring about peace and stability in the region (Waihenya, 2006). Kenya, Eritrea, and Ethiopia were also adversely affected by the conflict in Sudan. They had to host the millions of displaced Sudanese who crossed their boarders (Waihenya, 2006). Ultimately, they agreed to establish the IGAD peace process in Sudan. Kenya appointed General Sumbeiywo as its special envoy to the Sudan Peace process (Waihenya, 2006).

However, there was little progress except for the agreed declaration of principles in 1997, which would become the basis for any further negotiations between the government of Sudan and the SPLM/A. In the mid-1990s, IGAD partners' forum had grown to include the allies of Sudan, including Norway and the Netherlands. The main goal of this forum was to support IGAD and the declaration of principles. Egypt and Libya tried to undermine the efforts of IGAD by announcing a joint Libyan-

Egyptian initiative with points similar to the declaration of principles but differing on the matter of self-determination (Johnson, 2011). By the time of negotiations, the international context had changed, and it was the time that the Sudanese conflict had reached the moment of ripe. The September 11 terrorist attacks in the USA brought the Sudanese government to the negotiation table. Additionally, the Sudanese debt, at that time, had reached \$22 billion (Johnson, 2011). However, in 2003, the negotiations between the Sudanese government and the Sudan People's Liberation Movement and Army (SPLM/A) had halted.

Consequently, on 31 August 2003, Ali Osman Taha phoned Hilde Johnson, telling her about the halt of the negotiations, and he recommended negotiating on a higher level and requested that she arrange a secret meeting between himself and John Garang in Nairobi as he was going to attend the funeral of the Kenyan vice president. Osman Taha insisted on keeping this confidential. This meeting was the beginning of a long series of meetings that lasted for more than six months and culminated in Sudan's 2005 Comprehensive Peace Agreement (CPA). IGAD was involved in those meetings (Johnson, 2016).

The establishment of the Troika was based on the cooperative work between Hilde Johnson, the Norwegian Minister for International Development, and Clare Short, the UK Secretary State for International Development. They both had worked together on several development issues in Africa since 1997. The three pillars of the Troika, the UK, USA, and Norway, were collaborating closely, serving as the main organisers of leverage and international strategies coordinators. Norway's involvement with the mediation process was carried out by its official and unofficial actors (Kelleher, 2006; Kelleher & Taulbee, 2006). The Norwegian Ministry of Foreign Affairs and Norwegian

NGO's personnel cooperated to support the mediation efforts of the US, the UK, and Kenya led IGAD (Kelleher, 2006). They supported IGAD by offering it consultation, knowledge, and finance to cover the cost of mediation. Norway was one of the lead international observers in the mediation process and a valuable partner of IGAD. Usually, Norway occupied the position of chair or co-chair in the IGAD Forum (Kelleher, 2006).

The rationale of the Troika was to form a small team of countries with strong relations on both sides of Sudan, Sudan and South Sudan. The UK has a long colonial history with Sudan, and as such, the British had extensive knowledge of the country and deep contacts with important persons in Khartoum. It was also crucial to involve the USA in any solution in Sudan as no mediation effort would have likely succeeded without them. The USA is a prime example of a powerful country; they have the sticks and the carrots at their disposal. These sticks and carrots were essential for the success of the Troika. For example, The September 11 attacks had an impact on the Sudanese government and encouraged them to make peace in the south of Sudan. The Sudanese government was convinced that cooperation with the USA against terrorism and making peace with the south would improve their relations with the Americans, and they were keen to end their isolation. However, the USA alone could not have succeeded without the cooperation and advice of the UK and Norway (Kelleher, 2006; Kelleher & Taulbee, 2006).

It was essential to involve Norway in the mediation because Norway, via its NGOs, had close relations with the Southerners. NPA and NCA had worked for decades in Southern Sudan. The presence of Norway in the Troika gave South Sudan the confidence to accept the Troika and move forward with peace efforts (Johnson, 2011).

The Norwegian team included extensive experience and long-standing contacts in Southern Sudan. Johnson was a co-chair on the Sudan committee at the IGAD partner forum and contacted the Kenyan government and general Sumbeiywo. In March 2002, IGAD, represented by Sumbeiywo, consulted with the parties. Prior to the first summit of IGAD, General Sumbeiwyo drafted a paper entitled "One country, two systems", which left out the critical issue of self-determination. At that moment, the Troika decided to intervene with a significant push in the negotiations. In late April 2002, the Troika met in New York to determine their role in the mediation. They decided to send their envoys as observers to the negotiations and technical experts to fill support roles in the mediation. Additionally, they resolved to work behind the scenes to influence the parties to reach an agreement. General Sumbeiywo cooperated with the Troika and consulted the observers daily.

On July 16, 2002, Johnson met in Oslo with Jack Danforth, the American special envoy to Sudan. She told him about the tensions in the negotiations, confirming to him that the SPLM/A was losing confidence in the talks and had begun to express dismay. She referred to the paper of Sumbeiywo, where self-determination was formulated within the context of a united Sudan and not following the IGAD Declaration of principles (Johnson, 2011).

Johnson adopted the SPLM position. Furthermore, Norway could, to a degree, control the decision of the Troika members through Hilde's relations with Clare Short and Jack Danforth and could turn their opinion in support of SPLM/A.

It is worth considering the background of Hilde Johnson, the daughter of a missionary family in Tanzania, where she was born. In addition, she was aware of the history of

South Sudan and the self-determination that had taken root shortly after World War II. Johnson assures that many southerners believed that the Addis Ababa agreement of 1972 collapsed because it was weak regarding specific issues such as self-determination, security issues and international guarantees (Johnson, 2011). For example, the previous Sudanese governments accepted self-determination in the Khartoum peace agreement of 1997. Additionally, self-determination was mentioned in the constitution (Johnson, 2011).

Norway was the prominent director in the negotiations. When the negotiation reached a deadlock, Johnson called the Norwegian representative in the Troika, instructing him to discuss the situation with the other members to change the course of their approach to gaining agreement with Sumbeiywo. Following this Norwegian intervention, the chief mediator, Sumbeiywo, promptly changed the course of the negotiations and decided to put the matter of self-determination in the south to a referendum (Johnson, 2011).

Norway borrowed leverage from the USA in the Troika to increase pressure on the issue of self-determination. President Bush at that time took a stance of support for the southern cause. On 20 July 2002, the negotiations endorsed the Machakos protocol.

The comprehensive peace agreement would not have been negotiated and reached without this protocol. The protocol contained details related to the referendum, self-determination, approach to religion and the state and information on the transition period (Johnson, 2011). The Machakos Protocol granted Southern Sudan the right to exercise self-determination while Sudan was guaranteed an Islamist character under

Sharia Law. However, the negotiations nearly collapsed on September 1st, 2002, following the SPLA attack at Torit. The SPLA, in turn, claimed that Khartoum provoked them and attacked them first. At that time, the primary Troika role was focused on achieving a cessation of hostilities. Finally, in October, the disputants were back at the negotiation table.

However, the negotiations nearly collapsed again in July 2003 when the Sudanese government proposed the Nakuru draft to resolve the remaining issues to avoid the IGAD process. Nakuru draft is a draft framework for resolving the outstanding Issues Arising out of the Elaborations of the Machakos Protocol (Reeves, 2003). Ali Osman Taha saved the position and asked Hilde Johnson to contact Garang, the SPLM/A representative, to convince him to have direct talks in Nairobi. Again, Norway succeeded in helping to bring the two leaders, Osman Taha and Garang, together. They negotiated directly and intensely for the next 18 months, starting on 31 August 2002 and ending on January 2004. Finally, those negotiations resulted in the Comprehensive Peace Agreement signed on 9 January 2005 in Nairobi (Johnson, 2016). For six years, a sharing of power between the NCP and the SPLM was achieved. Thanks to the Comprehensive Peace Agreement, with a referendum on self-determination for the South to follow. The CPA suggested that John Garang be considered for first vice-president, with Ali Osman to be the second. Additionally, a specific formula was applied to central and state governments (Flint & De Waal, 2008).

The Norwegian mediation approach in the mediation of South Sudan/ Sudan conflict

As mentioned in the methodology chapter, Norway's mediation in the Oslo accords is different from its mediation between Sudan and South Sudan. In the Oslo Accords, Norway was the sole mediator. However, in the mediation between Sudan and South Sudan, Norway was among the Troika —Norway, the UK and the USA. The three external mediators coordinated their peace efforts at a very high political level with high leverage (Prendergast et al., 2002). Norway's mediation role in the Sudanese conflict was to ease the mediation process and support IGAD peace efforts. Relying on its track record of collaborative development projects and deploying its extensive field experience, Norway was able to add much to the peace efforts in the Sudanese conflict (International Crisis Group, 2002b). The existence of an international key player, such as the USA, provided Norway with the influence, support and leverage needed in the mediation process (International Crisis Group, 2002b; Kelleher, 2006; Kelleher & Taulbee, 2006).

The main argument, in this case, is that in the mediation of the negotiations that led to CPA, Norway was not an impartial mediator to achieve this goal and end the long-running conflict between Sudan and South Sudan. Additionally, Norway used a variety of mediation strategies, facilitation, formulation, and manipulation. The following section will focus on the Norwegian motivation for mediation, the mediation strategy, and Norwegian impartiality.

The Norwegian motivation

In its mediation between the government of Sudan and the rebels in South Sudan, Norway had no specific interest except for humanitarian concern and ending a long-running conflict that had claimed many lives. However, there is debate around the Norwegian motivation in the mediation between Sudan and South Sudan. Some evidence supports that Norway had no interest and was a norm-based mediator. For example, Neumann shows that Norway's motivation to mediate the Sudanese conflict is norm-based (Neumann, 2004). In the same vein, Egeland and Ingebritsen demonstrate that Norway is a norm entrepreneur who intervenes in mediation primarily to spread democratic concepts and help people around the world to live better lives (Egeland, 1988; Ingebritsen, 2002). Similarly, International Crisis Group reports affirm that Norway has no self-interest in Sudan and that Norwegians have no investment in the oil industry in Sudan (International Crisis Group, 2002a).

Kelleher (2006) and Prendergast et al.(2002) explain that Norway was a non-threat-ening mediator who had no direct interest in Sudan and that this is the reason why they were perceived as a very acceptable mediator for the Sudanese disputants in both the north and the south (Kelleher, 2006; Prendergast et al., 2002). Likewise, Hilde Johnson, the primary mediator in the Sudanese conflict, asserts that Norway's efforts to resolve the conflict in South Sudan amounted to a humanitarian issue. She explains that Norway started to consider involvement in South Sudan because of its long-entrenched conflict and the human price. She confirms that the South Sudan war is the longest civil war in Africa, so far, two million people have been killed, and four million people have been displaced (interview with Johnson, 2019). In the same

vein, Sumbeiywo, the IGAD representative, points out that Norway had no specific interest in the Sudan/South Sudan conflict. He suggests that Norway's interest is solely in stopping the war (interview with Sumbeiywo, 2019). Pagan Amum, South Sudan's representative, asserts that Norway has no intrinsic motivation. He claims that Norway's sole agenda in South Sudan is pursuing peace (interview with Amum, 2019). Similarly, Kjell Hødnebø, a Norwegian mediator in Sudan/South Sudan, shows that Norway has no national interest in either of those areas. He demonstrates that the Norwegians are interested in stopping the war, returning people to a peaceful land, and helping them. He adds that the Norwegian NGOs and the Norwegian government agree. Many people in the Norwegian government began their careers in NGOs, so they are activists in Norwegian foreign policy and share the same views (interview with Hødnebø, 2019).

Other evidence suggests that Norway did have an interest in the mediation between Sudan and South Sudan. Challenging the Norwegian norm entrepreneur image, Kelleher (2006) discovers that the Norwegians articulate an interest-based rationale for their conflict resolution policy and that Norway's peace efforts in Sudan had other motivations in addition to humanitarian impulses – oil interests and business interests. Norway's reputation makes it easier for Norwegian business people to work in Africa (Kelleher, 2006). Krøvel (2011) claims that the Norwegian Ministry of Foreign Affairs exploits the image of Norway as a peace mediator to raise the standing of Norway in the international arena and uses it as soft power to pursue Norwegian interests, including trade negotiations (Krøvel, 2011).

The claim that Norway has a specific interest, such as oil, could be refuted by evidence from the representative of the Government of Sudan, Hassan Omar. He demonstrates that neither oil nor the desire to support the marginalised was the main factor in Norway's desire to mediate (interview with Hassan Omar, 2019). While Hassan Omar negates the Norwegian oil interest in Sudan/South Sudan, he introduces another interest of the Norwegians in Sudan/South Sudan, which is "supporting the Southern Sudanese Christian population to acquire political gains" (interview with Hassan Omar, 2019). Hassan Omar is not Christian. He is a Muslim Sudanese; when he mentioned that Norway supports Christians, he means that Norway supports Christians because the majority of South Soudan are Christians. He would like to say that Norway was biased toward the SPLM in South Sudan because they are Christians and the Christians in the world support each other.

Indeed, the representatives of the Sudanese government are sure that Norway has a vested interest but, this interest is not in Sudan's oil industry. On the contrary, they believe that Norway's mediation has a religious rather than economic bias and is under the influence of the Norwegian church, as most of the people in South Sudan are Christians.

In the same vein, Abdelrahman Elkalifa, the representative of the government of Sudan, confirmed that Norway's interest in the mediation in Sudanese conflict is not in the oil industry. However, he also asserts that Norway's engagement in the Sudanese mediation is on a religious basis: "I do not believe that Norway's involvement was due to its interests in the oil industry. This could be valid for the USA, but not Norway, the involvement of which was based on religion" (interview with Elkhalifa, 2019).

Elkhalifa confirms that the cliché that the South of Sudan was a marginalised, oppressed Christian area is misleading and untrue. He says, "The rebel leaders who spent most of their life abroad were shocked to see how underdeveloped the far north was and questioned if it were only for marginalisation, oppression and Christianity why were they forcefully fighting after separation...unfortunately, the hypocrite western circles wanted to eclipse the negative British role and the missionaries who spread hatred against Islam intending to block the spread of Islam way south of the Sahara" (interview with Elkhalifa, 2019). Hassan Omar's perspective is not far from that of Elkalifa, as he asserts that Norway's involvement was motivated by religious reasons based on Christianity (interview with Hassan Omar, 2019).

Hassan Omar (2019) confirms that the Government of Sudan was portrayed as a perpetrator of heinous atrocities in Norway while the SPLM was always portrayed as the victim. He explains that this interpretation of the situation is in many instances declared even to the Government of Sudan (GOS) team so that they were not in a state of self-deceit and under the assumption of even-handed behaviour from the part of the Troika, including Norway (interview with Hassan Omar, 2019). Moreover, Hassan Omar asserts that this difference was the main factor that shaped the Norwegian team's perspective because they were there under pressure from Christian NGOs (interview with Hassan Omar, 2019). Pagan Amum, the representative of South Sudan, disagrees with Hassan Omar by arguing that Norway does not place religion as the core issue of its mediation. He adds that Norway saw the conflict in Sudan at that time as a political problem and their goal was to bring the two parties together (interview with Amum, 2019).

Hilde Jonson denies that the Norwegian mediation in Sudan has anything to do with religion. Instead, she asserts that Norway merely helped people who were suffering (interview with Johnson, 2019). She introduces the fact that many of the residents of South Sudan were Muslims, not Christian. One of the representatives of South Sudan in the negotiations was Yasir Arman, a political figure in SPLM and also a Muslim (interview with Johnson, 2019).

In sum, while there is enough evidence that Norway has no specific interest in the mediation of the conflict between the government of Sudan and South Sudan, there is a mixture of evidence to support the claim that Norway has national or state interest in the mediation of the Sudanese conflict (GOS/ South Sudan). Moreover, it is undoubtedly true that Norway has no economic interest in Sudan or South Sudan. Based on the previous debate, Norway likely had no interest in mediation between South Sudan and Sudan. Norway aimed to end the war and save millions of lives in South Sudan of people who had suffered from oppression, discrimination, and war for decades. Norway was a normative-based mediator who had no interest except for promoting peace. In order to achieve its goal to promote peace and end the war in South Sudan, Norway used its leverage.

The Norwegian leverage

Norway's leverage in its international mediation of the North-South Sudan peace process has three sources: Track One and a Half Diplomacy, the Norwegian economic resources, and a mediator among Troika. The first source of Norwegian leverage is Track One and a Half Diplomacy. The Norwegian officials (the Norwegian government) and unofficial actors (Norwegian NGOs) cooperate and work closely with each

other to coordinate joint efforts and activities in the name of peace-making (Kelleher, 2006). Norwegian capabilities can be seen reflected in their field experience. This is especially true of Norwegian NGOs personnel and their grassroots relations with the people and Sudanese leaders. Norway depends on its NGOs for their contacts and experience to pursue their peace agenda and ease their mission in international mediation. Norwegian NGO representatives have intensive international experience as they have dealt extensively with events in the region and have important contacts with relevant actors. Norwegian officials can use these experts when making important decisions for any international mediation attempt (Kelleher, 2006). In the case of the north-south Sudan conflict, it is not only the Norwegians who depend on the field experience of Norwegian NGOs but also their American and British counterparts (Kelleher, 2006). Regarding Sudan, through the Ministry of Foreign Affairs, the Norwegian government has conducted various humanitarian activities and has supplied significant funds to South Sudan. As mentioned, Norway's fund for South Sudan goes through the Norwegian Church Aid (NCA) and the Norwegian People's Aid (NPA) (Tvedt, 1994; Tvedt & Badal, 1994).

The Norwegian persuasion ability results from the confidence that Norway succeeded in building the southern and northern Sudanese people. This confidence came from Norway's strong and lasting relations with the Sudanese rebels in the south and the Sudanese government in the North. Generous Norwegian humanitarian aid was instrumental in developing these relations in each region. This was carried out via the cooperation of Norwegian officials and nonofficial representatives in Sudan, Track One and Track Two Diplomacy (Government / NGOs). The second source of Norwegian leverage is Norwegian economic resources. As a persuasion

tactic, Norway spent considerable amounts to fund the meetings and travel costs for the disputants (Kelleher & Taulbee, 2006). The long-standing humanitarian aid and the confidence between Norway and the Sudanese made Norway an acceptable mediator for both disputants in the conflict (Prendergast et al., 2002).

Finally, being a mediator among Troika is considered leverage to Norway. As mentioned in its mediation in the Sudanese conflict, Norway was not alone. It was in partnership with the UK and USA through the Troika and supported IGAD mediation efforts led by Kenya. Prendergast et al. (2002) assure that the Sudanese negotiations would not have reached a comprehensive agreement without this collation of multiple mediators. With its low leverage, Norway alone would not coax the disputants into an agreement. The existence of an international key factor such as the USA thus supplies the needed leverage to reach an agreement (Prendergast et al., 2002). This is particularly true in complicated, long-term peace processes such as the case of Sudan, where powerful and less powerful mediators work together to reach an agreement (International Crisis Group, 2002b). The three external mediators; Norway, the UK and the USA, coordinated their peace efforts at a very high political level and with high leverage (Prendergast et al., 2002). Norway's mediation role in the Sudanese conflict was to ease the mediation process and support IGAD peace efforts. Relying on its track record of collaborative development projects and deploying its extensive field experience, Norway was able to add much to the peace efforts in the Sudanese conflict (International Crisis Group, 2002b). The existence of an international key factor such as the USA provided Norway with the influence, support and leverage needed in the mediation process (International Crisis Group, 2002b; Kelleher, 2006; Kelleher & Taulbee, 2006). For example, Egypt was angry and reprimanded the

Sudanese government for signing the Machakos Protocol. Egypt did not welcome any change to the international water-sharing agreement. However, in mid-August, the Americans contacted the Egyptians on the political level to calm them down. US envoy Jack Danforth travelled to Cairo and returned with reports that the situation in Egypt had improved (Johnson, 2011).

By obtaining backing from the USA for its mediation with the Sudanese government, Norway was able to become a more successful mediator (International Crisis Group, 2002a; Kelleher & Taulbee, 2006). Sørbø et al. (2011) assure that with its low leverage, Norway needed to borrow the clout of a superpower, such as the USA, in order to be more effective in its mediation and help the Sudanese disputants to reach an agreement (Sørbø et al., 2011).

Gen. Lazaro Sumbeiywo, the IGAD envoy and the primary mediator in the Sudanese conflict, asserts that the CPA could not have been reached without the combination of Norway, the UK and the USA. He explains that the conflict in South Sudan was an overly complicated issue. It required the presence of the Americans, the knowledge of the British and the trust of the Norwegian (interview with Sumbeiywo, 2019). Norway cooperated with the US and the UK to support IGAD in its mediation between the government of Sudan and South Sudan. The Troika shares views on the conflict. However, if there is a misunderstanding between USA or UK, Norway is the one who is usually fixing the situation. For example, Norway succeeded to convince the Americans of the importance of self-determination for the people in South Sudan.

Mediation Strategy

In the mediation Sudan-South Sudan conflict, Norway used a variety of mediation strategies, facilitation, formulation, and manipulation strategy. In order to bring the parties together, Norway employed a facilitation strategy using its NGO's massive relations with the local community in both Sudan and South Sudan. However, Norway acted as a manipulator to encourage the parties to reach an agreement. In certain positions, Norway suggested proposals and played the role of the formulator.

Before discussing the Norwegian strategy in the mediation in the Sudanese conflict, it is worth mentioning that Norway was among Troika (Norway, UK, and the USA) who supported IGAD in its mediation between the disputants (interview with Johnson, 2019).

Troika played a role of manipulation in the mediation of the Sudanese conflict. Pagan Amum, the negotiator from South Sudan, confirms that-the USA brings with them their stick and carrots, the UK comes with their historical background and historical knowledge as Sudan's former coloniser, and Norway comes with the carrots alone to increase the impact of the carrots on the one hand and to reduce the negative impact of the stick on the other (interviwe with Amum, 2019). Hilde Johnson agrees with Pagan by asserting that the combination of the three countries in the Troika was perfect. The UK, with its close relationship with Khartoum, the USA with its carrots and stick, and Norway with its ties to Southern Sudan, made the perfect mediating team. She confirms that Norway as a sole mediator would not have succeeded in reaching the CPA without the other members of the Troika to push for peace in South Sudan (interview with Johnson, 2019).

Hassan Omar explains that Norway was not alone in its mediation in the Sudanese conflict and that the effort was a group effort. However, he asserts that although Norway may be given credit as the most active member in the Troika pushing for the promotion of peace (interview with Hassan Omar, 2019).

There is a divergence of views regarding Norway's strategy in the mediation between Sudan/South Sudan. There is enough evidence to prove that Norway was a facilitator. However, there is significant evidence that supports the claim that Norway behaved as a manipulator in the mediation between the Government of Sudan and South Sudan. Additionally, there some events prove that Norway was a formulator.

Based on the facilitation strategy, in intractable conflicts such as the Sudanese conflict and the Palestinian-Israeli conflict, the mediator must adopt a facilitation strategy to create a communication channel between the disputants. Parties in intractable conflict have no direct communication channel and thus have no agreement on the central issues.

Bercovitch (2004) explains that the facilitation strategy helps the disputants to communicate better and creates a friendly environment (Bercovitch, 2004). According to Kelleher (2006), the Norwegians transmitted the necessary information between the disputants who were otherwise unable to say it directly to one another (Kelleher, 2006). In this framework, Kelleher explains that the Norwegians serve as a go-between, suggest innovative ideas, help the disputants to communicate, and clarify critical situations. She asserts that the Norwegians succeeded in finding a middle ground to work towards a compromise. She adds, in doing so, encouraged the negotiations to go on. A key role the mediator fills is to suggest alternatives, especially when the

negotiations are deadlocked (Kelleher, 2006). As a part of its facilitation role and specifically in deadlock situations, by aiming to keep negotiations moving, Norway on several occasions hosted representatives from both disputants in a relaxed environment away from the conflict where ideas can more easily be exchanged (Kelleher, 2006; Kelleher & Taulbee, 2006).

Johnson assures that the Norwegians played a facilitation role in their mediation between Sudan and South Sudan (interview with Johnson, 2019). By employing the facilitation strategy, Norway had capitalised on the trust it had built with the disputants through the Norwegian NGOs. The Norwegian officials and NGO practitioners engaged in the relevant negotiations with both parties. Norwegians know much about the situation in Sudan and South Sudan and usually come forward with a constructive suggestion when the negotiations reach a deadlock (Kelleher, 2006). This is what Norway offers in its mediation process. However, Norway sometimes exceeds its role as a facilitator and acts as a manipulator.

Indeed, two kinds of manipulation represent the mediation between Sudan and South Sudan. The first type of manipulation is the manipulation that the Troika practised, and the second kind is the manipulation that Norway itself practised. While the first kind of manipulation was practised on the government of Sudan, the Norwegian manipulation was practised on both sides - towards South Sudan and the Sudanese government.

According to Hassan Omar, the representative of the GOS, the Sudanese government was manipulated by the Troika. He adds that there were many differences between the three countries - Norway, the USA and the UK. However, they distributed

different roles but were largely in agreement about how to pressure the government of Sudan to accept the mediation proposals (interview with Hassan Omar, 2019).

He adds that the Troika practised manipulation on the government of Sudan at a time when Sudan was weak and could be easily manipulated. The terrorist attacks in Dar Alsalam, Nairobi, and the subsequent American military intervention made the Sudanese regime more flexible and accepting of the idea of negotiating a settlement

(Hanssen, 2019).

However, the inclusion of the USA in mediation is a success factor. Hassan Omar asserts that the participation of the USA was essential; otherwise, it would have been unexpected for GOS to accept the Norwegian partnership knowing the Norwegian bias in favour of the SPLM (interview with Hassan Omar, 2019). He explains that the Comprehensive Peace agreement in 2005 could not have been reached without the backing of the US and the UK because Norway did not have the leverage that was available to the USA, nor did it have the trust of the main party, the GOS (interview with Hassan Omar, 2019).

Johnson explains that the Sudanese government was aware of their need for a peace agreement with South Sudan during the negotiations. After the 9/11 attacks on New York, the Sudanese shifted their position towards peace. The main reason for shifting their position was that the Sudanese government was anxious for the USA not to label them as a regime supportive of terrorists and instead put Khartoum on the map as a country of reasonable Muslims. Khartoum understood that if they made peace with the South Sudanese, they would benefit from a bilateral relationship with the USA, which is essential for the peace process (interview with Johnson, 2019).

Kjell Hødnebø, a Norwegian diplomat, explains that the presence of the Americans via Troika is significant for the Norwegian mediation between South Sudan and Sudan. He adds that Norway kept updating the USA because Norway could not do anything without having the Americans on board. So in the mediation between Sudan and South Sudan, the Norwegians succeeded in building a special relationship with the Americans. Norway is aware that it is a small country with no power, and because of that, Norway relies on intelligent strategies. However, there were some issues on which people would not compromise. It was then that the USA came in to twist their arms. This may explain the Norwegian eagerness to involve the Americans and develop a stronger relationship with them (interview with Hødnebø, 2019).

Kjell Hødnebø explains that at that time, some Sudanese were involved in the bombing of the Twin Towers in New York in 2001. Rumours followed this: the Alshefa factory in Khartoum produced chemical weapons, resulting in the Americans being bombed. These circumstances led to no relationship between Khartoum and the Americans. Consequently, Khartoum saw these negotiations as an effort to stop or prevent American sanctions. They thought that if they were viewed as the good guys and gave the SPLM something, they would ensnare the Americans. Instead, the Americans were the big sledgehammer, as Norway had no sledgehammer. "We can be only smart, and sometimes you need a sledgehammer to twist arms when it comes to the end of some talks" (interview with Hødnebø, 2019).

While Kjell Hødnebø shows that the USA manipulated the Government of Sudan by being a sledgehammer, Abdelrahman Elkalifa, the Sudanese government representative, argues that Norway also manipulated the government of Sudan, "Norway and others offered their utmost best help to the rebels and were inclined to put some

pressure on the government of Sudan." They were inclined to put pressure on the government delegation, but never on the other side, he added (interview with Elkhalifa, 2019). Hassan Omar assures that Norway, with the aid of the US, manipulated the Sudanese government to reach an agreement (interview with Hassan Omar, 2019). Kjell Hødnebø demonstrates that mediators can exert pressure on both disputants by different means (interview with Hødnebø, 2019).

Away from pressure manipulation, Norway used gratification manipulation by promising the two parties economic and development assistance for investment in the oil sector. In the same vein, Pagan Amum, the SPLM representative in the negotiations, assures that in specific points, Norway manipulated the parties to encourage them to reach an agreement and to achieve peace (interview with Amum, 2019). Amum shows that Norway is rich; the Norwegians offer developmental and economic assistance if the parties reach peace. He explains that while the Sudanese government and the SPLM were negotiating, the Norwegians promised to help both Sudan and South Sudan develop their oil sector further and contribute to the increased recovery of their oil wells. He assures that after the signing of the agreement, Norway helped both South Sudan and the government of Sudan to recover more oil and to increase the production of oil (interview with Amum, 2019).

Furthermore, Kjell Hødnebø demonstrates that Norway promised both sides to help them build a better oil administration and, in particular, assist in negotiations with oil companies. He adds that Norway also had two representatives attending oil management courses in Norway already during the talks, including the first oil minister from the south. Hødnebø explains that a critical part of this assistance was that Norway assisted them in conducting an oil production analysis, so both sides, especially the

south, could know and trust the numbers and amount of reported production. He adds that the Norwegian oil specialist was included in the ministry in Khartoum and given full access to all the data and the oil specialist's overview of the production and the estimates of future production helped the parties plan how much production revenue could be included in future budgets. Hødnebø adds that assistance in good and proper taxation of oil companies was another topic that both sides were given courses in addition to environmental issues (interview with Hødnebø ,2020). Amin Hassan Omar, the Sudanese government's representative, confirms that during the negotiations, Norway promised to assist both sides in developing their oil sector further (interview with Hassan Omar, 2019).

By promising both parties help in developing the oil sector, Norway acted as a manipulator who employed incentives and offered carrots to the disputants.

Another example proves that Norway was a formulator, while Sumbeiywo, the IGAD representative, gives assurances that Norway did not impose its position on any of the parties, the government of Sudan and the SPLM (interview with Sumbeiywo,2019). Pagan Amum asserts that Norway the USA supports IGAD financially and with ideas. They were involved in discussions with the SPLM and the government of Sudan and were making proposals to the mediation process, discovering the disputant's common ground, the concessions that parties were willing to make, and the possible compromises the parties could agree on (interview with Amum, 2019). This was confirmed by Kjell Hødnebø, who shows that Norway did make a proposal to the disputants and helped them to find compromises: "Yes, of course, we did that, we were active in the negotiations," Hødnebø says (interview with Hødnebø, 2020).

Another example of the formulator role of Norway in the mediation between Sudan and South Sudan is the issue of self-determination. Hilde Johnson was aware that self-determination would be a deadlock in the negotiation. She could convince The US envoy, Danforth, to add self-determination to the negotiation substance. They are, together, asked General Lazaro Sumbeiywo, the primary mediator representing IGAD, to include the issue of self-determination in the negotiations (Johnson, 2016). In summary, in the mediation in the Sudanese conflict, Norway used the three mediation strategies in different positions in the mediation process. Norway acted as a manipulator and a formulator, not only a facilitator mediator. The Norwegians promise to help both Sudan and South Sudan to develop their oil sector further and contribute to increasing the recovery of their oil wells can also be viewed as manipulation. This is further proven by the fact that after the signing of the agreement, Norway helped both the South of Sudan and the government of Sudan to recover more oil and increase their oil production. In addition to being a facilitator, Norway introduced proposals for the disputants.

Impartiality/ biased

A norm-based mediator is expected to be an impartial mediator. However, from the first moment, Norway took a position to support the SPLM. Furthermore, on many occasions, the Norwegians changed the position of the Troika to the benefit of the SPLM. By supporting the SPLM, the weaker party, Norway, showed that it was not an impartial mediator in the mediation between the government of Sudan (GOS) and SPLM. However, Norwegian impartiality in the Sudanese conflict remains a

debatable issue. While some argue that Norway was not an impartial mediator as Norway was biased towards the southern (SPLM), others argue that Norway was an impartial mediator.

Despite asserting the importance of the mediator's impartiality, Hilde Johnson, the primary Norwegian representative, suggests that Norway did not have to follow the mediation rules formally because IGAD was the official mediator. Because of this, Norway's role was behind the scenes. However, she demonstrates that the mediator should have credibility and not take sides but should be in a relationship with both sides at all times (interview with Johnson, 2019). While Johnson did not consider Norway as a formal mediator, Hassan Omar, the Sudanese government representative, affirms that Norway among Troika controlled the negotiations: "Although Norway was acting under a united African effort to solve the problem in South Sudan known as the IGAD initiative, and it was this initiative which brought the two parties to the table with a Kenyan mediator presiding, the Troika—including Norway were controlling the negotiations from the shadows" (interview with Hassan Omar, 2019). General Lazaro Sumbeiywo, IGAD representative and the head of the negotiations, supported the argument that the Norwegians were not biased to any party and that they approached both sides equally (interview with Sumbeiywo, 2019).

Indeed, the representatives of the Government of Sudan were not happy with the Norwegian mediation because they found out that Norway was biased toward the SPLM (interview with Hassan Omar, 2019 and Elkhalifa, 2019).

Related to the Norwegian impartiality, Hassan Omar shows that the small Norwegian team led by a former Minister and an active politician, Hilda Johnson, were posing at

the beginning as advocates of one side rather than a mediator. Although they were made aware that their bias was apparent, the answer was always justification rather than a change of behaviour (interview with Hassan Omar, 2019). He confirms that Norway did not deal with both parties the same way. They did not bother to claim that the legal assistance and technical assistance to the SPLM was not disguised (interview with Hassan Omar, 2019).

Hødnebø, the representative of Norway in the mediation between the Government of Sudan and the SPLM, justifies the Norwegian support for the SPLM because the SPLM is the weak party (interview with Hødnebø, 2020). While Hødnebø, the representative of Norway, did not deny the Norwegian biased towards SPLM, Pagan Amum, the South Sudan representative, claims that Norway was an impartial mediator. He demonstrates that the people of South Sudan saw Norway as a committed and honest party that stood against injustice without internal interest (Amum, 2019). He asserts that Norway dealt with both parties as equal parties and that Norway chose not to have a position to support this or that party (interview with Amum, 2019). Additionally, he suggests that the government of Sudan itself accepted Norway because Norway was viewed as an impartial mediator (interview with Amum, 2019).

Pagan Amum contradicts himself when presenting an example where Norway supported the SPLM but not the Sudanese government. In the example, Norway provided John Garang, the head of SPLM, with a satellite phone. Moreover, Norway even paid the bills of that telephone which cost millions of Norwegian kroner. Norway did not do the same for the Sudanese government in Khartoum. The Norwegian diplomat, Hødnebø, and the NPA representative, Hanssen, agree with Pagan and

confirm that Norway did provide those phones to SPLM, but not the Government of Sudan. Moreover, the Norwegians paid the phone bills (interview with Amum, 2019; Hanssen, 2019; Hødnebø, 2019). Despite these examples, Amum continues to view Norway as impartial and supports the idea that Norway was an honest mediator and a good faith mediator (interview with Amum, 2019).

However, Amum refuses to describe this Norwegian behaviour as biased behaviour. Furthermore, he justifies the behaviour by asserting that John Garang was the principal, the negotiators needed to reach him, and it was impossible to reach him without the satellite phones because there were no telephone lines in Southern Sudan. In addition, he asserts that Norwegians saw the need for that and provided it and even paid the bills for communication alone. He adds that the Sudanese government in Khartoum did not need satellite phones because they already had their telephones (interview with Amum, 2019). Pagan Amum also defends Norwegian behaviour by asserting that "it was even facilitation of both sides as Norway treated both sides in even level" (interview with Amum, 2019).

Amum explains that the SPLM had no means of communication, and the other side had a means of communication (interview with Amum, 2019). Furthermore, he asserts that this is the Norwegian style of mediation, to bring the two parties to an equal level (interview with Amum, 2019). Similarly, Hilde Johnson justifies the necessity of satellite telephones; otherwise, how could the SPLM communicate. She claims that Norway offered satellite telephones to SPLM precisely because the SPLM lived in the bush, and they needed to communicate with their commanders in the field to consult on the security issues. Otherwise, they would have had deeper troubles. She points

out that there were no telephones at that time in South Sudan, while Khartoum had phones (interview with Johnson, 2019).

Likewise, Kjell Hødnebø, the Norwegian diplomat engaged in the mediation process, justifies the Norwegian behaviour by explaining that "when you have a government that has all resources and people who have nothing, you have to help the people who have nothing". He adds that the Norwegians helped the SPLAM because it was a pragmatic approach to keep the talks ongoing. Hødnebø says: "Our sympathy would always come down with those who are the weaker. We helped them, we assisted them, we even gave them courses so they can understand the art of mediation and how to negotiate" (interview with Hødnebø, 2019). He explains that by helping the SPLM, the weaker party, Norway is shortening the gap in power between the disputants (interview with Hødnebø, 2019).

Kjell Hødnebø refuses to call this biased: "You can call this biased, you can do that, but I would call it evening the playfield in the negotiations, so the people can meet in par and they can be equal when the talks start" (interview with Hødnebø, 2019). In international mediation, to support one party but not the other is called bias, not impartiality, no matter how many people try to justify it. This bias was confirmed by Abdelrahman Elkalifa, the Sudanese government representative. Abdelrahman Elkalifa shows that Western observers were sympathetic to the rebel movement (interview with Elkhalifa, 2019). He argues that the Norwegian NGOs were missionaries and were acting in line with that (interview with Elkhalifa, 2019).

Abdelrahman Elkalifa demonstrates that what the Norwegians offered was primarily and almost exclusively to the South, not the rest of Sudan, which was not their

concern. He adds that their humanitarian behaviour was earmarked for the South alone. He believes that the payment of John Garang's satellite phone bills was detrimental to the peace process and enhanced the rebel's capabilities. He affirms that Norway never did anything like this for the Sudanese government. He assures that Norway did not pay one krone to the government representatives (interview with Elkhalifa, 2019).

The Norwegian bias towards SPLM was restricted to logistical support and included humanitarian assistance. Focusing on the humanitarian support to the South of Sudan, Norway is accused of being partial towards SPLM, not only on the humanitarian level but also at a political level (Larssen, 2009).

While Pagan Amum explains that Norway used to provide humanitarian assistance to the needy people on both sides (interview with Amum, 2019), Hassan Omar asserts that Norway did not pay a penny to the GOS team. Additionally, he argues that Norway abstained from giving any humanitarian aid to the Government of Sudan because they considered the Government to be the villain while the SPLM was not seen as an insurgency, but rather the victim, according to Hassan Omar (interview with Hassan Omar, 2019).

He adds that the Norwegians were not providing only humanitarian assistance but also giving money that could provide arms and ammunitions (interview with Hassan Omar, 2019). In the same manner, Abdelrahman Elkalifa shows that rich Norway employed its economic resources too much in favour of the rebels, but it did not employ such resources to the Sudanese government (interview with Elkhalifa, 2019).

The Norwegian bias towards the SPLM was represented in the satellite phones or the humanitarian assistance and Norwegian support of political issues related to SPLM, such as self-determination and the gathering of the SPLM factions. The self-determination right for the Southern Sudanese is an issue that could bring Norwegian impartiality into question. When Hilde Johnson met Jack Danforth, the US special envoy to Sudan, Hilde Johnson was aware of the deadlock between the Sudanese government and the SPLM. The problem was the self-determination of the Southern Sudanese. Hilde succeeded in convincing Danforth of the importance of South Sudanese self-determination, and they together communicated to General Lazaro Sumbeiywo, the primary mediator representing IGAD, to include and focus on the issue of self-determination in the negotiations (Johnson, 2016).

This action could be understood as Norway was adopting the position of SPLM, which indicates that Norway relinquished its impartiality. Hilde Johnson disagrees with this opinion. She says: "This is nothing to be partial about". Furthermore, she justifies," if you want to find a solution to a conflict, you have some basic redlines that the parties will never go beyond". She suggests that they would never sign up anything if those red lines were not there"(interview with Johnson, 2019). She recommends that the mediator should identify and understand the red lines of the parties. She considered self-determination a redline for the SPLM (interview with Johnson, 2019). She used the same argument as that of the SPLM but not those arguments put forth by the Sudanese government in Khartoum. She justified her opinion by saying that "if you do not address the redlines, you will not get agreement and you will end up into war" (interview with Johnson, 2019).

The second political issue that could be viewed as evidence of the Norwegian bias toward the SPLM, is the gathering of SPLM factions. During the negotiations, the SPLM was split into seven factions. Norway tried to unify them. Norway talked to all the factions, and the Norwegian NGOs paid for a big gathering of people to come together in a peace camp (interview with Hødnebø, 2019).

All of the above could prove that Norway was not an impartial mediator and was biased to SPLM. However, as mentioned above, Norway is always ready to justify its action and exclude itself from being biased. In rare cases, the Norwegians admit their bias. For example, Kjell Hødnebø and Halle Hanssen, Norwegian mediator and representative of NPA, admit the Norwegian partiality to the SPLM. "We were one-sided; we were overly critical to Khartoum. We supported the SPLM. We stood very firmly behind the liberation movement" (interview with Hanssen, 2019). Hødnebø confirms that Norway supported the SPLM, saying, "Norway always stands with the side of the weak party because we have been in a weak situation for 500 years" (interview with Hødnebø, 2019).

Despite his admitting to being biased, Hødnebø, still justifies the Norwegian position in the mediation process by claiming that Norway was impartial: "We tried to be even, we had meetings with both sides and tried to persuade both sides. We tried to build personal relations with both sides, the SPLM and the Sudanese government. ..So in building up personal relations with both sides, we tried to be even, and this is extremely important, especially in Africa and in the Middle East because in those countries if you are friends, you can trust people" (interview with Hødnebø, 2019).

In sum, there is strong evidence that Norway was biased towards the SPLM. Norwegian support to the SPLM is apparent. Norway supported them both financially and politically by adopting their position. Norway provided the SPLM with satellite phones but did not do so for the government of Sudan. On the political level, Norway's adoption of the position of SPLM is related to its self-determination. Self-determination was an essential demand for SPLM. Norway succeeded in convincing the members of the Troika of the importance of this demand and included it in negotiations. Additionally, Norway was working on gathering the fractions of rebels. The Norwegians justify their impartiality because they support the weak party, the SPLM.

CHAPTER SIX: CASE STUDY THREE: THE DOHA DOCUMENTS FOR PEACE IN DARFUR (DDPD),2011

Introduction

The conflict in Darfur since 2003 has created a complex humanitarian reality, killing and displacing thousands of people. Meanwhile, Sudan and the Darfur civil war have witnessed several attempts at international mediation to find a solution for the conflict. Qatar was at the forefront of the initiative to mediate and find a solution for this complex conflict. The State of Qatar contributed to resolving the conflict in the Darfur region of Western Sudan. Doha sponsored the negotiations between the Sudanese disputants and hosted the interlocutors for more than two years culminating in the Convention, or so-called Doha document for Peace, which was the product of a dialogue involving hundreds of stakeholders. In May 2011, Doha hosted the celebration of the Darfur Peace Agreement, which laid the foundations for security and stability there.

Qatar's efforts at the Convention were part of a significant project that would restore stability and development to Darfur and remove the legacy of war and tragedies of migration and displacement that came with it. The agreement included establishing Darfur development Bank with a capital of \$2 billion, to which Qatar committed itself and the parties to the Convention to follow up on the implementation of the provisions of the agreement. The Qatari effort to expand regional and international interest in the Convention continued, culminating in the Doha 2013 conference for the development and reconstruction of Darfur, in which USD 3 billion were pledged in support for the region. Qatar contributed the largest share.

The main argument in this chapter is that Qatar used a range of mediation strategies, facilitation, formulation, and manipulation strategy in the mediation Darfur conflict.

However, it preserved impartiality (not being biased in favour of one party) as a necessary component in the Qatari mediation. In order to illustrate this argument, this chapter sheds light on several topics: the Darfur conflict background, the Qatari role in mediation between the Sudanese disputants, and e the application of Qatari mediation throughout the Darfur conflict.

Darfur – Sudan conflict

In order to understand the conflict in Darfur, it is necessary to go back and review the history and the societal structure of Darfur. Darfur is a western Sudanic society linking the desert to the savannah (Burr & Collins, 2008). It hosts a diverse range of ethnic groups and tribes (Flint & De Waal, 2008). The ruling clan in Darfur was the Keira dynasty. Darfur itself is centred in the northern mountain region known as Marrah. The states of Darfur expanded their authority to the south to include the farming communities which adopted the Fur language. All Darforians are Muslims, and Islam is a state cult in Darfur. Most Darforians follow either the Tijaniyya Sufi sect, which originates in Morocco, the Ansar of the Mahdi, or both. In the fourteenth century, the Arabs came to Darfur. They were in two groups: the first group included scholars and traders and originated from different parts of the Arab world —the east and the west, the Nile and Arabia, the Maghreb, and West Africa. The second group of Arabs, known as the Juhayna Bedouins, came from the northwest searching for grass and water.

The largest and most powerful Arab tribe is the Rizeigat, who live in south-east Darfur under the tribal authority of the Madibu family (Flint & De Waal, 2008). Darfur was an independent state or sultanate until 1916 but was then absorbed into the British Empire in January 1917 (Flint & De Waal, 2008). Darfur was gradually and peacefully absorbed into the British- Egyptian Condominium. British power affirmed the Fur, which encouraged commerce and eased agricultural activities in Darfur. However, the British paid very little attention to investment in infrastructure and public services in Darfur (Natsios, 2012). In 1956, Darfur Became a part of the Republic of Sudan (Waihenya, 2006).

By the second half of the 19th century, Darfur had suffered several periods of famine, which led to political instability and outbreaks of war with temporary periods of peace. In the mid-1960s, 1970s, and again in the mid-1980s, drought-induced famines caused political unrest that eventually reached Khartoum. When famines strike anywhere in the world, especially when the government does not have the tools to cope with famine, ethnic conflict, violent crime, and population movements of the starving are soon to follow. This often results in political unrest, violence and even civil war (Natsios, 2012). The failure of the successive Sudanese governments to deal with those famines and to respond to the human suffering they caused was the main reason behind the civil war in Darfur (Natsios, 2012). Darfur witnessed three civil wars: the first took place between 1987 and 1989 and pitted the Arab tribes against the Fur tribes. The famine of the mid-1980s paralleled this civil war.

A few days after the Al- Bashir coup in July 1989, the first peace settlement was signed between the Arabs and the Furs. The second peace agreement was signed in

December 1989. However, the two agreements were not implemented, which exacerbated bitterness toward the Khartoum government.—Moreover, in 1994, Khartoum split Darfur Province into three new federal states within Sudan, which added to Fur frustration. Additionally, the neglect concerning economic development in Darfur accompanied the ongoing political discrimination of the non-Arab in Darfur. All this led to the second civil war in Darfur between 1995 and 1999. It erupted this time between the Darfuri Arabs and the Masalit tribe residing in Western Darfur. The Masalit, like the Fur, were African Muslims who saw themselves as the dominant power in Darfur. When Al-Bashir became president of Sudan, he started with the Masalit, considering them adversaries. Al-Bashir appointed a new local Arab authority in the Masalit area and gave that authority the right to appoint a sultan, who certainly should be Arab. All these tactics by Bashir led to the eruption of the second civil war in Darfur in 1999 (Kevane & Gray, 2008).

In 1999, Al-Bashir became aware of the flaw in his policy and decided to restore some form of representational balance to the council of the local government in Western Darfur by appointing an equal number of Arab and Masalit. The first two civil wars, the Arab-Fur and the Arab-Masalit conflicts happened because of the marginalisation of the non-Arab tribes and the widespread distribution of automatic weapons into the hands of the Arab militias. All this took place in a setting of chronic under-development with an encroaching northern desert (de Wall & Flint, 2005).

The third civil war erupted mainly because of a book, a black book, which was published in May 2000. The book was entitled The Black Book: Imbalance of Power and Wealth in Sudan. This book refers to research conducted by the Darforian rebel

group, the Justice and Equality Movement (JEM). The research was focused on the Sudanese elite who had ruled Sudan since 1956 and attempted to shed light on their ethnic and geographic origins. (De Waal, 2007a; Mans, 2004). The book asserts that the ruling elite is restricted to only three tribes of the Northern Nile who are the Shai-qiyya, Ja'aliyiin, and Danagla. Those three tribes stand for 5.4 percent of the Sudanese population but dominate 70 percent of the political, educational, social and economic Sudanese institutions. These institutions include banks, universities, the military officer corps, the government's civil service, the police and internal security apparatus, the judiciary, corporations, and the media. That is why the book played a significant role in inflaming the anger of the non-Arab tribes in Darfur.

This anger ultimately resulted in the union of the main tribes in Darfur: The Fur, Masalit, and Zaghawa. Together they formed a military alliance driven by four historical issues: the marginalisation of the periphery; poverty and underdevelopment; human rights abuses by Arab supremacist groups; and the absence of democratic institutions (De Waal, 2007b; Flint, 2010; Mans, 2004). While the real beginning of the third civil war in Darfur was February 2003, it started in July 2001, when Fur and Zaghawa leaders pledged to resist the Arab movement across Darfur (De Waal, 2007a; Mans, 2004). In February 2003, the rebels in Darfur formally announced the beginning of a rebellion (Dagne, 2010; Faris, 2007). The prominent anti-government rebels in Darfur included the Sudan Liberation Movement/Army (SLM/ SLA) with its broad base of support across Sudan's major ethnic groups (principally non-Arab but including some Arabs), and the Justice and Equality Movement (JEM), whose leaders have links with Sudan's Islamist movement (De Waal, 2007b; Mans, 2004).

The Darfur civil war was not a North-South or Muslim-Christian conflict. It came about when local Muslim peoples rebelled against pro-government militias known as Janjaweed (Mans, 2004). The Janjaweed militias are a segment of Darfur's camelherding Arab tribes; they are Arab immigrants of Chadian origin. (De Waal, 2007b; Flint, 2010; Mans, 2004). The Janjaweed have often financed themselves through robbery and pillaging (Mans, 2004). Despite this, they have enjoyed support from the government of Khartoum, with whom they have an agreement allowing them to pursue their agenda with impunity in exchange for suppressing outbreaks of rebellion (De Waal, 2007a; Mans, 2004). The Sudanese government and the Janjaweed practised systematic and large-scale destruction, targeting any life in Darfur (Nathan, 2006).

Towards the Doha Document for Peace in Darfur (DDPD)

The Qatari mediation was not the first mediation trial in the Darfur conflict. Since the eruption of the civil war in Darfur, there have been several attempts at mediation between the Government of Sudan in Khartoum and the rebels in Darfur. In late 2005 the seventh round of the Inter-Sudanese Peace Talks on the Conflict in Darfur commenced in Abuja, Nigeria, under the auspices of an African Union (AU) mediation team and the support of the UN, the UK and the USA, alongside other international partners. The talks aimed to arrange a comprehensive peace agreement between the Government of Sudan and the main rebel movements in Darfur: the Sudan Liberation Movement/Army (SLM) and the Justice and Equality Movement (JEM).

On 5th May 2006, the Darfur Peace Agreement (DPA) —known as the Abuja Agreement-was signed by the Government and by Minni Minawi, the leader of one of the

two SLM factions, but was rejected by JEM and Abdel Wahid al-Nur, the leader of the other SLM faction (SLM was split into two factions soon after its establishment). Thus the Agreement did not achieve peace (Nathan, 2006). The two groups which refused to sign (JEM and SLM-Wahid) continued fighting against the Sudanese Government in Khartoum, even fighting with one other in some instances resulting in the fragmentation of the group into almost 27 different rebel factions, every group claiming itself to be the representative of Darfur (Dagne, 2010). In 2008, the violence escalated to new heights, and the dispute reached a critical stage (Kamrava, 2011) which required further attempts of mediation in Darfur (Barakat, 2012). Amongst these renewed peace efforts was the Qatari mediation, which will be discussed in detail in the following sections.

Qatar was nominated to be the representative of the Arab League to mediate between the government of Sudan and various rebel factions. Barakat (2012) argues that this nomination was due to the long-established Qatari relief effort in Darfur via its Red Crescent Society (Barakat, 2012). However, this nomination would not have materialized without the many efforts of the country's Minister of State for Foreign Affairs, Ahmad bin 'Abdullah Al-Mahmud (Kamrava, 2011). In 2008, Minister Al-Mahmoud travelled to many countries around the world to meet the international officials experienced in the Darfur conflict and the conflict's stakeholders. He met with the officials from the US state department, The African Union, the United Nations, the Sudan Government, Darfur representatives and many others. His goal was to collect information about the perspectives of Darfur conflict resolution and guarantee the approval of those countries and organizations to pave the way for Qatari involvement in the mediation process. In addition, this helped to accomplish an even more important

goal; building trust between Qatar and the disputants and motivating them to accept Qatar as a mediator (Kamrava, 2011).

This is how Qatar manages to involve itself in mediation. It involves a calculated decision, made only after ensuring that the international community approves its mediation or does not disapprove (Kamrava, 2011). It is important to remember that Qatar was not alone in its mediation of the Darfur conflict. Although Qatar was the Arab league representative, other international actors were also included in the process, especially the African Union (AU) and UN mediators. AU and the UN created a Joint Mediation and Support Team (JMST) with the support of the USA (International Crisis Group, 2007). After failed mediation tries by JMST, JMST turned to Qatar, allowing it to become involved in the mediation.

In 2009, JMST and Qatar began new negotiations between the disputants (Aljazeera Channel, 2009). Qatar held a distinct role as a significant mediator (The Enough Project, 2010; Williams & Simpson, 2011) as the generous host of the negotiations with representatives from both sides. Qatar invited various parties to the talks- including Chad, Libya, Egypt, the Arab League, the African Union and the United Nations and hosted them in Doha's luxury hotels for months at a time (Barakat, 2012; Doherty, 2010; Kamrava, 2011; Roberts, 2010).

In 2009, Qatar and JMST made headway by convincing the largest rebel group in Darfur, the Justice and Equity Movement (JEM), and the Sudanese government in Khartoum to sign a memorandum of understanding (MOU). In the MOU, both sides committed themselves to make peace (Jones, 2011; Kamrava, 2011; Ulrichsen, 2013). However, some Darforian rebels rejected involvement in the negotiations as

they were to take place in Qatar, an Arab country. In addition, they did not trust the Sudanese government in Khartoum. However, this vision of the rebels has changed over time (Aljazeera Channel, 2009).

In February 2010 and under Qatar's sponsorship, the Sudanese's government and the Justice and Equality Movement (JEM) signed the power share agreement in Qatar (Hancock, 2010). The agreement formulated the roadmap for further negotiations in the future and included some principles relevant to the cessation of hostilities (Dagne, 2010). However, the agreement did not include all rebel groups. In his reaction to the agreement, SLM leader Abdelwahid Nur, who refused to engage in the negotiations, asserted that the agreement was an empty promise and that the Khartoum government specialised in signing and not implementing them (Aljazeera Staff writer, 2010).

In March 2010, The Sudanese government in Khartoum and the Movement of Liberation and Justice (LJM) signed two additional documents in Doha in the same year. One was a framework agreement to resolve the Darfur conflict, while the other was an LJM ceasefire (Dagne, 2010). The Sudanese official, Ghazi Salahudin, represented the government of Sudan. Meanwhile, Al-Tigani Sessi represented the Movement of Liberation and Justice (LJM). Indeed, LJM is an umbrella group consisting of 10 rebel groups united in February 2010 under the Movement of Liberation and Justice (Hancock, 2010). The negotiations and the mediation were complex, and Qatar was the leading player in the mediation with JEM and non-JEM.

In 2011 and after two and a half years of negotiations, The Doha Document for Peace in Darfur (DDPD) was finalised at the "All Darfur Stakeholders Conference". In

July 2011, the Sudanese government signed a protocol agreement and the Liberation and Justice Movement (LJM). Both sides committed themselves to the Doha document resulting from the comprehensive peace process in Darfur (DDPD), which included the conflict causes and its articulation such as Power Sharing, Wealth Sharing, Human Rights, Justice and Reconciliation, Compensation and Return, and Internal Dialogue among others (UNAMID, 2011).

The Qatari International mediation approach for Peace in Darfur

This section focuses on three components: Qatari motivation, a manipulation strategy, finally, impartiality.

The Qatari motivation

The point of departure for the Qatari mediation is the Qatari interest. The Qatari interest is focused on Qatari security. Security could have different forms such as food security, financial and economic security (investment), branding itself and security and stability in the Arab and Islamic region (Chacko, 2009; Minich, 2015; Roberts, 2015). The invasion of Kuwait by Iraq caused Qatar to feel insecure and inspired Qatar to strengthen itself by changing its foreign policy and taking mediation as a tool to extend its influence and, consequently, preserve its security.

Taguia, the main researcher of Aljazeera Research Center, confirms that Emir Hamad initiated a new strategy in Qatar in 1995. The Emir decided that Qatar should be independent of Saudi Arabia and have its own vision. Taguia explains that the Emir believed that relying on Saudi Arabia would not offer security to Qatar. After the Iraqi invasion of Kuwait, the Emir realised that Qatar was not secure anymore.

Additionally, because of the location of Qatar between Saudi Arabia and Iran, the two regional hegemonies encouraged the Emir to adopt an independent foreign policy to maintain the security of Qatar. Taguia confirms that mediation, in general, was one of the tools that the Emir decided to adopt in Qatari foreign policy (interview with Taguia, 2019).

In the same manner, Lulwa Al-Khater, the Qatari ministry of foreign affairs spokesperson, shows that Qatar's interest is in its security. She asserts that security for any country is one of the highest priorities. Qatari foreign policy is based on one rule, which is "no harm", Al-Khater confirms (interview with Al-Khater, 2019).

Indeed, international mediation benefits Qatar and strengthens its relations with the countries where Qatar played the role of mediator. Taguia explains that through mediation, Qatar can build trust between it and other countries where Qatar plays the role of mediator, and in turn, those countries can favour Qatar. Taguia presents an example of the siege of Qatar in 2017. He illustrates that what has happened during the siege of Qatar is based on the Qatari mediation in Darfur. He explains that, during the siege, President Al Bashir refused to be biased towards Saudi Arabia and the United Arab Emirates in the Qatari siege. Al Bashir appreciated the Qatari mediation and prevented Sudan from being split. Taguia assures that Qatar can secure goodwill for itself through mediation among other countries. He adds that world peace is essential for the international community. Additionally, the mediation is essential for Qatar because it introduces Qatar to the international community as a peace broker that does not support terrorism (interview with Taguia, 2019).

Some scholars show that Qatari food security is the main reason behind the Qatari mediation in the Darfur conflict. They base their suggestion on Qatar's need to import 90% of its food demands. To secure its food and in line with its strategic economic interests, Qatar has planned to establish extensive contacts with Sudan (Chacko, 2009; Minich, 2015; D. B. Roberts, 2015). Qatar began to support Sudanese agricultural projects to produce wheat, meat, vegetables, fruits and sugar (Open Briefing, 2014). Furthermore, in 2008, a joint company for agriculture investment, food industry, and animal husbandry was established by the agreement of Doha and Sudan (Farmlandgrab, 2008). The Qatari Hassad Food's agriculture company has contributed to the cultivation of 260,000 acres in the River Nile state. Additionally, the Widam Food Qatari company focuses on livestock and butchery, which exports to Qatar (Gulf Times staff writer, 2017). Farmlandgrab demonstrates that this was one of the many efforts from Doha to ensure Qatari food security (Farmlandgrab, 2008). In response to Qatar's suggestion that a food security interest in Darfur, Taguia argues that Qatari mediation was for no specific reason. He says," It was not based on an interest in Qatar; it is not true that Qatar mediation in Sudan was due to food security." Taguia affirms that Qatar's agriculture investment is located in Kenya, not Sudan. He adds that the Qatari investment in Sudan is not for the benefit of Qatar; it is for the benefit of the stability of Sudan, and the actual Qatari investment in Sudan is not feasible for Qatar because of the absence of stability there (interview with Taguia, 2019).

In the same manner, Al-Khater says that "it is not logical to claim that Qatar decided to be a mediator in Darfur because of economic reasons or food security." She asserts that, after considering the time, the resources, and the efforts necessary, it

would be easier for Qatar to invest in Darfur and Sudan without involving itself in mediation. She adds that there are many countries who have direct investments in Sudan through which they receive huge profits. Al-Khater states: "We have to be clear that Qatar has two types of investments: real investments where Qatar has revenue, and this is found in Europe, USA, and Asia. The other type of investment is an investment in aid and development and the Qatari investment in Darfur and Sudan is an example" (interview with Al-Khater, 2019).

Lolwa Al-Khater explains that the state of Qatar views the mediation not only as a commitment to the absence of violence or the signing of a peace agreement, but also Qatar aims to achieve sustainable peace in the conflict zones. She adds that to achieve sustainable peace, Qatar believes that development should accompany mediation, and this development includes development in education and health. Additionally, economic recovery, jobs creation, and rehabilitation of infrastructure are essential goals that Qatar worked on after reaching an agreement (interview with Al-Khater, 2019).

In the same vein, the ex-ambassador for Qatar in Sudan, Ali Al-Hamadi, claims that Qatar is not seeking financial or investment benefits due to its mediation. However, on the contrary, these mediations cost Qatar much money, especially in Darfur, where Qatar has pledged to provide material and economic assistance and improve people's standard of living to create a lasting and sustainable peace. He suggests that the country's strategy in mediation is based on the state's foreign policy, which focuses on several principles such as; the principle of consolidating international peace and security by encouraging the peaceful resolution of international disputes,

supporting the right of peoples to self-determination, not interfering in affairs in the interior of states, and cooperation with peace-loving nations (interview with Al-Hamadi, 2019). Al-Hamadi adds that Qatar considers mediation a necessary strategy that would decrease the external threats such as terrorism and the displacement of peoples (interview with Al-Hamadi, 2019).

Tigani Sisi claims that Qatar had no interest in mediation in the Darfur conflict, saying: "I have not come across any matters that indicates Qatar had any interest other than achieving peace in Darfur" (interview with Sisi, 2019). Likewise, Amin Hasan Omar, the Sudanese government's main negotiator in the Darfur negotiations, also asserts that Qatar had no interest in the mediation in the Darfur conflict (interview with Hassan Omar, 2019). It is expected that the Darfur representative and the Sudanese government's main negotiator in the negotiations would not confirm any specific interest for Qatar in Darfur.

Reviewing the Qatari development aids and investment after signing the Doha Document for Peace in Darfur (DDPD), and during the years 2012 and 2013, Qatar funded Sudan in several ways, including: official development assistance (ODA)—grants and concessional loans), state and private investment, and state and charitable humanitarian assistance. For example, in 2012-2013, Qatar invested 2 USD billion by purchasing multi-sector treasury bonds from the Central Bank of Sudan (Open Briefing, 2014).

Table 1. Diplomatic events covering development assistance from Qatar to Sudan in the period 2012-13.

Event	Date	Amount	Allocation	Туре
Diplomatic visits between Sheikh Tamin and President Bashir	Feb 2012	\$2 bn	Treasury bond purchase – varied (e.g. mining, oil and agriculture)	Investment (pledged, contribution unconfirmed)
Meeting of International Mechanism for Follow up to Implementation of Doha Document for Peace in Darfur	May 2012	\$31 mn	Health centres, schools and development projects	Humanitarian assistance (pledged, contribution unconfirmed)
		\$6 mn	Regional Authority of Darfur	Humanitarian assistance (pledged, contribution unconfirmed)
Visit/inaugural ceremony	June 2012	\$208 mn	Electricity line carriers project	Investment – project finance
Nile Valley Forum	Sept 2012	\$135 mn	Tourism and antiquities	Investment – loan
Doha Conference	Арг 2013	\$500 mn	\$88 mn contributed up front	Mixed ODA and concessional loans
		\$200 mn	Darfur Development Bank capital	Concessional loans and grants

Source: Open briefing: https://www.openbriefing.org/docs/Qatari-aid-finance-and-foreign-policy-in-Sudan.pdf

Indeed, the Qatari fund in Sudan and Darfur was for investment and development and to elevate poverty. Parallel to the Qatari mediation in Darfur in June 2010, the Qatar Charity announced a project of USD 1 billion to improve the livelihoods of 65,000 impoverished people living within the affected area. This project contributed to decreasing the level of poverty in Darfur by funding training for farmers and the rehabilitation of the land (Dabanga, 2010).

Qatar is motivated by its security in international mediation, including its mediation in the conflict of Darfur. Indeed, This appears in the position of AL Bashir, the president of Sudan, who did not support the siege on Qatar and did not cooperate with Saudi Arabia and UAE to cut diplomatic relations with Qatar.

In a different vein, certain scholars argue that the Qatari motivation in mediating the Darfuri conflict is based on religious and cultural considerations. Al-Hamadi explains that Qatar considers mediation efforts a "moral, cultural and religious duty" towards other peoples (interview with Al-Hamadi, 2019). Indeed, in its mediation, Qatar aims to achieve the security and stability of the Arab and Islamic region. Sudan is both an Arab and Muslim country. Qatar has a record of engaging itself with Muslim countries because Islam is an integral part of Qatari identity. It supports symbolic Islamic issues as Qatar does not want to see Muslims kill each other (Hansen, 2013). First and foremost, Qatar is a firmly Muslim country where Islam covers all aspects of state affairs (Roberts, 2012). However, this does not mean that the Qatari work for peace is restricted to Arab and Muslim countries. This may be flexible and, at times, doubtful. However, it is necessary to mention that the Qatar foreign policy is built on a principle of flexibility and does have a diverse relation to various parties and countries. For Example, Qatar has relations with Hamas and with Israel (Rabi, 2009).

Qatar chose to meditate on the Darfur conflict because the Sudanese and Darfuris are Muslims. To reconcile between Muslims is a duty for the Qataris as Qatar has cultural-religious ties with the government of Sudan and Darfur. The Qataris care about peace and stability in Sudan for many considerations, such as Qatar and Sudan's cultural, Islamic, and Arab ties. In public statements, pan-Arabism and Islam emerge as crucial elements of Qatar's foreign policy (Zureik, 2018). This was confirmed by Sheikh Tamim bin Hamad Al Thani in 2013 when he described Qatar's

active foreign policy as such: "We are Muslims and Arab; we respect the diversity of religious schools of thought and respect all religions in our country and abroad. As Arabs, we reject dividing the Arab communities based on sectarianism or doctrine because this protects social and economic immunity and prevents its modernization and development based on citizenship regardless of religious sects or thoughts" (Kaussler, 2015a, p5).

Ali Al-Hamadi, the ex-ambassador of Qatar in Sudan, argues that the security of Sudan is essential for the Qataris because Qatar has close, strong, and long-standing relations with the Sudanese people, which have been woven through their Arab, Islamic, and humanitarian ties. The Sudanese community is one of the oldest Arab communities in Qatar, and it operates in all fields. This interconnectedness has led to strong ties between the Qatari and Sudanese peoples. In addition to the privileged relations between the Qatari and Sudanese governments, everything that is happening in Sudan is felt in Qatar; hence the Qatari intervention of mediation in Darfur is not new to Qatar, as it has always supported Sudan (interview with Al-Hamadi, 2019). Ali Al- Hamadi demonstrates that Qatar hosts negotiations between conflicting parties or plays a key role in their dialogue in order to maintain regional peace and security (interview with Al-Hamadi, 2019). In the same manner, Lulwa Al- Khater, the Qatari Ministry of Foreign Affairs spokesperson, states that Qatar's main goal is the stability of the area as a whole, and Sudan is a part of this area. The Qatari mediation in Sudan is based on humanitarian considerations and regional security and stability as well (interview with Al-Khater, 2019). In its mediation, Qatar seeks real, sustainable stability which is based on development and society building (interview with Al-Khater, 2019).

Related to the claim that Qatar's involvement in the Darfur conflict was based on religious considerations, Amin Hassan Omar, the representative of the Sudanese government in the negotiations, describes this claim as nonsense. He asserts that Qatar was there because the Arab League elected five Arab countries chaired by Qatar to deal with the Darfur file, and Qatar was representing the Arab league while the joint representative was representing the UN and the AU (interview with Hassan Omar, 2019). Likewise, Altigani Sisi, Darfur representative in the negotiations, states that the joint mediation has given the Doha Forum much more credibility (interview with Sisi, 2019).

The Qatari mediation efforts' history shows that most of the Qatari mediation is among Muslim or Arab countries. This is not to say that Qatar mediation is restricted to the Arab and Muslim countries. Qatar seeks to promote peace on the regional and the international level. However, Qatar may prioritize its mediation in The Middle East and Arab countries. This could be because this area has lots of conflicts, which may affect the security of Qatar in case the violence spills out. Another motivation among the Qatari motivations in international mediation is branding Qatar as a peace broker. Lyon (2008) demonstrates that state branding is a strategy to multiply and amplify the efforts of Qatari foreign policy and influence. Qatari mediation takes place in the spot-light and often in front of the local and regional media. He adds that senior Qatari diplomats even go as far as to interview with the media while the mediation is still underway, emphasizing the positive role of Qatar in a manner that is clearly aimed at branding (Lyon, 2008).

Kamrava (2011) confirms that one of the other purposes for Qatar's mediation is branding itself with the image of an experienced mediator, regional diplomatic power-house and an impartial peace broker (Kamrava, 2011).

The Aljazeera channel is Qatar's tool for projecting its image. In the Darfur crisis, Aljazeera covered all the events there daily (Minich, 2015. A common criticism is that Qatari's focus on branding in its mediation sometimes seems to outweigh its focus on resolving the conflict. Dickinson (2012) claims that Qatar has been more concerned with getting a deal than ensuring it is a deal that is practical and able to address the reality appropriately (Dickinson, 2012b)..

In response, Al-Khater (2019) did not agree with the branding motivation. She demonstrates that Qatar, with its wealth, can establish strong branding and public relations without being a mediator in Darfur or any other place. Qatar would be more accessible and more feasible to hire private companies for its public relation and branding campaigns. She adds that Qatar chooses mediation for the promotion of peace and stability in the area (interview with Al-Khater, 2019).

In brief, Qatar engagement in mediation Darfur conflict is motivated by several motivations; Qatar security, branding Qatar as a peace broker, and finally maintaining the stability and the security of the Middle East, which is considered the security of Qatar. Paying billions of dollars to the Sudanese to achieve a peace agreement can brand Qatar. Achieving peace and helping to stop war could be a great tool and a solid message to brand Qatar in the international arena and secure Qatar.

The Qatari leverage

Qatari leverage in Darfur mediation is represented in three sources: gratification in the form of Qatari financial incentives, which Qatar promises to introduce to the Sudanese for building their infrastructure; personal diplomacy represented in the Emir and the elite; and finally, the cultural and religious ties. The first Source of the Qatari leverage is the gratification where Qatar introduced incentives for both Sudan and Darfur. Financial incentives are rewards that the mediator offers to the disputants in return for resolving their conflict. While Schrodt & Gerner (2004) assert that the financial incentives introduced to the weaker party lead to a decrease in conflict tensions (Schrodt & Gerner, 2004), Böhmelt (2010) confirms that the more incentives the mediator has at their disposal, the more likely they are to apply enforcing strategies and reach an effective outcome (Böhmelt, 2010).

Qatar uses financial resources as a leveraging tool. In the Darfur case, Qatar promised to invest a further USD 2 billion to address the chronic underdevelopment in the Darfur region and the creation of a Darfur development bank, if talks were successfully concluded (Sudan Tribune, 2011a; Walid, 2009). Although Qatar was not alone in its mediation in Darfur (the UN and AU were working alongside them), Qatar did play a distinctive role by using financial incentives. Additionally, Qatar was able to generously host the disputants in Doha along with the delegates from different parties (Chad, Libya, Egypt, the Arab League, the African Union and the United Nations) for long periods of time (Doherty, 2010; Roberts, 2010). It is worth mentioning that the Qatari investment in Darfur and Sudan, in general, is for the benefit of Sudan and

Darfur, aiming to rehabilitate the infrastructure in Darfur and Sudan in general (interview with Taguia, 2019).

It is essential to mention that Qatar used financial leverage in different stages of the mediation process. Parallel to the Qatari mediation in Darfur in June 2010, the Qatar Charity announced a project of USD 1 billion to improve the livelihoods of 65,000 impoverished people living within the affected area (Dabanga, 2010). By the end of the negotiations and to encourage the disputants to reach an agreement, Qatar promised to invest a further USD 2 billion, if talks were successfully concluded (Sudan Tribune, 2011a; Walid, 2009).

The second source of leverage is personal diplomacy. Usually, the Emir or the elites take part in the mediation process. Qatar's involvement in the Sudan conflict began with the preliminary efforts of Qatar's Minister of State for Foreign Affairs, Ahmad bin 'Abdullah al-Mahmud, in 2008. Qatari mediation was recognised as being extremely character-driven while heavily relying on the efforts of the country's Minister of State for Foreign Affairs (Kamrava, 2011). Al Mahmoud contacted the people directly involved with the Sudanese government as well as the people of Darfur and the conflict stakeholders (Kamrava, 2011). As mentioned before, he travelled to all countries considered primary or second-tier stakeholders to collect information about the Darfur conflict and their preferences for resolution and seek their approval for Qatar involving itself in the mediation process alongside the UN and African Union. Moreover, he was in contact with the international humanitarian organizations on the ground in Darfur and even travelled to Darfur to acquaint himself with locals to be

more aware of the reality and gain the trust and confidence of the darforians needed for Qatar to be accepted as a mediator (Kamrava, 2011).

The mission of the Qatari minister was not easy, especially among Darforians, many of whom distrusted Arabs after their mediation experiences with Egypt and the Arab league. The general Darforian consensus was that Arabs were biased towards the Sudanese government in Khartoum. Despite these adversities, the Qatari minister succeeded in generating among Darforians a sense of goodwill towards Qatar. Kamrava (2011) claims that Darforian representatives accepted Qatari mediation because the Qatari mediators offered a new and different perspective on the Sudanese conflict. This perspective was developed in three stages. Firstly, Al-Mahmoud secured the approval of the second-tier stakeholders. Next, Al-Mahmoud collected extensive first-had information about the conflict, using the international humanitarian organizations on the ground as a source. He even met Darforian people and listened to their accounts of the conflict to avoid bias. Finally, the conflicted parties were invited to Doha for negotiations (Kamrava, 2011).

Personal diplomacy is a significant factor in the Qatari mediation. The Qatari personal diplomacy and the direct intervention in the mediation process by the Emir or the Minister of Foreign Affairs forced the disputants to reach an agreement significantly contributed to the Qatari mediation's success. Altigani Sisi, the main negotiator representing Darfur, asserts that the presence of Emir is a tool to promote the disputants to reach an agreement. He adds that the negotiations would not have succeeded without the Emir's efforts, determination, political will, and follow-up. Altigani says that "Emir's intervention within the Arab League of States and the UN culminated in establishing the Doha forum as the only regionally and internationally

recognised forum for negotiating the Darfur problem" (interview with Sisi, 2019). He explains that "the Emir and the prime minister played a role by using their good offices to build the trust but never to apply pressure" (interview with Hassan Omar, 2019). However, Qatar has been criticised for relying too heavily on the personal attributes of its mediators, notably the Emir and the Foreign Minister (Roberts, 2011). The third type of Qatari leverage is cultural and religious ties. Qatar shares the same religion with Sudan's Government and the Darfuri rebels. All the parties are Muslims. Qatar shares the same culture with the Government of Sudan. Both Qatar and Sudan are Arab. The Qatari Arabic culture is an advantage because Qatar uses Arabic culture to frame a resolution for the conflict (Mac Ginty, 2008b). Unfortunately, this is not always beneficial. One of the main reasons why the rebels rejected Qatar as a mediator at the beginning was precise because Qatar is an Arab country. The rebels had a bad experience with Arab mediators such as Egypt, who appeared to be biased towards the Sudanese government in Khartoum (Kamrava, 2011). However, Qatar succeeded in convincing the Darfuris that Qatar would play an impartial role in the mediation between them and the Government of Sudan (Kamrava, 2011). The religious ties also contributed to the ability of Qatar's Minister of State for Foreign Affairs to talk to the Darfuris and create an environment of trust (interview with Al-Hamadi, 2019).

The mediation strategy

Qatar employed a spectrum of mediation strategies in the mediation in the Darfur conflict. Those strategies range from facilitation, formulation, and manipulation.

The facilitation strategy of Qatar is represented in offering privileges, hotels and aircraft to the disputants (Barakat, 2014), and to offer a safe environment for meetings.

AL-Hamadi, the ex- Qatari ambassador to Khartoum, assures that Qatar played the role of facilitator in its mediation in Darfur: "..and to facilitate the negotiation process between these parties" (interview with Al-Hamadi, 2019).

Qatar also used the manipulation strategy, which is represented in introducing financial investment and incentives to the disputants. The financial resources are used in different stages of the mediation. Parallel to the Qatari mediation in Darfur in June 2010, the Qatar Charity announced a project of USD 1 billion to improve the livelihoods of 65,000 impoverished people living within the affected area (Dabanga, 2010).

Qatar also used incentives at the end of negotiations to encourage the disputants to reach an agreement. Qatar introduced incentives, whatever the form of those incentives, whether investments, rehabilitation infrastructure or even development aids. In the end, Qatar offered incentives for the Government of Sudan and Darfur. According to the international mediation scholarship, when the mediator offers incentives to the disputants to reach an agreement, the mediator is acting as a manipulator. This is the case of Qatar in Darfur, acting as a manipulator. The mediator may use either carrot (incentives) or a stick (pressure) to manipulate the disputants. It is not easy for a low-leverage state like Qatar to pressure the disputants as a powerful mediator would.

However, Qatar behaves differently. Qatar utilised its wealth and financial power in Darfur negotiations to encourage the disputants to sign an agreement (Roberts, 2015). Moran suggests that Qatari mediation is more advanced than power-based mediator because Qatar is based its mediation on its financial resources as incentives for the disputants to come to an agreement (Moran, 2009). Qatar promised to invest a further USD 2 billion to address chronic underdevelopment in the Darfur region provided that talks were successful and led to the closure of the conflict (Sudan Tribune, 2011a).

Qatar was eager to be a mediator in the Darfur conflict and willing to spend enormous resources to reach this goal. Qataris were under pressure to broker a deal because Egypt waited to step in as a mediator should the Doha talks fail (The Enough Project, 2010). The Arab league named Qatar a mediator in the Darfur conflict, which provided Qatar with a regional mandate for involvement in the mediation of the Darfuri conflict (Kamrava, 2015). However, this is considered a challenge to Egypt, which considers Sudan to be its backyard (Sudan Tribune, 2011). Al-Hamadi (2019) illustrates that Qatar considers mediation a necessary strategy that would stimulate a business environment that encourages investment, raise the standard of living in these countries, and halt the bloodshed (interview with Al-Hamadi, 2019).

To justify the Qatari usage of financial resources in the Darfur conflict, Taguia, the head of Aljazeera Research Center, suggests that Qatar introduced financial resources as a part of its comprehensive vision of mediation. He explains that Qatar succeeded in developing its own model in mediation, a model which is built on peacebuilding and development (interview with Taguia, 2019). He adds that the Qatari investment in Sudan is not a manipulation tool. He illustrates that those

investments are part of the mediation process to achieve sustainable peace according to the comprehensive Qatari vision of mediation. He asserts that "by mediation, Qatar offers the people new alternatives to transfer them from the war economy to the peace economy and this is important to transfer them from the context of war to the context of peace" (interview with Taguia, 2019).

Taguia rejects using the word "manipulation" when describing Qatar mediation strategy. He suggests that Qatar does not buy positions or the parties in its mediation. He adds that to reach sustainable peace, there should be a rehabilitation of the infrastructure, a recovery of the economy, and creating jobs. He says, "People in the conflict zones should realise the difference between being in peace and war. They should know the value of living in peace by enjoying a better economy and better hospitals and schools." He explains that people should believe that their life in peace is much better than their life under war, so they will not return to a state of war...this is the Qatari strategic vision for mediation, Taguia says (interview with Taguia, 2019).

In the same vein, Lulwa Al- Khater, the Qatari Ministry of Foreign Affairs spokesperson, suggests that Qatar does not have a specific agenda to force the disputants to follow through with its mediation. She explains that Qatar does not manipulate the disputants to reach an agreement. Qatar behaves as a facilitator and does not intervene in the material of the mediation (interview with Al-Khater, 2019).

In the same manner, In the same vein, Amin Hassan Omar, the main negotiator who represents the government of Sudan, asserts that Qatar never exerted pressure on any party (interview with Hassan Omar, 2019).

Sisi puts forth that Qatari money was not used as a manipulation tool to push the disputants to reach an agreement. Most importantly, the Qatari pledge came at the end of these negotiations. The money for the reconstruction and development of Darfur was a genuine effort to help speedy recovery in Darfur (interview with Sisi, 2019). Additionally, he asserts that the investment in Darfur was an expected peace dividend. The root cause of the conflict is the lack of development and investment (interview with Sisi, 2019).

In addition to the facilitation strategy and the manipulation strategy, Qatar used a formulation strategy in the mediation process in Darfur's conflict. Al-Khater (2019) demonstrates that Qatar may make some suggestions for both disputants and leave them to decide. Qatar never manipulated any party (interview with Al-Khater, 2019). Similarly, Hamadi explains that Qatar uses soft diplomatic means based on the principle of persuasion through dialogue and argument, bringing points of view closer together, and removing all barriers to an agreement between the parties concerned (interview with Al-Hamadi, 2019). He adds that Qatar did not interfere in the mediation substantively, except at the request of the parties concerned, without interfering in the internal affairs of states, to bring views closer together to find sustainable solutions to disputes and differences (interview with Al-Hamadi, 2019).

In sum, Qatar used a variety of mediation strategies in the mediation Darfur conflict, facilitation, formulation, and manipulation. Qatar played the role of facilitator when it offered a safe environment for the disputants to meet. Qatar played the role of formulator when it suggested proposals and solutions to the disputants' problems during the mediation process. Finally, Qatar used a manipulation strategy, especially at the end of the negotiations, to encourage the disputants to sign an agreement. By using

incentives, Qatar does have a carrot. However, Qatar does not stick to pressure the disputants to reach an agreement.

Impartiality/ partiality

The Qatari case shows how a mediator can be impartial (i.e., not favouring a particular party) whilst using manipulative strategies to push for a solution. It is an exceptional case. In the literature, there is the assumption that impartial mediators tend to be low stake, low-leverage and hence will refrain from using manipulation. One of the Qatari strengths in mediation is its impartiality.

In 2008, Qatar was a new player in international mediation. Qatar was busy introducing its approach, which is based on impartiality. In a conflict like Darfur, where the Darfuris suffered from the bias of Arabs, the Darfuris needed an impartial mediator whom they could trust. In mediating the Darfuri conflict, Qatar stood at the same distance from the two parties, the government of Sudan and the Darfuri rebels. Qatar dealt with both parties the same way. Qatar was biased neither to the government of Sudan nor the Darfuri rebels. Qatar introduced financial help to both parties. The Qatari promise to be impartial is the main reason the Darfuris accepted Qatar as a mediator. However, Qatari impartiality is a debatable issue. Some belief in the impartiality of Qatar (Barakat, 2012; Ginty 2008; Kamrava; Mitchell, 2008; Nuruzzaman, 2011). Others think that Qatar was not an impartial mediator as they claim that Qatar was biased toward the Government of Sudan (Jibril, 2010).

Nuruzzaman (2015) asserts that Qatar, in its mediation in the Darfur conflict, cannot be anything except being an impartial mediator, especially in the eyes of the Darfuris. He adds that the Darfuris would never accept the mediation if they were not sure

Qatar would be an impartial mediator. The Darfuris would not accept Qatar because it is an Arab country. He explains that the Darfuris distrusted Arabs after their mediation experiences with Egypt and the Arab league. The general Darforian consensus was that Arabs were biased towards the Sudanese government in Khartoum (Nuruzzaman, 2015.). Nuruzzaman (2015) adds that a distinctive advantage of the Qatari foreign policy is that the Qataris talk to all parties. Qatar's involvement in the Sudan conflict began with the preliminary efforts of Qatar's Minister of State for Foreign Affairs, Ahmad bin 'Abdullah al-Mahmud, in 2008. Al-Mahmud based the strategy on "get to know and act." Nuruzzaman confirms that this strategy is based on a deep knowledge of the conflict, gaining the confidence of the disputants, and gaining the support of stakeholders to push forward a peaceful solution (Nuruzzaman, 2015). Building trust among the Darfuris and the Sudanese government that Qatar would be an honest peace mediator was necessary for Qatar to be accepted as a trustful mediator. The Darfuri representatives accepted Qatari mediation because the Qatari mediators offered a new and different perspective on the Sudanese conflict. Kamrava (2011) confirms that Qatar's Minister of State for Foreign Affairs worked hard to change the Darfuris' perspective and gain their confidence; he succeeded in generating a sense of goodwill towards Qatar by assuring them that Qatar would be an impartial mediator whom they could trust. In order to show the Darfuris that Qatar would be an impartial mediator and would not be biased to Khartoum, Qatar took many steps. Qatar invited both disputants to negotiations in Doha, and before this step, the country's Minister of State for Foreign Affairs himself went to Darfur to witness the reality of the situation there (Kamrava, 2011).

However, some claim that Qatar was biased toward the government of Sudan. They argue that Qatar pushed forward with negotiations to save the Sudanese president,

AL Bashir, from the international criminal court (ICC) (Jibril, 2010). In answering the claim that Qatar was biased to the Sudanese Government in Khartoum and that Qatari mediation came to save President Al-Bashir from the International Justice Court, Omar Hasan Amin, the main negotiator who represented the Sudanese Government, asserts that Qatar was not the primary mediator and the leading mediator was the Joint Representative of the UN. There were no differences between the two mediators. He confirms that Qatar was an impartial mediator, and there is no way for Qatar to be anything except impartial (interview with Hassan Omar, 2019).

The head of the Darfuri team, Tigani Sisi himself, confirms that Qatar was an impartial mediator: "The state of Qatar's approach to mediating between the Government of Sudan and the Liberation and Justice Movement was characterized by impartiality and a keenness to be of the same distance from both parties to the negotiations" (Sisi, 2019). He adds that both parties had equal access to the joint mediation facilities and experienced the same treatment regarding the time allocated for consultations. He confirms that he has not seen any bias from Qatar towards the Sudanese Government (interview with Sisi, 2019).

He explains that Qatar looked at the two parties as Muslims who have the same culture. There is no space to be biased toward any of the two parties in such a situation. Qatar built on the cultural heritage to bridge the gap between the two parties and to bring them closer (interview with Sisi, 2019).

Ali Al-Hamadi agrees with Sisi by confirming that Qatar is an impartial mediator. He explains that Qatar invests in its relationships to create trust. He explains that any mediation process depends on gaining confidence between the parties concerned,

and this trust is built only by a network of institutional and personal relationships. Also, it is reliant upon adopting positive attitudes from all the parties involved and trying to bridge the gaps and deal with the prevailing local culture, so that this relationship is based on a cultural heritage that works to bring the parties closer (interview with Al-Hamadi, 2019). He says about the Qatari coordination with the parties, "Coordination with the various parties to the crisis is coordinated with impartiality, balance and without prejudice to one party without one party gaining the trust of all parties on an equal footing, and the absence of interests of Qatar to prevail over another" (interview with Al-Hamadi, 2019). Similarly, Lolwa Al-Khater (2019) affirms that Qatar always stands at the same distance from all parties. In other words, Qatar is impartial and is not biased toward any party (interview with Al-Khater, 2019).

Amin Hassan Omar, the representative of the Sudanese government in the negotiations, asserts that the negotiations in the Qatar process were a combination of consultations with the stakeholders who were not part of the conflict like civil societies represented the Internal Displace persons (IDPs) and the refugee's civil societies. Additionally, the negotiations between the parties based on the outcomes of the consultations the stakeholders could impact the negotiations. He confirms that This was a guarantee against any bias on the part of the mediation team. Qatari impartiality shared with the Joint Representative is the main factor that made Qatar a successful mediator in the Darfur conflict (interview with Hassan Omar, 2019).

In sum, there is no evidence to prove that Qatar was not an impartial mediator. Qatar impartially deals with both parties, the Government of Sudan and the representative of Darfur.

CHAPTER SEVEN: CASE STUDY FOUR: FATAH-HAMAS DOHA AGREEMENT
(2012)

Introduction

The conflict between the Palestinian National Liberation Movement (Fatah) and the Islamic Resistance Movement (Hamas) is often thought to have begun in 2007, is the most prominent and intense stage of the struggle between the two sides. However, the roots of the conflict go back more than a quarter-century (Tuastad, 2013). The conflict between the two prevailed because of the differences in the ideologies adopted by the two movements. Hamas called for Islamic thought, and the military resistance of Israel, and Fatah adopted a secular ideology and maintained those negotiations with Israel as the way forward to end the Israeli occupation. The differences were exacerbated in 1993 when the PLO signed the Oslo Accords, something Hamas strongly opposed. After the establishment of the Palestinian Authority, according to the Oslo agreement, a dispute broke out between Fatah and Hamas, where the latter refused to discontinue its military actions against Israel, which Fatah movement saw as a serious direct threat against its political project (Tamimi, 2009; Tuastad, 2013).

Meanwhile, the Palestinian Authority's security services launched large-scale arrests against Hamas leaders, operatives, and activists between 1996 and 2000 (Tamimi, 2009). Then, in 2006, Hamas decided to enter the Palestinian parliamentary election, where it managed to win the majority of the Palestinian parliamentary seats. Fatah refused to hand over the reins of power to Hamas, while Hamas insisted on its right to rule over the Palestinian authority. This resulted in a sharp escalation in the conflict between the two movements. In June 2007, the tension reached its breaking

point, and fights broke out between Fatah and Hamas. The fighting ultimately resulted in the defeat of Fatah, and power in the Gaza Strip was taken over by Hamas (Zweiri, 2006). After 2006 there were several attempts at mediation for reconciliation between Fatah and Hamas. Examples of those attempts are February 2007 Mecca Agreement; March 2008 Sana'a Declaration; May 2011 Cairo agreement; and the February 2012 Doha agreement (Aljazeera, 2012).

Qatar mediated the latter agreement. Qatar is one of the countries that has tried to reconcile Fatah and Hamas, and not just once, Qatar also has tried to mediate in 2006 between the two movements. The Fatah-Hamas Doha Agreement (2012), the Doha Palestinian Unity Agreement- is a case study for Qatari mediation and will be discussed in this chapter. The main argument in this chapter tis hat Qatar, as an interest-based mediator, used a range of different mediation strategies, facilitation, formulation, and manipulation in the mediation process. Additionally, Qatar acted as an impartial mediator in its mediation between Fatah and Hamas.

To reveal the essence of the Qatari mediation approach in the mediation between Fatah and Hamas, this chapter will first offer a brief background on the conflict between Fatah and Hamas, then expound on the role of Qatar in the conflict, and finally address the Qatari approach that was employed in the mediation process. In analysing Qatar's mediation, special attention will be paid to the Qatari motivation, the mediation strategy that Qatar adopts in the mediation, the Qatari impartiality and finally, whether Qatar is a pure mediator or power mediator.

Fatah and Hamas conflict

The dissonance between Hamas (the Islamic Resistance Movement) and Fatah (the National Liberation Movement) materialised in a full-on conflict in 2006 with Hamas's entrance into the political arena. It had won the Palestinian parliamentary election in January 2006, where it succeeded in securing 74 of the 132 seats in the Palestinian parliament, granting it a clear majority (Aljazzera Staff write, 2017; Klein, 2007; Kurz, Dekel, & Berti, 2018; Mishal & Sela, 2006). However, the roots of the conflict between the two parties went back to when Hamas was founded in 1987 as an armed branch of the Muslim Brotherhood in the Gaza Strip and the West Bank (Tuastad, 2013). From its inception, Hamas never recognised the Palestinian Liberation Organization (PLO) as the sole legitimate representative of the Palestinians (Tamimi, 2009). Moreover, Hamas, in its founding charter, considered PLO as a secular organization and its values incompatible with the values of Hamas, holding PLO in contempt for not bearing the Islamic nature of Palestine (Mishal & Sela, 2006). Hamas shared a common cause with the other resistance groups in Palestine, including Fatah, in its struggle against Israel (Milton-Edwards & Farrell, 2010; Tuastad, 2013). By signing the Oslo agreement, many members of Hamas considered Fatah as traitors. This sentiment sparked a conflict between Hamas in exile, represented by Khaled Mishaal, and Hamas in the West Bank and the Gaza Strip represented by Ahmad Yasin. Ahmad Yasin refused to label Fatah as traitors and still views Yasir Arafat as a brother.

On the one hand, Ahmad Yasin was afraid of the prospect of a civil war breaking out between Fatah and Hamas in Gaza, which is what was likely to happen if Hamas accused Fatah of betrayal (Chehab, 2007). On the other hand, Yasir Arafat was afraid that the popularity of Hamas in the streets of Palestinian would encourage people to revolt against the Oslo accord. He contacted the Muslim Brotherhood in Egypt to prevent such an outcome, asking them to convince Hamas to be represented in the PLO and have a specific quota at the Palestine National Council (PNC). Hamas suggested new elections for the PNC (Tamimi, 2009), believing it could get 40% of the PNC seats. Of course, Arafat did not agree to this (Tuastad, 2013). Hamas failed in securing seats in the PLO and PNC; it boycotted the Palestinian legislative council (PLC) election in 1996. However, Hamas decided to participate in the 2006 PLC election (Tamimi, 2009; Tuastad, 2013).

Surprisingly, Hamas won a majority in the election and so began its turn to run the official government of the Palestinian Authority (Kassem, 2012; Shamir & Hecht, 2014). After winning the PLC elections, Hamas initially demanded that Fatah create a national unity government. However, Hamas's proposal was rejected by Fatah, which created rising tensions between the two movements (Zweiri, 2006). Moreover, Fatah refused to hand over the reins of power to Hamas while Hamas insisted on its right to control the Palestinian authority, and thus the conflict between the two movements escalated. Unable to bring Fatah on board, Hamas's elected representatives proceeded to form a cabinet consisting only of Hamas-elected representatives and appointed Ismail Haniyeh as prime minister in March 2006 (Kurz et al., 2018; Morro, 2007).

In June 2007, after several failed attempts to reconcile the two movements, hostilities erupted in the Gaza Strip, and the political rivalry deteriorated into violent conflict.

Hamas launched a military offensive (Schanzer, 2008). The civil war lasted four days,

at the end of which Hamas had effectively taken control of Gaza, including the Palestinian Authority (PA) government buildings (Kurz et al., 2018; Shamir & Hecht, 2014) and forced Fatah into exile. (Brown, 2010; Schanzer, 2008). The Palestinian president, Abbas, was sure that he had lost Gaza, so he dismissed the unity government that the Saudis had helped to create. The conquering of Gaza by Hamas was an event that people expected due to the long-lasting animosity between Fatah and Hamas (Schanzer, 2008).

While Hamas assured that this war had been a defensive one in which Hamas had only defended itself, Hamas maintained that it was forced to enter this war to protect itself from Fatah, who was accused of being a collaborator with Israel and the USA (Erlanger, 2007). Amnesty International and the Palestine Centre for Human Rights blamed both sides for the violence and asked Fatah and Hamas to cease hostilities and protect civilians (Amnsety International, 2007; Palestine Center for Human Rights, 2007).

Since that time, Hamas and Gaza Strip have undergone diplomatic and economic sanctions from Israel, the USA, and the European Union (EU), with strict movement limitations for people and goods. Meanwhile, the PA in West Bank received economic and military support from the EU and USA (World Bank, 2011). Since that time, the Palestinian Authority has been split into two fractions: Gaza Strip under Hamas Authority and West Bank under the authority of Fatah (Shamir & Hecht, 2014). Since 2006 Fatah and Hamas have reached a series of agreements aimed at achieving Palestinian reconciliation and ending the internal division, but these agreements—which were sponsored by Arab states— have not been implemented on

account of various obstacles. The most prominent of these agreements are as follows: the Mecca Agreement 2007, the Cairo Agreement 2011, the Doha Agreement 2012 and the Shati agreement 2014 (Aljazzera Staff, 2017; Yaari & Zilber, 2014).

In 2007, Saudi Arabia mediated the two movements in negotiations culminating in the Mecca Agreement. This agreement was the basis of understandings that resulted in a unified government. Fatah and Hamas signed a reconciliation agreement on 8 February 2007 in Mecca under the patronage of the then King of Saudi Arabia, Abdullah bin Abdelaziz Al Saud, to stop the internal fighting in the Gaza Strip and form a government of national unity. He participated in the deliberations prior to the Mecca Agreement, meeting with Palestinian President Mahmoud Abbas and Parliamentary member Mohammed Dahlan from the Fatah movement, and with Prime Minister Ismail Haniyeh and the chairman of the Political Bureau of the Hamas movement, and Khaled Mishaal from Hamas (Aljazzera Staff write, 2017; Morro, 2007). However, the agreement was short-lived, and tensions quickly escalated between the two movements. Fatah's refusal to transfer control of the PA's security forces to the interior ministry headed by Hamas was a denial of power-sharing (Aljazzera Staff write, 2017; Yaari & Zilber, 2014).

The Palestinian factions later met in Cairo on December 20, 2011, under Egyptian auspices, to discuss the mechanisms for implementing the Palestinian National Accord agreement, signed by the factions in Cairo on May 4, 2011. The Palestinian national dialogue in Cairo focused on addressing all the issues that resulted from the Palestinian division through an inclusive dialogue involving all factions, organizations and independent forces, where key committees were formed to fulfil the imperatives of reconciliation: elections, community reconciliation, The formation of a Government

of national unity, public liberties and confidence-building, the operationalization of the Legislative Council and the restructuring of the PLO to intervene with the non-represented factions, in particular, Hamas and the Islamic Jihad which were accumulated in what is known as the Cairo Agreement 2011(Johannsen et al., 2011; Yaari & Zilber, 2014).

After the failure of the Cairo Agreement 2011, Qatar intervened in the mediation between Hamas and Fatah which ended with the signing of the Doha Agreement in 2012. Both Fatah and Hamas signed this reconciliation agreement in Doha, Qatar on February 6, 2012, with Palestinian President Abbas—on behalf of Fatah and Khaled Mishaal—on behalf of the Hamas movement—to accelerate Palestinian national reconciliation. The agreement —which was attended and sponsored by the then Emir of Qatar, Sheikh Hamad bin Khalifa Al-Thani—provided for the operationalization and development of the PLO, through the reconstitution of the Palestinian National Council in parallel with the presidential and legislative elections, and the formation of a national consensus government of independent professional competencies headed by Abbas, whose mission would be to facilitate the presidential and legislative elections and initiate the reconstruction of Gaza, and the continuation of the work of the committees formed after the Cairo Agreement (Aljazzera Staff write, 2017; Milton-Edwards, 2013; Yaari & Zilber, 2014).

Further to the 2012 Doha Agreement, the Shati agreement was signed between the two movements in April 2014. The Shati agreement is considered one of the most crucial reconciliation agreements between Fatah and Hamas. The dialogue sessions were held at the home of Ismail Haniyeh in the Shati refugee camp west of Gaza City, from where its name is derived. Following the agreement, a Palestinian

consensus government was formed, supposed to be followed by six months of elections. However, that did not happen. The parties then reaffirmed their commitment to the agreements reached in "Cairo 2011" and the "Doha Agreement of 2012" and their reference to the implementation of national reconciliation. Ismail Haniyeh –the leader of Hamas- announced from his home in the Shati refugee camp that the schism between Hamas and Fatah was over.

However, in reality, the agreement did not end the schism between the two groups (Aljazzera, 2017; Yaari & Zilber, 2014). Ehud Yaris and Neri Zilber comment on the announcement of Ismail Haniyeh that this is not the first time that Ismail Haniyeh has announced an end to the schism. They say that he has announced it several times in the last ten years. However, there is no implementation of any agreement. Additionally, many significant issues were not settled or even addressed in the agreement, such as the head of the unified government, whether it should be Mahmoud Abbas, whether it would include members of Hamas and other similar issues (Yaari & Zilber, 2014).

Regarding the subject of this research which discusses Qatari mediation, the thesis will focus on the Doha agreement between Hamas and Fatah (2012), which will be discussed in detail in the next section.

Qatar role in the mediation Fatah- Hamas.

The relationship between Qatar and the Palestinians was fostered in two ways: through its humanitarian aid and the Aljazeera satellite channel. The Qatari humanitarian aid to Palestine was deployed via the Qatari charity fully registered in the Gaza Strip and West bank in 1997. Meanwhile, Aljazeera had given favourable coverage to

the Palestinian case since its inception in 1996 (Hansen, 2013). October 2006 was the first instance of Qatari intervention in Palestinian affairs as a mediator between the feuding Palestinian factions, Hamas and Fatah, aiming to resolve their conflict and form a united government (Myre, 2006). Qatar's intervention was in the wake of the failed Egyptian mediation trial (Hansen, 2013). Qatari mediation between the feuding Palestinian parties was conducted by Sheik Hamad bin Jassim al-Thani, who shuttled between the leaders of the two movements, Mahmoud Abbas and Ismail Haniyeh. However, the mediation yielded no progress as the core obstacle was the recognition of Israel, which Fatah accepted and Hamas did not (Hansen, 2013; Myre, 2006).

In 2008, following the Israeli attack against Gaza in December, the relationship between Hamas and Qatar was strengthened. Yousef Alqaradawi even went on a tour throughout the Middle East to support Hamas (Hansen, 2013). The year 2011 saw the outbreak of the Arab Spring, and witnessed many changes such as the collapse of the Mubarak regime and the evacuation of Hamas from Syria, in addition to the rising strength of the Egyptian brotherhood represented by Yousef Alqaradawi in Qatar (Hansen, 2013). Those changes created a new environment and opened a new window for Qatar to mediate again between Fatah and Hamas (Hansen, 2013).

In late 2011 and early 2012, Qatar pushed ahead with one of its most ambitious mediation projects: facilitating unity negotiations between the rival Palestinian factions Fatah and Hamas (Antwi-Boateng, 2014; Barakat, 2012). The first meeting between Khaled Mishaal and Mahmoud Abbas took place in Qatar (Fromherz, 2017).

On 7 February 2012, Fatah and Hamas signed The Fatah-Hamas Doha agreement. Hamas was represented by Khaled Mishaal, while Mahmoud Abbas represented Fatah. The agreement was reached under Qatar's sponsorship in the presence of Qatar's Emir Sheikh Hamad (Ma'an staff, 2012; Sawafta, 2012). According to this agreement, Mahmoud Abbas would head a unity caretaker government and would be at the same time the president and prime minister of the future technocrat unity government (Barakat, 2012; Issacharoff & Khoury, 2012; Sawafta, 2012).

The Israeli Prime Minister, Benjamin Netanyahu, condemned the agreement, ensuring the impossibility of making peace with Hamas. Moreover, he gave a choice to the Palestinian Authority in West Bank, asking them to choose between peace with Israel or peace with Hamas (Barakat, 2012; Ravid, 2012). A few days later, Abbas replied, reassuring Benjamin Netanyahu that the new interim cabinet would continue its commitment to the obligations and agreements signed by the Palestine Liberation Organisation (Maan staff writer, 2012). With this announcement, Abbas drew anger from Hamas, which did not recognise the existence of Israel. Despite the blockade, Doha agreement 2012 was followed by the Emir's visit to Gaza, where he promised aid and projects to the stricken Gazans (Rabbani, 2012). In the case of the conflict between Hamas and Fatah, Israel is considered a primary stakeholder, and this is due to the impossibility of excluding Israel from any solution or agreement between Fatah and Hamas. Any mediation trial results would affect Israel's security interest (Yaari & Zilber, 2014). Israel would not accept a Palestinian unity government that included Hamas's members and never count Hamas as a peace partner. This was the sentiment of Benjamin Netanyahu against the Doha agreement when he condemned the

agreement, assuring the impossibility of making peace with Hamas. Moreover (Barakat, 2012; Ravid, 2012).

Approval of Israel for any future Palestinian government is critical and could explain the reason behind the failure of so many trials at mediation between Hamas and Fatah. There are no official diplomatic relations between Qatar and Israel. However, there is contact between Israeli officials and their Qatari counterparts on a smaller scale. The Israeli position towards Qatar is a double-sided one. On the one hand, Israel did not welcome the mediation of Qatar in the Palestinian conflict. The Israeli Knesset member, Avigdor Liberman, opposed Qatar being a mediator to the conflict and opposed its participation in Gaza reconstruction, accusing Qatar of hosting Hamas's command. On the other hand, Israel needed Qatar to support the humanitarian projects in Gaza to ease the economic pressure and postpone the confrontation with Gaza (Kurz et al., 2018).

The Qatari International mediation approach in mediation between Hamas and Fatah

This section focuses on three components; the Qatari international mediation represented in Qatari motivation, mediations strategies, and impartiality.

The Qatari motivation

The Qatari interest ranges from Qatar's security to the branding of Qatar as a peace broker. This is applied to all mediation trials that Qatar has conducted without exception. The establishment of the Qatari mediation is based on the security of Qatar (interview with Taguia, 2019). However, the Qatari interest in mediation is a debatable

issue. Some scholars, such as Rabi (2009), argues that Qatar changes its foreign policy based on its national interest and changes in the geopolitical realities (Rabi, 2009). Khatib (2013) argues that Qatar involved itself in mediation for security reasons and protection (Khatib, 2013a). In the case of the Qatari mediation between Fatah and Hamas, Rabbani (2012) demonstrates that, in the case of the mediation in Palestine, Qatari interest lies in branding itself as an honest peace broker and an alternative to Egypt (Rabbani, 2012). Some scholars further assert that Qatar has no direct or specific interest in the Gaza Strip or the West Bank as they have nothing to offer to Qatar (interview withTaguia, 2019). Despite this debate about the Qatari interest in mediation between Fatah and Hamas, some evidence shows those suggestions.

For the argument that Qatar had no interest in the mediation between Fatah and Hamas, Katzman (2016) confirms that the Qatari fund is only for humanitarian and civilian projects and benefits the residents of the Gaza Strip (Katzman, 2016).

For his part, the chairman of Qatar's Gaza Reconstruction Committee Ambassador, Mohammed Al-Emadi, demonstrates that Qatar deals with Gaza as a humanitarian case due to its exceptional circumstances because two million people in the Gaza Strip live in unhumanitarian circumstances. He says, "in helping Gaza Strip, we care about justice and protecting humanity not more…we are not looking forward political gains" (interview with Emadi, 2019).

Reconciliation between Fatah and Hamas is an asset added to the Qatari branding as a peace broker. Rabbani (2012) argues that Qatar involvement in the Fatah-Hamas conflict was due to branding itself as a peace mediator, which contributed to

granting Qatar a higher status in the world and portraying Qatar as a peacemaker (Rabbani, 2012). Branding Qatar is an intangible interest. However, Mosa Abu-Marzook, a Hamas representative, argues that Qatar has no tangible interest in the mediation between Fatah and Hamas. He confirms that Qatar did not get any tangible benefit from its mediation between Fatah and Hamas. He demonstrates that Qatar is not seeking to invest in Gaza Strip or West Bank. He explains that Qatar seeks to resolve international conflict peacefully and civilly, as Qatar has the skills and the capacity to do so, he assures (interview with Abu-Marzook, 2019).

In the same manner, Hossam Badran, a representative of Hamas, suggests that: Qatar decided to be a mediator between Hamas and Fatah on a humanitarian basis. He asserts that Qatar is honest and serious about helping and supporting the Palestinian people. He adds that the Qataris believe that Palestinian reconciliation is for the benefit of all Palestinians, and there would be no progress on the Palestinian level without the reconciliation. He suggests that the Qataris believe that reconciliation is the first step to getting on the right track for the Palestinians. Consequently, he identifies that Qatar has neither self-interest nor a private agenda in its mediation between Fatah and Hamas. He says, "they do not have a specific agenda or any specific interest in the mediation between Fatah and Hamas... Qatar does not need anything from either Hamas or Fatah...There is no financial interest for the Qataris in Palestine "(interview with Badran, 2019). To confirm his point of view, he says that "on many occasions, Qatar told us more than once: If you want to hold reconciliation meetings in Doha and announce them in press conferences in Cairo, Qatar has no problem, which confirms what the Qataris care about is reconciliation", according to Badran

(interview with Badran, 2019). Additionally, Badran argues that Qatar is motivated by its strong desire to end the division in Palestine.

Suggesting that Qatar involved itself in mediation for security reasons and protection (Khatib, 2013a), Abu-Marzook refutes this suggestion by assuring that what protects Qatar is its unique and exceptional relations with international powers such as the USA and regional powers such as Turkey and Iran. He explains that the good Qatari relations with those powers protect Qatar rather than mediation and conflict resolution (interview with Abu-Marzook, 2019). In contrast, Fatah's representative, Muneer Salama argues that Qatar has a lot to gain from emerging as a champion of Palestinian unity in the region. He claims that Qatar has an interest in the mediation between Fatah and Hamas. He demonstrates that Qatar wants to play a central political role in the region. He says, "whoever has the most money can influence decision-making, and Doha wants to play any role that gives it space to become an important hub in the Arab region." He says, "Qatar wants to be among the major political players in the region who influence the political map in the region." He identifies that Qatar has the money and owns the Muslim Brotherhood, so that's the only reason that it can play that role (interview with Salamah, 2019).

Related to Qatar's relationship with the Muslim Brothers, Walid Al-Awad, a political member of the Palestine People's Party, generalises that Qatar mediation between Fatah and Hamas aimed to support the Muslim brotherhood. He claims that Qatar initiated the mediation between Hamas and Fatah to strengthen the Muslim brotherhood, as Hamas is a part of the Muslim Brotherhood. He asserts that Qatar played the mediator role between Hamas and Fatah to instil the idea that the Egyptian

regime in Egypt is weak and unable to have a regional and international role (interview with Al-Awad, 2019).

Not to mention that the Egyptian regime was in a transition at the time of the Qatari mediation in February 2012. President Mubarak stopped being a president of Egypt in February 2011, a year before signing the Doha agreement between Fatah and Hamas. Refuting Al-Walid's argument, Khaled Mishaal, the ex-head of Hamas, suggests that Qatar had no problem with Egypt until the year 2017 when Egypt forced a siege on Qatar in cooperation with Saudi Arabia, the United Arab Emirates and Bahrain, Khaled Mishaal asserts (interview with Mishaal, 2019). Additionally, he identifies that Qatar usually does not mind help in holding the negotiations in Doha and declared it in Cairo without mentioning the Qatari role (interview with Mishaal, 2019). The debate on the Qatari interest in mediation between Hamas and Fatah ranges between denying and confirming the Qatari direct or specific interest in Gaza Strip and West Bank. However, this debate cannot deny that Qatar was seeking to play the role of regional peace broker by promoting peace between Fatah and Hamas. Qatar indeed looks at Fatah and Hamas as Palestinian factions that should be reconciled. However, Qatar has a lot to gain from emerging as a champion of Palestinian unity in

The Qatari leverage in the mediation between Fatah and Hamas

the region.

In the case of its mediation between Hamas and Fatah, Qatar has derived its leverage from three sources: its incentives which take the form of financial resources, the role of the Emir and his diplomacy and finally, the cultural and religious ties.

As for financial incentives, Qatar usually uses its enormous wealth as a tool in its mediation. Antwi-Boaten (2014) asserts that Qatar flexes its financial muscle through the mediation process by promising enormous sums for infrastructure development in return for a peace agreement (Antwi-Boateng, 2014). He confirms that Qatar is the primary source of financial assistance and development investment in the Gaza Strip. He explains that Qatar has provided Gaza with more than USD 500 million since 2012 (Milton-Edwards, 2013). Ishaq (2012) affirms that Qatar developed the opportunity to mediate between Hams and Fatah by using its financial leverage to aid the Gaza strip (Ishaq, 2012). Badran, the representative of Hamas, argues that the money that Qatar pledged (after the agreement) to Gaza was the money that Qatar promised to pay for Gaza rehabilitation. He adds that those sums of money were the contribution of Qatar into international donations to Gaza (interview with Badran 2019). He argues that the Qatari money pledged to Gaza Strip and the West Bank could not be classified as an investment. He explains that Gaza Strip and the West Bank are not the proper places and the safe environment for the Qatari investment. He says, "Qatar helped Gaza for humanitarian considerations" (interview with Badran, 2019). Al- Khater explains that those financial resources are introduced according to the Qatari comprehensive peacebuilding approach (interview with Al-Khater, 2019).

Qatar usually uses its financial resources to remove obstacles in the negotiations. For example, during the mediation process, the Gaza employees' salaries appeared as an obstacle to signing the agreement. At that moment, Qatar offered to pay the salaries of those employees, according to Khaled Mishaal (interview with Mishaal, 2019). While Salamah asserts that the Qatari money is considered leverage to Qatar

which is used in the mediation (interview Salamah, 2019), Badran confirms that "the money that Qatar paid to Gaza is considered a duty towards the Palestinian issue, that is what the Qataris believe", Badran says (interview with Badran, 2019). Khaled Mishaal asserts that the Qatari money is considered leverage for Qatari mediation; however, the Qatari money is not the fundamental factor. He explains that the Qatari money is not considered number one in the Qatari mediation. He assures many countries have money; however, they cannot be a successful mediator. He confirms that the eagerness of the Qataris to keep good relations with all parties is a crucial factor that helps Qatar to be a successful mediator. He says, "dealing with all parties the same way is considered leverage for Qatar and the Qatari experience with Gaza and West Bank. He adds that the smartness of the Qatari leadership in mediation is an essential factor, according to Khaled Mishaal (interview with Mishaal, 2019). Khaled Mishaal did not elaborate on" the smartness of the Qatari leaders. However, he believes that Qatari leaders' involvement in the mediation process is fundamental in reaching an agreement.

The second source of the Qatari leverage is Qatari personal diplomacy. In the case of the mediation between Fatah and Hamas, the mediation efforts and responsibilities were divided between the Emir and Sheik Hamad bin Jassim al-Thani who are considered the decision-making structure (Hansen, 2013). The Qatari Emir and Sheik Hamad bin Jassim al-Thani succeeded in using their capacity to bring Hamas and Fatah to the negotiation table and convincing them to sign the Doha agreement. Qatari personal diplomacy represented by Emir Hamad in dealing with Fatah-Hamas's reconciliation is a double-edged sword; on the one hand, the Emir succeeded in influencing the disputants to sign an agreement (Hansen, 2013). On the other

hand, the dependency on personal diplomacy was worrying to Hamas, according to Milton-Edwards & Farrell. Milton-Edwards & Farrell (2010) assert that Hamas was afraid that the power shift from the Emir Hamad to his son Tamim could alter Qatari support. For this reason, Hamas has located its external leadership in diverse locations throughout the Middle East, they explain (Milton-Edwards & Farrell, 2010).

Badran (2019) explains the importance of personal diplomacy in the Qatari mediation. He says, "the involvement of the Qatari Emir in the negotiations between Fatah and Hamas supported the negotiations and contributed to removing the obstacle in the negotiations...for example, one of the obstacles in the negotiations was "the head of the coming Palestinian government"...the close relation of the Emir with Khaled Mishaal on one side and his relationship with Mahmoud Abbas on the other side, removed this obstacle". Badran asserts that Emir suggested Mahmoud Abbas be the head of the coming government, and Khaled Mishaal agreed without reference to Hamas leaders in Gaza Strip who were not satisfied initially. However, Khaled Mishaal succeeded in convincing them of the necessity of this step to end the division, according to Badran (interview with Badran, 2019). Mishaal convincing Hamas to accept Abu Mazen as a leader is based on trust, not Qatari finance. Mishaal was confident that Hamas leaders trust. Qatari finance is not the reason because Hamas is a movement that no one can buy its positions, according to Mishaal (2019).

The third source of the Qatari leverage is the cultural and religious ties. The relationship between the Qatari elite with Mohamoud Abbas and on one side and the relationship between Hamas leaders and the Qatari elite from the other side is also considered leverage in Qatar's mediation between Fatah and Hamas. Milton-Edwards

(2013) claims that in the mediation between Fatah and Hamas, Hamas Accepted Qatar as a mediator due to the intervention of the Egyptian Muslim Brotherhood in Egypt and Hamas's relationship with Yousef Alqaradawi, who is located in Qatar and considered almost as a spiritual leader for Hamas. Meanwhile, Fatah Accepted the Qatari mediation due to its financial needs (Milton-Edwards, 2013).

Badran and Mishaal (2019) confirm that Hamas did not need the intervention of the Egyptian Muslim Brotherhood to convince them of reconciliation with Fatah. Hamas is already convinced of the necessity of reconciliation. Hamas needs neither the Egyptian Muslim Brotherhood nor Yousef Alqaradawi to convince them to be reconciled with Fatah (Interview with Badran; Mishaal, 2019). Mishaal asserts that Hamas behaves according to its fixed principles (interview with Mishaal, 2019). He adds that Emir Hamad supports the Islamic and Arab issues, and since the mid- 1990s, the Emir decided to open the Qatari foreign policy to the region (interview with Khaled Mishaal, 2019).

Mediation Strategy/ies

According to the literature on international mediation, the three known mediation strategies are facilitation, formulation, and manipulation. The mediator is not involved in the mediation process in the facilitation strategy. The facilitator's role is to offer a safe and friendly environment for the disputants to meet and offer hotels and light tickets. Formulation strategy includes suggestion proposal and controlling the agenda of the meetings. Generally, facilitation and formulation strategies show little intervention of the mediator in the mediation process. However, the manipulation strategy is the most active and coercive. Through the manipulation mediation, the mediator can

exert different forms of pressure on the disputants, and the mediator can even issue new proposals and guide the disputants towards signing an agreement (Beardsley et al., 2006; Bercovitch, 2011b; Bercovitch et al.,1991; Foster, 2011; Kleiboer, 1996). Manipulative mediators may offer carrots (compensation) to disputants to incentivise peace (Carnevale, 1986; Zartman & Touval, 1996). Those carrots may include direct compensation, favourable economic policies to the disputants, or diplomatic concessions (Beardsley et al., 2006).

Qatar used various strategies in the mediation between Hamas and Fatah, a manipulation strategy, facilitation and formulation strategy. By introducing its financial resources as incentives to the disputants, it is evident that Qatar acted as a manipulator mediator in its mediation between Fatah and Hamas. However, there is a debate on the mediation strategy that Qatar employed in the mediation between Fatah and Hamas.

Qatar is a low-leverage state, so it is expected to play a role in mediation different from that of a power mediator. However, some suggest that Qatar in its mediation seems to emulate the role of a superpower by manipulating the disputants to reach an agreement, using its vast financial resources (Antwi-Boateng, 2014; Barakat, 2014; Hansen, 2010). Antwi-Boateng (2014) asserts that Qatar flexes its financial muscle through the mediation process by promising enormous sums for infrastructure development in return for a peace agreement (Antwi-Boateng, 2014). He exemplifies Qatar primary source of financial assistance and development investment in the Gaza Strip (Milton-Edwards, 2013). The Qatari financial aid to Gaza created an opportunity for the Qatari mediation between Hamas and Fatah (Ishaq, 2012). Milton-Edwards (2013), Ishaq (2012) and Antwi-Boateng (2014) argue that Qatar used a

manipulation strategy by employing the Qatari money to push Hamas and Fatah to reach an agreement.

A member of Fatah Movement and an officer in Mahmoud Abbas Office, Muneer Salamah, claims that Qatar was a manipulator in the mediation between Fatah and Hamas. He mentions that Qatar manipulated Hamas by offering it economic temptations and Qatar may manipulate Fatah by putting pressure on it, blackmail or even threat (interview with Salamah, 2019). When Salameh was asked to give examples of either the economic temptations for Hamas or of threat and blackmail toward Fatah, he mentions that "I have no real examples", Salamah says (interview with Salamah, 2019).

Al-Emadi, the Qatari ambassador to Gaza Strip, suggests that people in the Gaza Strip suffer considerably from war and have been under siege with destroyed houses and infrastructure. He explains that sending money to the Gaza Strip does not mean that Qatar is helping Hamas. It means that Qatar is helping the two and half million people in Gaza who suffer and who need help. He assures that the money does not go to Hamas; the money goes to the Gaza Reconstruction Committee, established by the Qataris and headed by Al-Emadi, the Qatari ambassador to Gaza. Al- Emadi confirms that this committee coordinates with the Palestinian Authority in Ramallah but not with Hamas. He demonstrates that Hamas does not intervene in the committee's business, and Hamas does not receive any money from Qatar. He adds that the Qatari money is used in building schools, hospitals and electricity stations in the Gaza Strip (interview with Al-Emadi, 2019). Al-emadi focuses on the receiver of the money. However, he did not ignore that Gaza received money from Qatar.

The Qatari Ministry of Foreign Affairs confirms sending money to Gaza. The Qatari Ministry of Foreign Affairs states that Qatar established the Qatari National Committee for the reconstruction of Gaza in 2012. Its role is to manage projects related to Gaza and under the supervision of the Qatari Ambassador Mohammad AlEmadi. During his visit to Gaza, the Qatari Emir promised USD 407 million for Gaza reconstruction. Gaza reconstruction projects include housing and sports facilities (Qatari Ministry of Foreign Affairs, 2012).

Lulwa Al-Khater, the spokesperson of the Qatari Foreign Ministry, argues that Qatar is not playing the role of a manipulator by using its vast financial resources. She shows that Qatar's strategy in mediation is based on the Qatar comprehensive vision of mediation and peace. Qatar believes that mediation is a tool to achieve peace. Al-Khater demonstrates that Qatar uses its vast financial resources to strengthen peace and humanitarian development in a different field, create jobs, and focus on infrastructure rehabilitation (interview with Al Khater, 2019). Al-Khater confirms the purpose of the money. Additionally, she confirms that Qatar uses vast financial resources in international mediation.

Indeed, Qatar does own the carrots to encourage the disputants to reach an agreement. Those carrots are represented in the Qatari economic resources, which Qatar introduced as incentives to Gaza and West Bank. Badran, a Hamas representative, does confirm that it is true that Qatar offers financial aid to Gaza. However, he justifies that Qatar offered those financial aids to Gaza based on the donor conference for reconstructing Gaza (interview with Badran, 2019).

Qatar helps Gaza financially before the mediation, during the mediation and even after the mediation between Fatah and Hamas. Qatar usually uses financial resources when the negotiations reach a deadlock during the mediation process. A clear example that Qatar used its economic resources as incentives in the mediation process is when a deadlock faced the disputants Hamas and Fatah because of the salaries for Gaza employees.

According to Mishaal, Qatar suggested paying the salaries for the employees in Gaza for six months. He explains that the Qataris suggested a solution for the obstacle and used their money to resolve the problems. (interview with Mishaal, 2019).

Some would argue that Qatar introduced those financial incentives based on its comprehensive model in mediation based on peacebuilding and development. They assure that Qatar's money sent to Gaza is for building schools, hospitals, roads, and the rehabilitation of the Gaza infrastructure (interview with Emadi, 2019; Taguia,2019). However, when Qatar introduced financial resources as incentives, Qatar acted as a manipulator in the mediation between Fatah and Hamas. Regardless of the reason behind offering financial incentives (carrots) to the parties, it is defined as manipulation. The other side of manipulation is practising pressure on the disputants. There is no evidence that Qatar practices any pressure (stick) on either Fatah or Hamas.

In the case of the mediation between Fatah and Hamas, the mediation efforts and responsibilities were divided between the Emir and Sheik Hamad bin Jassim al-Thani who are considered as the decision-making structure (Hansen, 2013). Badran (2019) assures that although the Qatari personnel were attended the negotiation meeting

between Fatah and Hamas, Fatah and Hamas team were meeting in Doha in a comfortable environment away from pressure-(interview with Badran, 2019). Mishaal confirms that there was no pressure from Qatar on any party. He says, "Actually the presence of the Emir personally is a message to express that Qatar is caring that the two disputants reach an agreement... Additionally, the presence of Emir Hamad was to help in solving problems that were considered an obstacle in the way of reaching an agreement", Mishaal says. The saying of Badran and Mishaal assure that Qatar used a formulation strategy during the mediation process. To confirm the formulation strategy and to prove that the Qataris proposed solution when it is required, Mishaal and Badran explain that Fatah and Hamas could not agree on the head of the government and the Emir Hamad realised the need for a suggestion to overcome that administrative issue (interview with Badran, 2019). Mishaal assures that Emir Hamad suggested that Abu Mazen be the head of the coming government. Both Hamas and Fatah freely accepted the suggestion, and it was the trademark of the Doha agreement (interview with Mishaal, 2019).

Fatah main negotiator, Azzam Al-Ahmad, affirms that Qatar used a formulation strategy during the mediation process. Al-Ahmad confirms that Qatar sometimes intervened and made suggestions to both sides individually. He says," Qatar made written suggestions to each party and asked for a written response" (interview with Al-Ahmad, 2019).

Badran assures that this does not mean that Qatar practised any pressure on either side to agree. He adds that the personal relationship between the Emir from one side and Khaled Meshaal and Abu Mazen from the other side contributed to the acceptance from both parties of the Emir's suggestion (interview with Badran, 2019).

Khaled Mishaal states that Qatar has a good relationship with Hamas and Fatah. Qatar grants both of them freedom in the negotiations, neither pressuring nor manipulating them (interview with Mishaal, 2019).

In brief, Khaled Mishaal and Hussam Badran (from Hamas) believe that Qatar never placed any pressure on Fatah or Hamas during the negotiations, and they confirm that the attendance of the Qatari personnel such as Emir or the Minister of Foreign Affairs is not considered a factor of pressure. They explain that the attendance of the Qatari personnel gives the negotiators the freedom to represent themselves. They confirm that the Qataris never used pressure as a tool in the mediation process, they say (interview with Badran, 2019; Mishaal, 2019).

Hussam Badran explains that Qatar is a small country with a small population and is not a great country like the USA, so Qatar cannot pressure Hamas or Fatah. Furthermore, Qatar is not a significant enough neighbour to Palestine to practise pressure, for example, or even control the borders. He adds that the Qataris do not own the tools to pressure Hamas or Fatah. He explains that Qatar does not have the stick to pressure the disputants (interview with Badran, 2019).

The saying of Badran confirms that Qatar coerces the two parties. It agrees with the saying of Azzam Al-Ahmad. Al-Ahmad denies the occurrence of pressure on both parties. He asserts that Qatar had no pressure from Qatar on either Hamas or Fatah during the mediation process (interview with Al-Ahmad, 2019).

In sum, Qatar used a spectrum of mediation strategies during the mediation process.

Those strategies are facilitation, formulation, and manipulation. Qatar used its financial resources before the mediation process, and this is Qatar's way to encourage the

disputants to accept it as a mediator. Qatar used a facilitation strategy to offer a safe environment for the disputants' meetings. When there is a deadlock, Qatar plays the role of a formulator by suggesting proposals and solutions resolve the deadlock. Qatar used its financial resources to encourage the disputants to sign an agreement. It happened when Qatar promised to pay the Gaza employees' salaries. This money is considered a financial incentive. Introducing financial incentives to the disputants to encourage them to reach an agreement is called manipulation. It proves that Qatar acted as a manipulator in its mediation between Hamas and Fatah. While Qatar acted as a manipulator in its mediation between Fatah and Hamas, Qatar did not give away its impartiality even when hosting the Hamas leader in Doha.

Impartiality/ partiality

Based on the above section, Qatar used a manipulation strategy in the mediation between Fatah and Hamas. The manipulator does not pay much attention to impartiality (Smith, 1994). The Qatari mediation can be an exceptional case. Qatar is an impartial mediator in the two case studies. In the two cases, Qatar mediated between Muslims and Arabs. Qatar preserved equal relations with all the parties. Qatar hosted the disputants in the same hotels and dealt with them equally. When Qatar introduces incentives to one party, it also incentives the other party.

While some scholars state that Qatar was an impartial mediator between Fatah and Hamas (Barakat, 2012), other scholars suggest that Qatar was biased in favour of Hamas (Hansen, 2013).

There is some evidence that Qatar was an impartial mediator between Fatah and Hamas. Impartiality is a crucial element in Qatari mediation, and the credibility of the Qatari mediation lies in its impartiality (interview with Mishaal, 2019; Taguia, 2019). Al-Ahmad confirms that he did not witness any Qatari temptation to be biased towards either Hamas or Fatah (interview with Al-Ahmad, 2019).

Qatar had good, same relations with both parties in its mediation between Fatah and Hamas. Badran says, "Khaled Meshaal is located in Qatar and other Hamas members as well, and Fatah members including Mahmoud Abbas have a distinct relationship with the Qataris (interview with Badran, 2019). He confirms that Qatar deals with both parties without being partial to them (interview with Badran, 2019).

If the representatives of both Hamas confirm that Qatar was biased towards neither Fatah nor Hamas, this would be convincing evidence that Qatar was an impartial mediator in the mediation between Fatah and Hamas.

Qatar dealt equally with both sides despite the differences in the ideology and the political platform between Fatah and Hamas. Qatar did not deal with Fatah and Hamas based on ideological considerations. Mishaal (2019) confirms that Qatar dealt with both parties based on political rationality. He adds that the Qataris dealt with both parties as Palestinians. He explains that the negotiations between Fatah and Hamas were Palestinian-Palestinian negotiations (interview with Mishaal, 2019).

Some experts promote the idea that Qatar is biased towards Hamas (Hansen, 2013).

Badran refutes this claim when he explains that those people built their claim that

Qatar hosted Khaled Mishaal and other members of Hamas, which is true. He adds
that this does not mean that Qatar is biased toward Hamas. Badran asserts that

Qatar hosts many different leaders from different countries and different backgrounds (interview with Badran, 2019).

The people who argue that Qatar was biased toward Hamas based their argument on several issues. They based their arguments on Emir Hamad's visit to Qatar in October 2012, the Qatari financial support to Gaza, hosting Khaled Mishaal, Aljazeera's handling Hamas news, and finally, the relations between Qatar and the Muslim brotherhood, which is represented in hosting Yousef Al-Qaradawi.

For Emir Hamad's visit to Gaza in October 2012, Muneer Salameh, Fatah's member, considers the Emir Hamad's visit to Gaza as evidence of the partiality of Qatar towards Hamas. Salameh argues that the visit of the Qatari Emir is a political message from Qatar to say that they are supporting Hamas. Additionally, he considers that the Qatari support to Gaza proves that Qatar is biased toward Hamas (interview with Salamah, 2019). Using the visit of the Qatari Emir to Gaza in 2012 as evidence that Qatar is biased toward Hamas needs to be validated.

Khaled Mishaal explains that it is rational to clarify that the reason for the visit of the Qatari Emir to Gaza and not Ramallah is because the West Bank is under Israeli occupation (interview with Mishaal, 2019). Additionally, Fatah- Hamas Doha Agreement was in February 2012, and the visit of Emir Hamad was in October 2012 after the agreement. This indicates no relationship between the Qatari impartiality and the Emir's visit to Gaza.

Some use Hamas's strong relationship with Qatar and Qatar's welcoming of Khaled Mishaal after he left Syria and settled in Doha to show that Qatar was biased toward Hamas in the negotiations. Furthermore, Qatar supports Hamas' capabilities in Gaza by paying salaries and infrastructure projects in Gaza (Kurz et al., 2018).

Stig Hansen considers the reconciliation between Hamas and Fatah as a central interest in Qatari foreign policy. However, he claims that Qatar was biased in favour of Hamas. Hansen explains that Qatar entered the negotiations with a clear bias towards Hamas, which did not constitute an obstacle to the negotiations between the two movements because Qatar used the money to allay Fatah's fears and attract Fatah's goodwill (Hansen, 2013).

Hansen exemplifies that during the Qatari mediation between Hamas and Fatah, Aljazeera was biased toward Hamas and anti-Fatah. He adds that Aljazeera's previous director, Wadah Khanfar, influenced Aljazeera to cast a positive light on Hamas. Hansen presents an example to show the bias of Qatar towards Hamas. He uses the killing of Hamas leader Ahmed Yassin in 2004 as an example. He confirms that Aljazeera TV producers were clad in black (Hansen, 2013).

The usage of the Aljazeera channel as a tool to describe Qatar as biased toward Hamas is not enough to prove that Qatar was biased toward Hamas in the mediation between Fatah and Hamas. Mishaal (2019) disapproves of Hansen's claim as he says, "Aljazeera sometimes presents news or programs that Hamas does not like...Aljazeera reflects events and does not makeup events...Aljazeera covered the siege of Yasser Arafat...it is not fair to say that Aljazeera is biased toward Fatah because it covers the siege of Yasser Arafat... Aljazeera hosts many of the leaders from both Fatah and Hamas," according to Mishaal (interview with Mishaal, 2019).

Regarding the Qatari relation with Muslim Brotherhood, Hansen explains that Qatar is a country with ties to the Muslim Brotherhood, which seems to have replaced the old Syrian and Iranian allies of Hamas (Hansen, 2013). Hansen explains that Qatar has been the sponsor and the critical backer for the Muslim brotherhood for many

decades. He adds that Qatar has hosted leaders of the Muslim Brotherhood in Doha, providing them with financial support and even granting them the Qatari nationality. Yousuf Alqaradawi is an example. Al Qassemi (2011) and Khatib (2013) claim that Qatar has used the Aljazeera satellite channel as a tool to express public support for the Muslim Brotherhood (Al Qassemi, 2011; Khatib, 2013b). Minich (2015) claims that Qatar's friendly relations with the Egyptian Muslim Brotherhood assisted in signing the Doha Agreement in 2012 (Minich, 2015). However, this does not mean that Qatar was biased toward the Muslim Brotherhood or even Hamas. Qatar hosted many different political figures, for example, the son of Mahmoud Abbas. Hosting him does not mean Qatar is biased in favour of Mahmoud Abbas.

The evidence used above to prove that Qatar was biased toward Hamas are not enough to prove what happened inside the mediation room. None of those claims offers any evidence to show that Qatar was biased toward Hamas or either side through the mediation process and inside the negotiation rooms. One example can refute the claim that Qatar was biased toward Hamas. Mishaal (2019) and Badran (2019) assure that the two parties, Hamas and Fatah, reached a deadlock during the negotiations. They explain that the deadlock was when Fatah and Hamas could not agree on the head of the government. The Emir suggested that Abu Mazen be the head of the government, and Hamas and Fatah freely accepted the suggestion, and it was the hallmark of the Doha agreement (interview with Mishaal and Badran, 2019). If Qatar had been biased toward Hamas, Qatar would suggest that the head of the government would be from Hamas. However, Qatar did not nominate anyone from Hamas. Qatar nominated Mahmoud Abbas to be the head of the

government. Qatar could use the money to allay Fatah's fears and attract Fatah's goodwill. Qatar could pay Mahmoud Abbas to agree. However, Qatar did not do that. It proves that Qatar was impartial in mediation between Fatah and Hamas.

CHAPTER EIGHT: FINDINGS AND CONCLUSION

Introduction

Through engagement in international mediation, different low-leverage states adopt different approaches in international mediation. Therefore, through careful analysis of some of the international mediation cases of two states, Norway and Qatar, this thesis presented two different approaches for low-leverage states in international mediation: norm-driven mediation and interest-driven mediation.

The present thesis answered the main research question: Why do low-leverage states engage differently in international mediation? In addition are the following subquestions: What are the motivations and the capabilities of interest-based low-leverage mediators? What are the motivations and the capabilities of norm-based low-leverage mediators?

Both interest-based low-leverage and norm-based low-leverage mediators have different motivations and different capabilities. Those differences enable the low-leverage states to engage differently in international mediation, which is discussed in detail in the following section.

This chapter is laid out in four sections. The first section provides a table that summarizes the main findings of the thesis. The second section is a discussion on the Norwegian approach to international mediation followed by the Qatari approach to international mediation. The third section discusses norm-based mediation versus interest-based mediation. Finally, the conclusion.

Qatar and Norway in international mediation: a summary of the thesis findings

Table (2) a summary of the thesis findings

	Norway	Qatar
	(Norm-based mediator)	(Interest-based media-
		tor)
Motivation	Norm-based motivation	Interest-based motivation
Capabilities	Track One and a Half di-	1. Economic resources as
(leverage)	plomacy.	incentives to the dispu-
	2. Personal relations be-	tants after reaching an
	tween the Norwegian	agreement.
	NGOs and the disputants	2. Personal Diplomacy
	(trust-building)	represented in the Emir
	3. Economic resources to	and the elite.
	cover the meetings of the	3.The cultural and reli-
	disputants	gious ties between Qatar
	The need of the dispu-	and the disputants.
	tants.	
Impartiality	Not impartial	Impartial
Mediation strategy	Facilitation+ Manipulation	Manipulation

Pure/Power mediator	Have no interest in acting	Have interests and act
	as a pure mediator.	as a power mediator.
	Employs manipulation	2. Offers incentives to the
	strategy and acts as a	disputants=acted as a
	power mediator.	power mediator.
	3. Not impartial mediator=	3. Impartial mediator=
	acted as a power media-	acted as a pure mediator.
	tor	

Why do Qatar and Norway engage differently in international mediation?

Both Norway and Qatar are low-leverage states. However, each of them is engaged differently in international mediation. Norway engages itself as a norm-based mediator in international mediation. Meanwhile, Qatar is classified as an interest-based mediator in international mediation. The two countries behave differently because each of them has a different motivation, capabilities and backgrounds, and different utilization of their economic resources. Despite both of them being considered rich countries with vast economic resources, the employment of their resources is varied in their attempts at international mediation. Norway and Qatar employ their economic resources according to their motivations.

For the difference in motivations, the departure point is that states usually engage in international mediation based on interest; otherwise, they will not be involved in

mediation. It is applied to both norm-based mediators and interest-based mediators. Although the norm-based mediators such as Norway engage themselves in international mediation to promote their international standing and influence in international politics, they gear towards survival by seeking status in the international community. However, they adhere to norms as an essential part of their international mediation and make a balance between its interest and the norms.

In direct contrast, interest-based low-leverage states, Qatar, are mainly engaged in mediations due to interest-based motives. These interests may also include the low-leverage state's investment, security, prestige or international standing. Qatar engages itself in international mediations for security reasons. Qatar believes that being an international mediator would increase its international status and influence in the international community, and by this way, Qatar could secure itself. Accordingly, Qatar considers its international mediation as a tool for expanding its future investment and protecting its current investment. Furthermore, Qatar seeks to promote stability in the Middle East, which is based on morality.

The difference between Qatar and Norway is that Norway makes a balance between the norms and its interest and, Qatar does not make a balance between its interest and the norms of morality. As a result, Qatar engages in mediation to achieve its interest which is represented in security, branding itself, and investment in the disputant's territories.

As a norm-based-low-leverage state, Norway builds its national identity based on universal norms and introduces itself as good stewards working for the common good of the international community. For Norwegians, their peace policy is affected by their

ideas about peace. Their engagement in peace activities is related to the state identity and its values. In their foreign policy and international mediation, Norwegians are guided by normative concepts. Indeed, this is why Norway is called a moral entrepreneur or norm entrepreneur. Norway has developed a national consensus built around social-democratic norms that stress international concerns for those around the world who are oppressed. Acting as a norm entrepreneur, Norway contributes foreign aid to the poor worldwide, seeks to enhance gender equality and the role of women, and endeavours to promote peace by pursuing security and mediation in conflicts. Thus, Norwegian foreign policy is a foreign policy of peace, designed to make the world a better place.

In order to conduct humanitarian activity and peace work, Norway shares its resources with others. It aims to bolster the liberal approach to the world. Additionally, The Norwegians are convinced that such actions are essential for the creation of a better world and believe that they have an essential role to play in the process.

The motivations of both Qatar and Norway are based on their background and history. Being a colony under its neighbours for 500 years enables Norway to understand the experience of being ruled by others. So, the history of Norway reflects on its mediation type.

In the same manner, the history of Qatar, in addition to its location, affect its mediation type and its mediation drive. As an interest-based mediator, Qatar's international mediation was established based on Qatari security due to its vulnerable location between Saudi Arabia and Iran, the two regional hegemons. Qatar had involved itself in

international mediation to protect itself from regional hegemon countries, Saudi Arabia and Iran.

Qatari foreign policy is formulated to increase its international profile to protect itself from the dangers of small state vulnerability and avoid what happened to Kuwait in 1990. Based on its security, Qatar is working on branding itself as a peace mediator who has experience with a track record in international mediation.

For a norm-based mediator, this thesis uses Norway and its mediation in both the Palestinian-Israeli conflict and the Sudanese conflict as examples. It has been found that Norway had no specific, direct self-interest in both cases. Norway has no self-interest in the Palestinian-Israeli conflict, and the Norwegian motivation was to promote peace between the Palestinians and the Israelis. In the same manner, Norway, in its mediation between South Sudan and Sudan, had no self-interest and the Norwegians aimed at stopping the war and bringing peace to the people of Southern Sudan who had suffered much and also the Norwegian goal was to make South Sudan a better place to live.

The motivation of the mediator is connected with the manner of using the resources or leverage. For example, norm-based mediators use their economic resources as humanitarian aids to help the disputants. Offering those aids to the disputants contribute to building trust between Norway and the disputants. In contrast, the interest-based mediator uses economic resources to incentivise the disputants to reach an agreement.

Additionally, the two states have different capabilities or leverage. However, each of them utilises its resources differently. While Norway uses its resources as

humanitarian aid to the disputants to gain their trust of the disputants, Qatar uses its financial resources as incentives to encourage the disputants to reach an agreement. Norway's leverage is represented in its Track One and a Half diplomacy, personal relations between the Norwegian NGOs and the disputants (trust-building), and economic resources to cover the meetings of the disputants. Qatar's leverage is represented in its economic resources as incentives to the disputants after reaching an

Low-leverage states in international mediation: norm-based mediation versus interest-based mediation

Norm-based mediation: Norway mediation in Oslo Accord and the CPA

agreement, personal diplomacy, and cultural and religious ties.

According to the theoretical framework, norm-based mediators are expected to be impartial. They consider impartiality an essential component for their credibility and acceptability in the mediation process and usually adopt a facilitation strategy. What is found in this thesis is different. This section focuses on Norway's mediation strategy to mediate the two conflicts and Norwegian impartiality.

The mediation strategy in Oslo Accord and the CPA

The two cases of Norwegian mediation showed that Norway used a spectrum of mediation strategies in different stages of the mediation process.

In the case of the Oslo Accord, Norway utilized a facilitation strategy at the beginning of negotiation in the first phase of the negotiations, which extended until May 1993. In this stage, a facilitation strategy is essential to create a secure environment of trust

between the Israelis and the PLO representatives. There was no possibility that Norway coerces any parties or intervenes in the mediation process. The Norwegian role was restricted by being a facilitator who conveyed messages between the two disputants. However, this behaviour changed in the second stage of the mediation process. The turning point was the involvement of the new Norwegian foreign minister, Holst. Holst decided to play an active role in the mediation between the Israelis and the PLO. Holst was eager to reach an agreement between the Palestinians and the Israelis. In order to reach an agreement, Holst materially and immaterially manipulated Yasir Arafat. It means that Norway used a manipulation strategy in the second stage of the negotiations. The negotiations at that time reached a deadlock, and Holst realized that negotiations would collapse if he did not psychologically manipulate Yasir Arafat and explain to him the cost of non-agreement. Norway manipulated the needs of Yasir Arafat for a solution that would return him to the regional and international political stage. Additionally, Holst materially manipulated Yasir Arafat by promising him to raise funds for building the Palestinian Authority.

In the mediation of the Sudanese conflict, Norway also used a spectrum of mediation strategies that range from facilitation, formulation and manipulation.

Mostly, Norway was a facilitator who offered a safe environment for the disputants. However, on specific occasions, Norway played a role of a formulator. For example, Norway suggested including self-determination in the mediation substance. After a while of negotiations encouraging the disputants to reach an agreement, Norway promised both sides, the Government of Sudan, and the rebels in South Sudan, to help them invest in the oil industry. Norway manipulated both the Sudanese government and the representative of South Sudan. It is essential to mention that

Norway mediated through Troika in its mediation between South Sudan and Sudan.

Norway was an active mediator, convincing both the USA and the UK to adopt similar positions towards the SPLM. Being a member of the Troika, Norway had a strong position by borrowing the leverage of the USA to coerce the Sudanese government.

In both the Israeli- Palestinian conflict and the Sudan South Sudan conflict, Norway used a spectrum of the three mediation strategies: facilitation, formulation, and manipulation. However, when Norwegians utilized those strategies, they mainly focused on facilitation. Indeed, the Norwegians started with a facilitation strategy, and they sued the formulation strategy in the middle of the negotiations, but the Norwegians used the manipulation strategy to reach an agreement.

However, Norway did not use all manipulation tools because Norway is not a superpower like the USA. For example, one of the manipulation tools is threatening the disputants. Norway cannot threaten the disputants because Norway is not a superpower like the USA. Norway is a low leverage state.

This behaviour of Norway was unexpected. Norway is a norm-based mediator, which means that Norway is expected to adopt the facilitation strategy. Norway adopted the manipulation strategy in its mediation in both the Palestinian-Israeli conflict and the Sudanese conflict,

The Norwegian impartiality

As a norm-based mediator, Norway is expected to be impartial in its mediation. However, Norway was not always an impartial mediator in mediation between the Palestinian-Israeli conflict and the Sudanese conflict. Norway gave away its impartiality in certain stages of the mediation process.

In mediation Oslo Accord, Norway was impartial in the first stage of the mediation until May 1993. However, Norway started to play the role of biased mediator in the second stage of the mediation. Norway was biased toward the strong party, the Israelis. Holst was aware that there would be no agreement without achieving the Israeli demands. Consequently, Holst adopted the Israeli position and manipulated Arafat to accept the Israeli demands. For example, Arafat asked for an extra corridor to connect Gaza with the West Bank. At that time, Israel rejected it, and Norway adopted the Israeli demand and stood for it. The Norwegian practised unhidden bias toward the Israeli side by adopting the Israeli demands and position and considering them as redlines.

In its mediation between the SPLM (South -Sudan rebels) and the government of Sudan, Norway was partial to the weak side, the SPLM. The Norwegians failed to manage the power disparity equitably. It is in direct opposition to the Norwegian behaviour in mediation between strong Israel and the weak Arafat and PLO.

In sum, Norway is expected to be an impartial mediator who employs a facilitation strategy as a norm-based mediator. However, in its mediation in the Oslo Accord 1993 and the Comprehensive Peace Agreement (CPA) in Sudan 2005, Norway was not always an impartial mediator. Norway used a spectrum of the three mediation strategies: facilitation, formulation, and manipulation.

Interest-based mediation: Qatar's mediation DDPD and Fatah- Hamas Doha Agreement.

Interest-based mediators are expected to act as manipulators, and impartiality is unnecessary for them. Studying two cases of Qatari mediation found that Qatar was a manipulator mediator and preserved a position of an impartial mediator who had significant relations with all parties on the same levels.

Mediation strategies in the Qatari mediation

In the case of the mediation Darfur conflict, Qatar used a spectrum of the three mediations strategies in different stages of the mediation process: facilitation, formulation, and manipulation. Before the mediation, Qatar used its leverage to encourage the disputants to accept Qatar to be a mediator. For example, parallel to the Qatari mediation in Darfur in June 2010, the Qatar Charity announced a project of USD 1 billion to improve the livelihoods of the impoverished people living within the affected area.

Qatar used a facilitation strategy during the mediation process by offering the disputants the place for meetings, flight tickets, and hotels. At the end of the negotiations, Qatar used its leverage again and manipulated the disputants by promising them to pay lots of money to invest or rehabilitate Darfur and Sudan's infrastructure.

Qatar used a spectrum of mediation strategies, facilitation, formulation, and manipulation for mediation between Hamas and Fatah. Before the beginning of the mediation process, Qatar offered financial aid to Gaza and investment in West Bank. Qatar usually uses its leverage (finance) before mediation to encourage the disputants to

accept Qatar as a mediator. During the mediation process between Hamas and Fatah, Qatar used a facilitation strategy to offer a safe environment for the disputants to meet. Qatar sought to offer Fatah and Hamas teams a comfortable meeting environment at this stage.

When the negotiations reached a deadlock, and upon the request of the disputants, Qatar played the role of formulator and suggested proposals to resolve the deadlocks. For example, Fatah and Hamas were arguing about the head of the next government. Qatar used its formulation strategy and suggested that Mahmoud Abbas, Abu Mazen, be the head of the next Palestinian government. At the last stage of the negotiations, Qatar used the manipulation strategy. For example, Qatar promised to pay the salaries of the employees in Gaza to reach an agreement. For the Qatari mediation strategy, by using its vast financial resources, Qatar was expected to be playing the role of a manipulator.

Qatar's impartiality in the mediation process

The interest-based mediators are usually concerned about leverage in the mediation, and they do not give much attention to impartiality. Qatar, an interest-based mediator, gives a different example that contradicts the known ones. Qatar is not expected to act as an impartial mediator. However, Qatar shows impartiality in the two cases of mediation discussed in this thesis. Qatar deals with each party in an impartial way and without being biased toward any one party. In its mediation between Hamas and Fatah, Qatar does have a close relationship to Hamas. However, Qatar also enjoys a strong relationship with Mahmoud Abbas (Fatah). For the Qatari mediation in Darfur,

Qatar dealt with both sides as Muslims. Qatar acted as an impartial mediator in the mediation between the rebels in Darfur and the government of Sudan.

In conclusion, Both Qatar and Norway are low-leverage states who represent two different approaches to mediation and engage differently in international mediation.

In order to achieve their motivations, Norway and Qatar employed a spectrum of different mediation strategies. The difference between Norway and Qatar is that Qatar used its leverage before and at the end of the mediation process. Qatar used the leverage before the mediation to pave the way for mediation to encourage the disputant to accept it as a mediator. During the mediation process, Qatar used a facilitation strategy. At the end of the negotiations, Qatar used its leverage again and promised financial aid to manipulate the disputants to reach an agreement.

In brief, the departure point of the Qatari and the Norwegian mediation to gain international status. However, each country behaves differently in international mediation. Norway employs its norm and uses its identity as a norm entrepreneur to be a mediator. Norway uses what is skilful in it, which is norm entrepreneurship. While Norway relies on Track One and a Half diplomacy, personal relations between the Norwegian NGOs and the disputants (trust-building), it uses its economic resources to cover the meetings of the disputants.

In contrast, Qatar, skilful in money investment, relies on its economic resources as promised incentives to promote the disputants to reach an agreement. The Qataris use cultural and religious ties to deal with the disputants with impartiality. Qatar can only be impartial because most of the Qatari mediation trials are between Muslims and Arabs.

Conclusion

This thesis examines two types of low-leverage mediation; norm-based mediation and interest-based mediation represented by Norway and Qatar, respectively.

Based on the literature, norm-based low-leverage mediators should act as impartial mediators who employ a facilitation strategy. In contrast, interest-based low-leverage mediators should be manipulators who do not give much attention to impartiality.

An analysis of two case studies that represented nom-mediation and two case studies that represented interest-based mediation found that norm-based low-leverage mediators use different mediation strategies, the facilitation, the formulation, and the manipulation, in different stages of the mediation process. Those mediators use the facilitation strategy at the beginning of the mediation process and use the formulation studies when required by the disputants. However, they use the manipulation strategy at the end of the mediation process to lead the disputants to sign an agreement. After many years of humanitarian assistance to the disputants, those mediators use their NGOs to build trust between them and the disputants.

Slightly different, the low-leverage-interest-based mediators also use a range of the three known mediation strategies. However, the interest-based mediators introduce their leverage, investment or aid shortly before engagement in the mediation process to encourage. They use the facilitation strategy by offering a safe environment for the disputants to meet and the logistic tools to facilitate the meetings. They use the formulation strategy when the negotiations reach a deadlock. They offer proposals to resolve the deadlocks. At the end of the mediation process and to encourage the

disputants to sign an agreement, they present a manipulation strategy by promising financial investments or aids to the disputants in case they sign an agreement. For impartiality, this thesis found out that norm-based low-leverage mediator was not always an impartial mediator. However, the interest-based mediator preserves impartiality.

This thesis introduces several significant theoretical contributions to the field. This thesis presents two types of low-leverage states in international mediation: normbased mediators and interest -based mediators. It sheds light on the way they both behave in international mediation.

The thesis found that both Norway and Qatar use a spectrum of mediation strategies. Additionally, this thesis found out that Norway was not always an impartial mediator. Moreover, and unexpectedly, Qatar acted as an impartial mediator that preserved the same relations with all parties. The Qatari case shows how a mediator can be impartial (i.e., not favouring a particular party) whilst using manipulative strategies to push for a solution. It is unique because there is an assumption in the literature that impartial mediators tend to be a low stake, low-leverage and hence will refrain from using manipulation.

This thesis could be a beginning or guidance for further studies on low-leverage states in international mediation. This thesis assures that the mediation strategies cannot be classified according to a dichotomy division. The low-leverage estates use a spectrum of those strategies. This thesis could be a tool to raise awareness of the necessity to redefine three important concepts in international mediation: impartiality, manipulation and leverage, in the context of low-leverage states. For example, the

manipulation strategy is different from the manipulation strategy for a superpower like the USA. The USA has wide options of manipulation tools that are not available to low-leverage states, such as threats. The same could be applied to impartiality. In the literature, impartiality means not to be biased (not favouring a particular party). In the thesis, Qatar was impartial because Qatar was favouring the two parties to the same degree. Those concepts were a challenge to the researcher.

In the end, this thesis could be a guideline for the decision-makers in Norway and Qatar to focus on the advantages and disadvantages of their mediation and avoid the shortcomings in any future mediation trials.

In general, Both Qatar and Norway do not represent enough cases to study the behaviour of low-leverage states in international mediation. In order to find out more about the behaviour of low-leverage states in international mediation and whether they are norm-based mediators or interest-based mediators, it is recommended that there should be more studies that include other such cases. In order to deeply understand the interest-based mediator such as Qatar, it is recommended to include the Qatari mediation between USA and Taliban in any future study or any other mediation case that took place after 2013. The two cases of the Qatari mediation in this thesis are before 2013, which was in the age of Emir Hamad, the father. The current Emir, Tamim, became Emir in 2013. Although he conforms to follow his father steps, there may be differences in the Qatari mediation in the era of Emir Tamim. The same for the Norwegian mediation, it should be further studies that include the Norwegian latest mediation trials.

In order to conduct this thesis, some obstacles were faced in this thesis. Those obstacles are represented by finance shortages, security concerns, difficulties in reaching the expected participants, and sometimes refusal by participants to be interviewed. The interviews represented the primary data for this thesis and required travel to be completed, and the researcher needed to conduct plenty of interviews. However, travel was sometimes not possible due to financial shortages and security issues. For example, travelling to places such as Sudan, South Sudan and Palestinian Authority was not recommended due to security issues as those places are considered not safe places. However, the researcher managed to overcome those obstacles by conducting most interviews via telephone or the internet. The University of Birmingham allowed the researcher to use the University telephones for interviewing the participants in the USA, Israel, Sudan, and Palestinian Authority.

Another obstacle is represented in the difficulties in interviewing some of the nominated participants. Some of the participants refused to be interviewed due to political reasons. For the interviewees who refused to be interviewed, the researcher succeeded in finding alternatives to them and keeping the identities anonymous of the replacement interviewees.

REFERENCES

Interviews:

Abu-Koush, M. A (2019). Palestinian negotiator in Oslo Back Channel, 17 February.

Abu-Marzook, M.(2019). Hamas representative, 25 February.

Al-Ahmad, A. (2019). Fatah representative, 5 March.

AL-Ansari, M. (2019). Professor at Qatar University, 27 May.

Al-Awad, W. (2019). A representative of the Palestine People's Party, 9 February.

Al-Hamadi, A. (2019). Qatari ex-embassador to Sudan, (29 May.

Al-Khater, L.(2019). The spoksperson for the Qatari Ministry of Foreign Affairs, 27 May.

Amum, P. (2019). SPLM representative,7 May.

Asfour, H. (2018). A Palestinian negotiator in Oslo back channel, 27 December.

Badran, H. (2019). Hamas representative, 26 February.

Beilin, Y. (2019). An Isreali politician who founded the Oslo channel, 28 January.

Egeland, J. (2019). The Norwegian ex-foreign minister, 17 February.

Elkhalifa, A.(2019). The Governmnt of Sudan representative, 21 April.

Emadi, M. (2019). The Qatari envoy to the Palestinian Authority, 24 February.

Gur, S. (2019). The head of Shemon Peres Office, 4. February.

Hanssen, H. J. (2019). A representative of the Norwegian People's Aid, NPA, 11 March.

Hassan Omar, A. (2019). The representative of the Government of Sudan, 23 March.

Hødnebø, K. (2019). A Norwegian Diplomat who participated in the CPA mediation, 14

March

Hødnebø, K. (2019). A Norwegian Diplomat who participated in the CPA mediation, 16 April.

Hødnebø, K. (2020). A Norwegian Diplomat who participated in the CPA mediation, 16 April.

Johnson, H. F. (2019). The Norwegian ex-Minster of International Development, 15 March.

Larsen, T. (2019). The ex-head of FAFO and the Norwegian representative in Oslo Back Channel, 29 February.

Al-Ansari, M. (2019). A professor at Qatar University, 27 May.

Mishaal, K. (2019). Hamas ex-leader, 28 February.

Palestinian senior official. (2020), 1February.

Palestinian senior official. (2019), 18 February.

Palestinian senior official. (2019),17 February.

Saed Erekat.(2019). A member of PLO, 18 February.

Salamah, M. (2019). Fatah representative, 2 February.

Singer, Y. (2019). An Isreali representative in Oslo Back Channel, 28 January.

Sisi, T. (2019). The representative of Darfur in the negotiations between Darfur and the Government of Sudan, 9 April.

Sumbeiywo, L. (2019). The main mediator in CPA and the head of IGAD, 17 May.

Taguia, H. (2019). The head of Aljazzera research center, 27 May.

Waage, H. (2019). A Norwegian professor, 31 January.

Zwairi, M. (2019). A professor at Qatar University, 27 May.

Documents

- Curtis, M. (2010). *Doublethink: The two faces of Norway's foreign and development policy.* Paper presented at the Forum for Environment and Development: Oslo.
- Egeland, J. (1998). Peacemaking and the Prevention of Violence in Situations of
 Internal Strife and Intrastate Conflict: The Role of Governments and NGOs.
 Retrieved from
- Goonatilake, S. (2004). *Norway, a 25 year odyssey: From sympathizer to colonial intruder.* Paper presented at the Conference on Road Map to Peace in Sri Lanka in Oslo, Norway.
- Hanssen-Bauer, J. (2005). *The Norwegian "model" for conflict resolution*. Paper presented at the Conferência sobre o Prêmio Nobel da Paz e o trabalho para paz da Noruega.

- Helgesen, V. (2003). Peace, mediation and reconciliation: The Norwegian experience. *Norwegian State Secretary of Foreign Affairs presentation, Brussels, 21*.
- Hølst, J. (1993). Memo to Shemon Perse from the Norwegian Foreign minister Johan Hølst.
- Holst, J. J. (1983). *Norway's Role in the Search for International peace and Security*:

 Norsk utenrikspolitisk institutt.

Books

- Agha, H., Feldman, S., Khalidi, A., & Schiff, Z. (2003). *Track-II Diplomacy: Lessons from the Middle East*: Mit Press Cambridge.
- Al-Ramahi, A. (2008). Sulh: A crucial part of Islamic Arbitration.
- Aruri, N. (2003). Dishonest broker: The role of the United States in Palestine and Israel: South End.
- Ashrawi, H. (1996). This side of peace: A personal account: Simon and Schuster.
- Assefa, H. (1987). *Mediation of Civil Wars: Approaches and Strategies--the Sudan Conflict*: Westview Press.
- Barston, R. P. (2014). *Modern diplomacy*: Routledge.
- Baskarada, S. (2014). Qualitative case study guidelines.

- Beardsley, K. (2011). The mediation dilemma: Cornell University Press.
- Beilin, Y., & Simpson, P. (1999). *Touching peace: From the Oslo Accord to a final agreement*: Weidenfeld & Nicolson London.
- Bennett, A., & George, A. (2004). Case study and theory development in social sciences. In: Cambridge, Mass: MIT Press.
- Bercovitch, J. (2004). International mediation and intractable conflict.
- Bercovitch, J. (2011a). Conflict management and the Oslo experience: Assessing the success of Israeli–Palestinian peacemaking. In *Theory and Practice of International Mediation* (pp. 149-166): Routledge.
- Bercovitch, J. (2011b). Theory and practice of international mediation: selected essays: Routledge.
- Bercovitch, J., Kremenyuk, V., & Zartman, I. W. (2008). *The SAGE handbook of conflict resolution*: Sage.
- Bersagel, A. G. (2008). *Multi-Party Mediation in Intrastate Conflicts:*" The Norwegian Model in Practice".
- Bien, W. J., Bouckaert, P., Hanson, A., Khachikian, A., King, K., Laudy, M., . . . Tang, R. (2000). Words over war: Mediation and arbitration to prevent deadly conflict: Rowman & Littlefield Publishers.
- Bingham, G. (1986). Resolving environmental disputes: A decade of experience:

 Conservation Foundation Washington, DC.
- Blanchard, C. M. (2011). Qatar: Background and US relations: DIANE Publishing.

- Brysk, A. (2009). *Global good Samaritans: human rights as foreign policy*: Oxford University Press.
- Burr, M., & Collins, R. O. (2008). *Darfur: The long road to disaster*. Markus Wiener Publishers.
- Carter, J. (1995). Talking peace: A vision for the next generation: Puffin.
- Chai, S.-K. (2001). Choosing an identity: A general model of preference and belief formation. University of Michigan Press.
- Chehab, Z. (2007). *Inside Hamas: The untold story of the militant Islamic movement*:

 Nation Books.
- Copnall, J. (2014). A poisonous thorn in our hearts: Sudan and South Sudan's bitter and incomplete divorce. Hurst.
- Corbin, J. (1994a). Gaza first: The secret Norway channel to peace between Israel and the PLO: Bloomsbury Pub.
- Corbin, J. (1994b). The Norway channel: The secret talks that led to the Middle East peace accord: Atlantic Monthly Press.
- Creswell, J. W. (1994). Research design: Qualitative & quantitative approaches: Sage Publications, Inc.
- Crocker, C. A., Hampson, F. O., & Aall, P. (2001a). Is more better? The pros and cons of multiparty mediation. *Turbulent Peace: The Challenges of Managing International Conflict*, 497-513.

- Crocker, C. A., Hampson, F. O., & Aall, P. R. (2001b). *Turbulent Peace: the challenges of managing international conflict*: United States Inst of Peace Pr.
- Crocker, C. A., Hampson, F. O., & Aall, P. R. (2004). *Taming intractable conflicts:*Mediation in the hardest cases: US Institute of Peace Press.
- Crocker, C. A., Hampson, F. O., & Aall, P. R. (2007). *Leashing the dogs of war:*conflict management in a divided world: US Institute of Peace Press.
- Dagne, T. (2010). Sudan: The Crisis in Darfur and Status of the North-South Peace

 Agreement: Diane Publishing.
- Damen, R. (Writer) & A. Channel (Director). (2013). The Price of Oslo. In.
- Davis, L. (2014). *EU Foreign Policy, Transitional Justice, and Mediation*: Abingdon: Routledge.
- Daymon, C., & Holloway, I. (2010). Qualitative research methods in public relations and marketing communications: Routledge.
- De Waal, A. (2007b). War in Darfur and the Search for Peace: Harvard University Press.
- de Wall, A., & Flint, J. (2005). Darfur: A Short History of a Long War. In: London: Zed Books.
- Deng, F. M. (2011). War of visions: Conflict of identities in the Sudan. Brookings Institution Press.
- Doherty, R. E. (2010). Wealthy Qatar tries to build niche as conflict mediator. In: Reuters.

- Eban, A. S. (1983). *The New Diplomacy: International Affairs in the Modern Age*: Random House Inc.
- Egeland, J. (1988). Impotent Superpower-Potent Small State: Potentials and

 Limitations of Human Rights Objectives in the Foreign Policies of the United

 States and Norway: Oxford University Press, USA.
- Egeland, J. (2008). A billion lives: An eyewitness report from the frontlines of humanity: Simon and Schuster.
- Elon, A. (1993). The peacemakers. The New Yorker, 20.
- Eriksson, J. (2015). Small-State Mediation in International Conflicts: Diplomacy and Negotiations in Israel-Palestine. New York: I.B.Tauris.
- Fazal, T. M. (2011). State death: the politics and geography of conquest, occupation, and annexation: Princeton University Press.
- Fierke, K. M. (2015). Critical approaches to international security: John Wiley & Sons.
- Flint, J. (2010). Rhetoric and Reality: the failure to resolve the Darfur conflict:

 Citeseer.
- Flint, J., & De Waal, A. (2008). Darfur: a new history of a long war: Zed Books Ltd.
- Folberg, J., & Taylor, A. (1986). Mediation: A comprehensive guide to resolving conflicts without litigation.
- Fromherz, A. J. (2017). *Qatar: A modern history*: Georgetown University Press.
- George, A. L., & Bennett, A. (2005). Case studies and theory development in the social sciences: mit Press.

- Gerner, D. J. (2018). One land, two peoples: The conflict over Palestine: Routledge.
- Ghimire, B. (2017). Small States Suffer When Big States Bark. Fair Observer.
- Goetschel, L. (2013b). Small states inside and outside the European Union: interests and policies: Springer Science & Business Media.
- Goonatilake, S. (2006). Recolonisation: Foreign Funded NGOs in Sri Lanka: Sage.
- Gulbrandsen, A. (2010). Bridging the Gulf: Qatari business diplomacy and conflict mediation.
- Hammond, A. (2014). *Qatar's Leadership Transition: Like Father, Like Son:*Universitäts-und Landesbibliothek Sachsen-Anhalt.
- Hanf, K., & Soetendorp, B. (2014). Adapting to European integration: small states and the European Union: Routledge.
- Hansen, B., Toft, P., & Wivel, A. (2008). Security Strategies and American World

 Order: Lost Power: Routledge.
- Hansen, S. J. (2013). Religion, Prestige and Windows of Opportunity?: Qatari Peace

 Making and Foreign Policy Engagement: Noragric.
- Hanssen, H. J. (2017). Lives at Stake Norway: Skyline.
- Harris, P., & Reilly, B. (1998). *Democracy and deep-rooted conflict: options for negotiators*: International Institute for Democracy and Electoral Assistance.
- Hellmüller, S., Federer, J. P., & Zeller, M. (2015). The role of norms in international peace mediation. *Bern et al.: swisspeace/NOREF*.
- Helmreich, W. (2017). The Third Reich and the Palestine Question: Routledge.

- Hey, J. A. (2003). *Small states in world politics: Explaining foreign policy behavior.*Lynne Rienner Publishers.
- Hirschfeld, Y. P. (2014). *Track-two diplomacy toward an Israeli-Palestinian solution,*1978-2014: Woodrow Wilson Center Press Washington, DC.
- Hopmann, P. T. (1996). *The negotiation process and the resolution of international conflicts*: Univ of South Carolina Pr.
- Huntington, S. P. (2006). *Political order in changing societies*: Yale University Press.
- Ingebritsen, C. (2006). Scandinavia in world politics: Rowman & Littlefield.
- International Crisis Group. (2002a). Capturing the moment: Sudan's peace process in the balance: International Crisis Group.
- Ishaq, N. (2012). Hamas: Palestinian Political Power and the Arab Uprisings.
- Jakobsen, P. V. (2012). *Nordic approaches to peace operations: A new model in the making*: Routledge.
- Jesse, N. G., & Dreyer, J. R. (2016). Small States in the International System: At Peace and at War. Lexington books.
- Jesse, N. G., & Williams, K. P. (2010). Ethnic conflict: a systematic approach to cases of conflict: CQ Press.
- Johnson, H. F. (2011). Waging peace in Sudan: The inside story of the negotiations that ended Africa's longest civil war. Apollo Books.
- Johnson, H. F. (2016). South Sudan: The untold story from independence to the civil war. IB Tauris.

- Jonas, S. (2018). Of centaurs and doves: Guatemala's peace process: Routledge.
- Jones, D. (1999). Cosmopolitan Mediation?: Conflict Resolution and the Oslo

 Accords: Manchester University Press.
- Jones, L. (2011). Doha agreement could actually worsen chances for peace in Darfur. *Christian Science Monitor*.
- Jones, P. (2015). *Track two diplomacy in theory and practice*: Stanford University Press.
- Kaiya, G. (2016). *Norway's engagements in South Sudan's peacebuilding process.*Norwegian University of Life Sciences, Ås,
- Kaldor, M. (2013). New and old wars: Organised violence in a global era: John Wiley & Sons.
- Kamrava, M. (2015). Qatar: Small state, big politics: Cornell University Press.
- Katzenstein, P. J. (1985). *Small states in world markets: Industrial policy in Europe*:

 Cornell University Press.
- Katzenstein, P. J. (1996). *The culture of national security: Norms and identity in world politics*: Columbia University Press.
- Katzman, K. (2016). Qatar: Governance, Security, and US Policy. Retrieved from
- Kaussler, B. (2015a). Tracing Qatar's Foreign Policy and Its Impact on Regional Security. *Research Paper*.
- Kaussler, B. (2015b). Tracing Qatar's Foreign Policy and Its Impact on Regional Security. *Arab Center for Research and Policy Studies. Doha, Doha Institute*.

- Kelman, H. C., Bercovitch, J., & Rubin, J. Z. (1992). Mediation in International Relations: Multiple Approaches to Conflict Management.
- Kleiboer, M. (1998). *The multiple realities of international mediation*: Lynne Rienner Publishers.
- Krane, J., & Wright, S. (2014). Qatar 'rises above'its region: Geopolitics and the rejection of the GCC gas market.
- Lagu, J. (2006). Sudan: odyssey through a state: from ruin to hope: MOB Center for Sudanese Studies, Omdurman Ahlia University.
- Lakatos, I. (2017). The Potential Role of Small States and Their Niche Diplomacy at the UN and in the Field of Human Rights, with Special Attention to Montenegro. *Pecs J. Int'l & Eur. L.*, 58.
- Lall, A. S. (2014). *Multilateral negotiation and mediation: instruments and methods*: Elsevier.
- Larres, K. (2002). Churchill's cold war: the politics of personal diplomacy: Yale University Press.
- Larssen, I. (2009). Peace and human rights: a comparative analysis on the role of human rights in Norwegian peace processes in Sudan. Universitetet i Tromsø,
- Mac Ginty, R. (2008b). Traditional and indigenous approaches to peacemaking. In Contemporary Peacemaking (pp. 120-130): Springer.
- Magnette, P., & Nicolaïdis, K. (2005). Coping with the Lilliput syndrome: large vs. small Member States in the European Convention. *Eur. Pub. L., 11*, 85.

- Mahmud'Abbas. (1995). Through Secret Channels: The Road to Oslo; Senior PLO

 Leader Abu Mazen's Revealing Story of the Negotiations with Israel: Garnet.
- Makovsky, D. (2018). Making Peace with the PLO: The Rabin Government's Road to the Oslo Accord: Routledge.
- Marsh, S. (2017). Personal diplomacy at the summit. In *Churchill and the Anglo- American Special Relationship* (pp. 124-149): Routledge.
- Mason, M. (2010). Sample size and saturation in PhD studies using qualitative interviews. Paper presented at the Forum qualitative Sozialforschung/Forum: qualitative social research.
- Merriam, S. B., & Tisdell, E. J. (2015). *Qualitative research: A guide to design and implementation*: John Wiley & Sons.
- Milton-Edwards, B., & Farrell, S. (2010). *Hamas: the Islamic resistance movement*: Polity.
- Mishal, S., & Sela, A. (2006). *The Palestinian Hamas: vision, violence, and coexistence*: Columbia University Press.
- Mitchell, C. (2003). Mediation and the Ending of Conflicts. In *Contemporary Peacemaking* (pp. 77-86): Springer.
- Mitchell, C. R. (1988). New approaches to international mediation: Praeger Pub Text.
- Moore, C. W. (2014). The mediation process: Practical strategies for resolving conflict: John Wiley & Sons.
- Moran, D. (2009). Qatar Steps into the Breach. Security Watch.

- Morro, P. (2007). International Reaction to the Palestinian Unity Government.
- Nan, S. A. (2005). Contributions to Georgian-South Ossetian Peacemaking. *Paving the way: Contributions of interactive conflict resolution to peacemaking*, 161.
- Nathan, L. (2006). *No Ownership, No Peace: The Darfur Peace Agreement*: Crisis States Research Centre London.
- Natsios, A. S. (2012). Sudan, South Sudan, and Darfur: what everyone needs to know: OUP USA.
- Neumann, I. B. (2004). Lilliputians in Gulliver's World?: small states in international relations.
- Nye, J. S. (2010). The new public diplomacy. *Project syndicate*, 10.
- Powers, S., & Gilboa, E. (2007). The public diplomacy of Aljazeera. In *New media* and the new *Middle East* (pp. 53-80): Springer.
- Prendergast, J., Galic, M., Dubey, R., Roessler, P., & Macgregor, D. (2002). God, oil and country: changing the logic of war in Sudan. *ICG, London*.
- Princen, T. (2014). Intermediaries in international conflict: Princeton University Press.
- Pulliam, S. (2013). Qatar's Foreign Policy: Building an International Image.

 Khamasin: The Journal of the American University in Cairo's Department of Political Science, 3.
- Qurei, A. (2006). From Oslo to Jerusalem: The Palestinian story of the secret negotiations: Ib tauris.

- Rabin, Y. (1993). *Peace and Israel's National Priorities*. Paper presented at the talk given to the Conference on Peacemaking One Year after Madrid at Tel Aviv University.
- Ravitch, S. M., & Carl, N. M. (2015). *Qualitative research: Bridging the conceptual, theoretical, and methodological*: Sage Publications.
- Roberts, D. (2010). Qatari Mediation. *University of Durham (nd)*.
- Rolandsen, Ø. (2005). Guerrilla Government: political changes in the Southern Sudan during the 1990s: Nordic Africa Institute.
- Rothstein, R. L. (1968). *Alliances and small powers*: Columbia University Press New York.
- Schanzer, J. (2008). Hamas vs. Fatah: the struggle for Palestine: St. Martin's Press.
- Schulze, K. E. (2013). The arab-israeli conflict: Routledge.
- Shanmugaratnam, N. (2008). Between war & peace in Sudan & Sri Lanka: deprivation & livelihood revival: James Currey.
- Shanmugaratnam, N., Mamer, D., & Kenyi, M. R. (2002). From emergency relief to local development and civil society building: a review of Norwegian People's Aid's interventions in Southern Sudan. *DCG report*.
- Shemesh, M. (2012). *The Palestinian entity 1959-1974: Arab politics and the PLO*: Routledge.
- Singer, L. (2018). Settling disputes: Conflict resolution in business, families, and the legal system: Routledge.

- Sisk, T. D. (1996). Power sharing and international mediation in ethnic conflicts: US Institute of Peace Press.
- Sisk, T. D. (2009). *International mediation in civil wars: bargaining with bullets*:

 Routledge.
- Slim, R. M. (1992). Small-state mediation in international relations: the Algerian mediation of the Iranian hostage crisis. In *Mediation in international relations* (pp. 206-231): Springer.
- Sørbø, G. M., Goodhand, J., Klem, B., Nissen, A. E., & Selbervik, H. (2011). *Pawns of peace. Evaluation of Norwegian peace efforts in Sri Lanka, 1997-2009*:

 Norad.
- Spiegel, M. A. (1985). The Namibia negotiations and the problem of neutrality. *International Mediation in Theory and Practice, Boulder, CO*, 111-139.
- Stake, R. E. (1995). The art of case study research: Sage.
- Stake, R. E. (2013). Multiple case study analysis: Guilford Press.
- Steinmetz, R. (2016). Small states in Europe: challenges and opportunities:

 Routledge.
- Tamimi, A. (2009). Hamas: unwritten chapters: Hurst & Company.
- Taulbee, J., Kelleher, A., & Grosvenor, P. (2014). Norway's peace policy: Soft power in a turbulent world: Springer.
- Tessler, M. (2009). A history of the Israeli-Palestinian conflict: Indiana University Press.

- Thomas, G. (2017). How to Do Your Research Project: A Guide for Students: Sage.
- Thorhallsson, B. (2017). The role of small states in the European Union: Routledge.
- Touval, S. (1982). The peace brokers: Mediators in the Arab-Israeli conflict, 1948-1979: Princeton University Press.
- Touval, S. (1992). The superpowers as mediators. In *Mediation in international* relations (pp. 232-248): Springer.
- Touval, S., & Zartman, I. W. (1985). *International mediation in theory and practice*:

 Westview Press with the Foreign Policy Institute, School of Advanced

 International Studies, Johns Hopkins University.
- Tvedt, T., & Badal, R. K. (1994). Short-cut to decay: the case of the Sudan: Nordic Africa Institute.
- Ulrichsen, C. C. (2013). Qatar's Mediation initiatives. *Norwegian Peacebuilding Resource Centre, Policy Brief (February 2013)*.
- Ulrichsen, K. (2014). Qatar and the Arab Spring: Oxford University Press.
- Urichsen, K. (2014). Qatar and the Arab Spring: Policy Drivers and Regional Implications. In: Carnegie Endowment for International Peace.
- Volkan, V. D. (1991). Official and unofficial diplomacy: An overview. The Psychodynamic of International Relationships: Unofficial Diplomacy at Work, edited by VD Volkan, D. Julius, and J. Montville. Lexington, MA: Lexington Books, 2.

- Vuković, S. (2014). Three Degrees of Success in International Mediation. In: Sage Publications Sage UK: London, England.
- Waihenya, W. (2006). *The Mediator: Gen. Lazaro Sumbeiywo and the Southern*Sudan Peace Process: East African Educ Press.
- Walraven, K. v. (1999). Conflict policy in some Western countries: some explorative notes.
- Walter, B. F. (2002). *Committing to peace: The successful settlement of civil wars*:

 Princeton University Press.
- White, G. (2005). The Oslo War Process. Section, 6, 2-3.
- Wilkenfeld, J., Young, K., Quinn, D., & Asal, V. (2007). *Mediating international crises*:

 Routledge.
- Williams, P. R., & Simpson, M. (2011). Drafting in Doha: an assessment of the Darfur peace process and ceasefire agreements.
- Wivel, A., & Mouritzen, H. (2004). *The geopolitics of Euro-Atlantic integration*: Routledge.
- World Bank. (2011). Building the Palestinian State: Sustaining Growth, Institutions, and Service Delivery. In: World Bank Jerusalem.
- Yaari, E., & Zilber, N. (2014). The Hamas-Fatah Reconciliation Agreement: Too Early to Judge. Washington Institute. In.
- Yarrow, C. M. (1978). *Quaker experiences in international conciliation*: Yale University Press.

- Yin, R. K. (2013). Case study research: Design and methods Sage
- Yin, R. K. (2014). Case Study Research Design and Methods (Vol. 5). CA: Sage:

 Thousand Oaks,.
- Yin, R. K. (2017). Case study research and applications: Design and methods: Sage publications.
- Young, O. R. (2015). *The intermediaries: Third parties in international crises*:

 Princeton University Press.
- Zartman, I. W. (2007). *Peacemaking in international conflict: Methods & techniques*: US Institute of Peace Press.

Journal Articles

- Abusharaf, R. M. (2010). Debating Darfur in the world. *The ANNALS of the American Academy of Political and Social Science*, 632(1), 67-85.
- Acland, A. F. (1993). Sudden outbreak of common sense. *Army Quarterly & Defence Journal*, 123(3), 382.
- Aggestam, K. (1996). Two-Track Diplomacy: Negotiations Between Israel and the PLO Through Open and Secret Channels. *Davis Papers on Israel's Foreign Policy*(53).
- Al Qassemi, S. S. (2011). How Saudi Arabia and Qatar became friends again.

 Foreign Policy, 21.

- Alesina, A., Özler, S., Roubini, N., & Swagel, P. (1996). Political instability and economic growth. *Journal of Economic growth*, *1*(2), 189-211.
- Alesina, A., & Rodrik, D. (1994). Distributive politics and economic growth. *The quarterly journal of economics*, *109*(2), 465-490.
- Amer, A. A. (2013). Hamas ties to Qatar have cost. Al-Monitor Palestine Pulse.
- Antwi-Boateng, O. (2014). The rise of Qatar as a soft power and the challenges. *European Scientific Journal, ESJ, 9*(10).
- Anziska, S. (2018). *Preventing Palestine: A Political History from Camp David to Oslo*: Princeton University Press.
- Archer, C., & Nugent, N. (2002). Introduction: small states and the European Union.

 Current politics and economics of Europe, 11(1), 1-10.
- Aruri, N. H. (2003). *Dishonest broker: the US role in Israel and Palestine*: South End Pr.
- Baillie, S. (1998). The position of small states in the EU. In *Small States Inside and Outside the European Union* (pp. 193-205): Springer.
- Bandarage, A. (2011). The Norwegian Model: Political Economy of NGO
 Peacemaking. *The Brown Journal of World Affairs*, 17(2), 221-242.
- Barakat, S., 2012. *The Qatari spring: Qatar's emerging role in peace-making. London: LSE.* Kuwait Programme Working Paper 24.

- Barakat, S. (2014). *Qatari mediation: between ambition and achievement*: Brookings Institution.
- Beardsley, K. (2008). Agreement without peace? International mediation and time inconsistency problems. *American Journal of Political Science*, *52*(4), 723-740.
- Beardsley, K. (2009). Intervention without leverage: Explaining the prevalence of weak mediators. *International Interactions*, *35*(3), 272-297.
- Beardsley, K. (2013). Using the right tool for the job: Mediator leverage and conflict resolution. *Penn St. JL & Int'l Aff.*, 2, 57.
- Beardsley, K., & Lo, N. (2014). Third-party conflict management and the willingness to make concessions. *Journal of Conflict Resolution*, *58*(2), 363-392.
- Beardsley, K. C., Quinn, D. M., Biswas, B., & Wilkenfeld, J. (2006). Mediation style and crisis outcomes. *Journal of Conflict Resolution*, *50*(1), 58-86.
- Beber, B. (2012). International mediation, selection effects, and the question of bias.

 Conflict Management and Peace Science, 29(4), 397-424.
- Bercovitch, J. (1997). Mediation in international conflict: An overview of theory, a review of practice. *Peacemaking in international conflict: Methods and techniques*, 125-154.
- Bercovitch, J., Anagnoson, J. T., & Wille, D. L. (1991). Some conceptual issues and empirical trends in the study of successful mediation in international relations. *Journal of Peace Research*, 28(1), 7-17.

- Bercovitch, J., & Fretter, J. (2007). Studying international mediation: Developing data sets on mediation, looking for patterns, and searching for answers.

 International Negotiation, 12(2), 145-173.
- Bercovitch, J., & Houston, A. (2000). Why do they do it like this? An analysis of the factors influencing mediation behavior in international conflicts. *Journal of Conflict Resolution*, *44*(2), 170-202.
- Bertram, E. (1995). Reinventing governments: The promise and perils of United Nations peace building. *Journal of Conflict Resolution*, 39(3), 387-418.
- Bjola, C. (2014). The ethics of secret diplomacy: a contextual approach. *Journal of Global Ethics*, *10*(1), 85-100.
- Björkdahl, A. (2002). From Idea to Norm-Promoting Conflict Prevention: Department of Political Science, Lund University.
- Björkdahl, A. (2013). Ideas and norms in Swedish peace policy. *Swiss political science review*, 19(3), 322-337.
- Böhmelt, T. (2010). The effectiveness of tracks of diplomacy strategies in third-party interventions. *Journal of Peace Research*, 47(2), 167-178.
- Böhmelt, T. (2011). Disaggregating mediations: The impact of multiparty mediation.

 *British Journal of Political Science, 41(4), 859-881.
- Böhmelt, T. (2015). The spatial contagion of international mediation. *Conflict Management and Peace Science*, 32(1), 108-127.

- Braveboy-Wagner, J. (2003). The English-speaking Caribbean states: a triad of foreign policies. *Small states in world politics: explaining foreign policy behavior*, 31-51.
- Bremmer, I. (2009). State capitalism comes of age: the end of the free market? Foreign Affairs, 40-55.
- Brown, N. J. (2010). The Hamas-Fatah Conflict: Shallow but Wide. *Fletcher F. World Aff.*, 34, 35.
- Bruck, C. (1996). The wounds of peace. The New Yorker, 14, 78-79.
- Brynjar, L. (1998). Implementing the Oslo Peace Accords: A Case Study of the Palestinian-Israeli Peace Process and International Assistance for the Enhancement of Security: Forsvarets forskningsinstitutt.
- Brysk, A. (2005). Global Good Samaritans? Human Rights Foreign Policy in Costa Rica. *Global Governance: A Review of Multilateralism and International Organizations*, 11(4), 445-466.
- Carnevale, P. J. (1986). Strategic choice in mediation. *Negotiation Journal*, *2*(1), 41-56.
- Carnevale, P. J. (2002). Mediating from strength. *Studies in International Mediation*, 25-40.
- Carnevale, P. J., & Arad, S. (1996). Bias and impartiality in international mediation.

 Resolving international conflicts: The theory and practice of mediation, 39-53.

- Carnevale, P. J., & Pegnetter, R. (1985). The selection of mediation tactics in public sector disputes: A contingency analysis. *Journal of Social Issues, 41*(2), 65-81.
- Cho, J., & Trent, A. (2006). Validity in qualitative research revisited. *Qualitative research*, 6(3), 319-340.
- Collier, D. (2011). Understanding process tracing. *PS: Political Science & Politics,* 44(4), 823-830.
- Cooper, A. F., & Momani, B. (2011). Qatar and expanded contours of small state diplomacy. *The International Spectator*, *46*(3), 113-128.
- Cull, N. J. (2009). Public diplomacy: Lessons from the past. *CPD Perspectives on Public diplomacy*, *2*(19).
- Darke, P., Shanks, G., & Broadbent, M. (1998). Successfully completing case study research: combining rigour, relevance and pragmatism. *Information systems journal*, 8(4), 273-289.
- De Waal, A. (2007a). Darfur and the Failure of the Responsibility to Protect.

 International Affairs, 83(6), 1039-1054.
- Dickinson, E. (2012a). Qatar builds a brand as mediator. *Christian Science Monitor*, 28.
- Dooley, L. M. (2002). Case study research and theory building. *Advances in developing human resources*, *4*(3), 335-354.

- Egeland, J., 1998, March. Peace-making and the prevention of violence in situations of internal strife and intrastate conflict: The role of governments and NGOs. In Inter-American Development Bank Seminar, Cartagena de Indias (Vol. 14).
- Ennis, C. A., & Momani, B. (2013). Shaping the Middle East in the midst of the Arab uprisings: Turkish and Saudi foreign policy strategies. *Third World Quarterly*, 34(6), 1127-1144.
- Farsakh, L. (2017). The "Right to Have Rights": Partition and Palestinian Self-Determination. *Journal of Palestine Studies*, 47(1), 56-68.
- Finnemore, M., & Sikkink, K. (1998). International norm dynamics and political change. *International organization*, *52*(4), 887-917.
- Fisher, R. J. (2016). Assessing the contingency model of third-party intervention in successful cases of prenegotiation. In *Ronald J. Fisher: A North American Pioneer in Interactive Conflict Resolution* (pp. 133-153): Springer.
- Fisher, R. J., & Keashly, L. (1991). The potential complementarity of mediation and consultation within a contingency model of third party intervention. *Journal of Peace Research*, 28(1), 29-42.
- Ford, P. (2000). Norway as Peacemaker. Christian Science Monitor, 31.
- Foster, M. (2011). Small States in Peacemaking Roles. *Jackson Shoool of International Studies*, 2(1).
- Gerami, A. (2009). Bridging the theory-and-practice gap: Mediator power in practice.

 Conflict Resolution Quarterly, 26(4), 433-451.

- Gigleux, V. (2016). Explaining the diversity of small states' foreign policies through role theory. *Third World Thematics: A TWQ Journal, 1*(1), 27-45.
- Gilboa, E. (1998). Secret diplomacy in the television age. *Gazette (Leiden, Netherlands)*, 60(3), 211-225.
- Goetschel, L. (2013a). Introduction to Special Issue: Bound to be peaceful? The changing approach of Western European small states to peace. *Swiss political science review*, 19(3), 259-278.
- Goonatilake, S. (2005). Norway, a 25 year odyssey: From sympathizer to colonial intruder. *Peace in Sri Lanka: Obstacles and Opportunities, World Alliance for Peace in Sri Lanka (Australia*, 22-23.
- Greig, J. M., & Regan, P. M. (2008). When do they say yes? An analysis of the willingness to offer and accept mediation in civil wars. *International Studies Quarterly*, *52*(4), 759-781.
- Grøn, C. H., & Wivel, A. (2011). Maximizing influence in the European Union after the Lisbon Treaty: From small state policy to smart state strategy. *Journal of European Integration*, 33(5), 523-539.
- Group, I. C. (2009). Yemen: Defusing the Saada time bomb. *Middle East Report*(86).
- Gurses, M., Rost, N., & McLeod, P. (2008). Mediating civil war settlements and the duration of peace. *International Interactions*, *34*(2), 129-155.
- Hartley, J. (2004). Case study research. Essential guide to qualitative methods in organizational research, 323-333.

- Hassan, H. A. (2010). Dimensions of the Darfur crisis and its consequences: An Arab perspective. *African Security Review, 19*(1), 20-32.
- Henderson, S., & Levitt, M. (2009). Qatar Challenges Washington on Hamas. *Policy Watch* (1468).
- Henriksen Waage, H. (2007). The 'Minnow'and the 'Whale': Norway and the United States in the peace process in the Middle East. *British Journal of Middle Eastern Studies*, 34(2), 157-176.
- Herriott, R. E., & Firestone, W. A. (1983). Multisite qualitative policy research:

 Optimizing description and generalizability. *Educational researcher*, *12*(2), 14-19.
- Hey, J. A. (2004). Small States in World Politics: Explaining Foreign Policy Behavior. *International Politics*, 41(2), 284-286.
- Höglund, K., & Svensson, I. (2009). Mediating between tigers and lions: Norwegian peace diplomacy in Sri Lanka's civil war. *Contemporary South Asia, 17*(2), 175-191.
- Ingebritsen, C. (2002). Norm entrepreneurs: Scandinavia's role in world politics.

 Cooperation and conflict, 37(1), 11-23.
- Jakubec, P. (2015). Small States Status Seeking: Norway's Quest for International Standing. *International Issues & Slovak Foreign Policy Affairs*, *24*(3), 104.
- Jibril, A. (2010). Past and Future of UNAMID: Tragic Failure or Glorious Success.

 Human Rights and Advocacy Network for Democracy (HAND), 20.

- Johannsen, M., AbuZayyad, Z., El Ouazghari, K., Harik, J. P., Kurz, A., & Rabah, J. (2011). The Reconciliation of Hamas and Fatah. *Policy Brief*(3), 1-8.
- Johnson, D. H., & Matthews, R. O. (2005). The Root Causes of Sudan's Civil Wars.

 International Journal, 60(2), 608.
- Kamrava, M. (2011). Mediation and Qatari foreign policy. *The Middle East Journal,* 65(4), 539-556.
- Karmon, E. (2013). Hamas in dire straits. Perspectives on Terrorism, 7(5), 111-126.
- Kassem, A. S. (2012). Islamic Resistance Movement (Hamas): Performance

 Evaluation. *American International Journal of Contemporary Research*, 2(9),

 128-139.
- Katete, G. O. (2010). The regional outcome of the Comprehensive Peace Agreement five years down the road: Are external forces playing a progressive role?

 International Journal of African Renaissance Studies, 5(1), 161-170.
- Kelleher, A. (2006). A Small state's multiple-level approach to peace-making:

 Norway's role in achieving Sudan's comprehensive peace agreement. *Civil Wars*, 8(3-4), 285-311.
- Kelleher, A., & Taulbee, J. L. (2006). Bridging the gap: Building peace Norwegian style. *Peace & Change, 31*(4), 479-505.
- Kelman, H. C. (1958). Compliance, identification, and internalization three processes of attitude change. *Journal of Conflict Resolution*, *2*(1), 51-60.
- Keohane, R. O. (1969). Lilliputians' Dilemmas: Small States in Internatinal Politics. *International organization*, 23(2), 291-310.

- Kern, S. (2007). Sovereign wealth funds–state investments on the rise. *Deutsche Bank Research*, 10, 1-20.
- Kesteleyn, J., Riordan, S., & Ruël, H. (2014). Introduction: business diplomacy. *The Hague journal of diplomacy*, 9(4), 303-309.
- Kevane, M., & Gray, L. (2008). Darfur: rainfall and conflict. *Environmental Research Letters*, 3(3), 034006.
- Khatib, L. (2013a). Qatar's foreign policy: the limits of pragmatism. *International Affairs*, 89(2), 417-431.
- Khatib, L. (2013b). Qatar's foreign policy: the limits of pragmatism. *nternational Affairs*, 89(2), 417-431.
- Kleiboer, M. (1996). Understanding success and failure of international mediation. *Journal of Conflict Resolution*, *40*(2), 360-389.
- Kleiboer, M. (2002). Great power mediation: Using leverage to make peace. Studies in international mediation: Essays in honor of Jeffrey Z. Rubin, 127-140.
- Klein, M. (2007). Hamas in power. The Middle East Journal, 61(3), 442-459.
- Kochan, T. A., & Jick, T. (1978). The public sector mediation process: A theory and empirical examination. *Journal of Conflict Resolution*, 22(2), 209-240.
- Krøvel, R. (2011). A symbiotic relationship: Norwegian diplomacy and Norwegian journalism on war and peace in Guatemala. *Conflict & Communication*, *10*(1).
- Kydd, A. (2003). Which side are you on? Bias, credibility, and mediation. *American Journal of Political Science*, *47*(4), 597-611.

- Lee, D., & Smith, N. J. (2008). Small states in the international political economy. *The Round Table*, *97*(395), 201-203.
- Leira, H. (2013). 'Our Entire People are Natural Born Friends of Peace': The

 Norwegian Foreign Policy of Peace. Swiss political science review, 19(3), 338356.
- Lieberfeld, D. (1995). Small is credible: Norway's niche in international dispute settlement. *Negotiation Journal*, *11*(3), 201-207.
- Lieberfeld, D. (2008). Secrecy and "Two-Level Games" in the Oslo Accord: What the Primary Sources Tell Us. *International Negotiation*, *13*(1), 133-146.
- Lindner, I., & Strulik, H. (2004). Distributive politics and economic growth: the Markovian Stackelberg solution. *Economic Theory*, 23(2), 439-444 (2004).
- Lundberg, K. (1996). The Oslo channel: Getting to the negotiating table. Cambridge,

 Mass.: John F. Kennedy School of Government Case, 1333.
- Maass, M. (2009). The elusive definition of the small state. *International Politics*, 46(1), 65-83.
- Mac Ginty, R. (2008a). Indigenous peace-making versus the liberal peace.

 Cooperation and conflict, 43(2), 139-163.
- Mahmoud, M. E.-T. (2004). Inside Darfur: ethnic genocide by a governance crisis.

 Comparative Studies of South Asia, Africa and the Middle East, 24(2), 3-17.
- Mans, U. (2004). Briefing: Sudan: the new war in Darfur. *African Affairs*, 103(411), 291-294.

- Mapendere, J. (2000). Consequential Conflict Transformation Model, and the Complementarity of Track One, Track One and a Half and Track Two Diplomacy. *Atlanta, GA: The Carter Center*.
- Mapendere, J. (2005). Track one and a half diplomacy and the complementarity of tracks. *Culture of Peace Online Journal*, *2*(1), 66-81.
- Milton-Edwards, B. (2013). Hamas and the Arab Spring: Strategic Shifts? *Middle East Policy*, 20(3), 60-72.
- Minich, R. (2015). Conflict Mediation: The Qatari Experience. *Historia i Polityka*(14 (21)), 121-133.
- Mohammadzadeh, B. (2017). Status and Foreign Policy Change in Small States:

 Qatar's Emergence in Perspective. *The International Spectator*, *52*(2), 19-36.
- Montville, J. (1991). Track Two Diplomacy: The Arrow and the Olive Branch. *The Psychodynamics of International Relations*, 2, 161-175.
- Moolakkattu, J. S. (2005). Peace facilitation by small states: Norway in Sri Lanka. Cooperation and conflict, 40(4), 385-402.
- Nissen, A., & Waage, H. H. (2015). Weak Third Parties and Ripening: Revisiting

 Norwegian Interventions in Guatemala and the Israeli-Palestinian Conflict. *International Negotiation, 20*(3), 389-413.
- Nuruzzaman, M. (2015). Gulf Cooperation Council (GCC), Qatar and dispute mediations: a critical investigation. *Contemporary Arab Affairs*, 8(4), 535-552.
- Nye Jr, J. S. (2008). Public diplomacy and soft power. *The ANNALS of the American Academy of Political and Social Science*, 616(1), 94-109.

- Pace, M. (2018). Norway's ambiguous approach towards Israel and Palestine. *Global Affairs*, 1-12.
- Panke, D. (2012). Dwarfs in international negotiations: how small states make their voices heard. *Cambridge Review of International Affairs*, 25(3), 313-328.
- Peres, S., & Landau, D. (1995). Battling for Peace: A Memoir. Random House Inc.
- Peterson, J. E. (2006a). Qatar and the world: Branding for a micro-state. *The Middle East Journal*, 60(4), 732-748.
- Peterson, J. E. (2006b). Qatar and the world: Branding for a micro-state. *The Middle East Journal*, 732-748.
- Pruitt, D. G. (1997). Ripeness theory and the Oslo talks. *International Negotiation*, 2(2), 237-250.
- Pruitt, D. G. (2008). Back-channel communication in the settlement of conflict. *International Negotiation, 13*(1), 37-54.
- Quinn, D., Wilkenfeld, J., Eralp, P., Asal, V., & Mclauchlin, T. (2013). Crisis managers but not conflict resolvers: Mediating ethnic intrastate conflict in Africa. *Conflict Management and Peace Science*, 30(4), 387-406.
- Rabbani, M. (2012). Qatar and the palestinians. *PERSPECTIVES*, 42.
- Rabi, U. (2009). Qatar's relations with Israel: challenging Arab and Gulf norms. *The Middle East Journal*, 63(3), 443-459.
- Rathmell, A., & Schulze, K. (2000). Political reform in the Gulf: The case of Qatar. *Middle Eastern Studies*, 36(4), 47-62.

- Reid, L. (2017). Finding a peace that lasts: mediator leverage and the durable resolution of civil wars. *Journal of Conflict Resolution*, *61*(7), 1401-1431.
- Richmond, O. P. (2000). NGOs and an Emerging Form of Peacemaking: Post-Westphalian Approaches. *Paper, International Studies Association (March 2000), CIAO*.
- Rickli, J.-M. (2008). European small states' military policies after the Cold War: from territorial to niche strategies. *Cambridge Review of International Affairs*, *21*(3), 307-325.
- Rickli, J.-M. (2016). New alliances dynamics in the Gulf and their impact on the small GCC states. *Third World Thematics: A TWQ Journal, 1*(1), 132-150.
- Roberts, D. (2014). Qatar and the Muslim Brotherhood: Pragmatism or Preference? *Middle East Policy, 21*(3), 84-94.
- Roberts, D. B. (2012). Understanding Qatar's foreign policy objectives. *Mediterranean Politics*, 17(2), 233-239.
- Rolandsen, Ø. H. (2011). The making of the Anya-Nya insurgency in the Southern Sudan, 1961–64. *Journal of Eastern African Studies*, *5*(2), 211-232.
- Rolandsen, Ø. H., & Leonardi, C. (2014). Discourses of violence in the transition from colonialism to independence in southern Sudan, 1955–1960. *Journal of Eastern African Studies*, 8(4), 609-625.
- Schrodt, P. A., & Gerner, D. J. (2004). An event data analysis of third-party mediation in the Middle East and Balkans. *Journal of Conflict Resolution*, 48(3), 310-330.

- Schulz, M. (2016). Small-State Mediation in International Conflicts: Diplomacy and Negotiations in Israel-Palestine by Jacob Eriksson. *The Middle East Journal*, 70(3), 486-487.
- Shamir, E., & Hecht, E. (2014). Gaza 2014: Israel's attrition vs Hamas' exhaustion.

 Parameters, 44(4), 81.
- Shlaim, A. (2005). The rise and fall of the Oslo Peace process. *International relations* of the Middle East, 241-261.
- Siniver, A. (2006). Power, impartiality and timing: Three hypotheses on third party mediation in the Middle East. *Political Studies*, *54*(4), 806-826.
- Skjelsbaek, K. (1991). The UN Secretary-General and the mediation of international disputes. *Journal of Peace Research*, 28(1), 99-115.
- Smith, A., & Stam, A. (2003). Mediation and peacekeeping in a random walk model of civil and interstate war. *International Studies Review*, *5*(4), 115-135.
- Smith, J. D. (1994). Mediator impartiality: banishing the Chimera. *Journal of Peace Research*, 31(4), 445-450.
- Smith, W. P. (1985). Effectiveness of the biased mediator. *Negotiation Journal*, *1*(4), 363-372.
- Steinberg, R. H. (2002). In the shadow of law or power? Consensus-based bargaining and outcomes in the GATT/WTO. *International organization*, *56*(2), 339-374.

Stokke, K., 2012. Peace-building as small state foreign policy: Norway's peace engagement in a changing international context. *International studies*, *49*(3-4), pp.207-231.

- Svensson, I. (2007a). Bargaining, bias and peace brokers: How rebels commit to peace. *Journal of Peace Research*, *44*(2), 177-194.
- Svensson, I. (2007b). Mediation with muscles or minds? Exploring power mediators and pure mediators in civil wars. *International Negotiation*, *12*(2), 229-248.
- Thorhallsson, B., & Wivel, A. (2006). Small states in the European Union: what do we know and what would we like to know? *Cambridge Review of International Affairs*, 19(4), 651-668.
- Touval, S. (1975a). Biased intermediaries: Theoretical and historical considerations.

 *Jerusalem Journal of International Relations, 1(1), 51-69.
- Touval, S. (1975b). Biased International: Theoretical and hostirical considerations.

 **Jerusalem Journal of International Relations, 1(1), 51-69.
- Touval, S., & Zartman, I. W. (2007). International mediation after the cold war.

 Leashing the dogs of war, 437-454.
- Turner, D. W. (2010). Qualitative interview design: A practical guide for novice investigators. *The qualitative report, 15*(3), 754.
- Waage, H. H. (2005). Norway's role in the Middle East peace talks: Between a strong state and a weak belligerent. *Journal of Palestine Studies*, *34*(4), 6-24.

- Werner, S., & Yuen, A. (2005). Making and keeping peace. *International organization*, 59(2), 261-292.
- Wivel, A. (2013). From Peacemaker to Warmonger? Explaining Denmark's Great Power Politics. *Swiss political science review, 19*(3), 298-321.
- Zambakari, C. (2013). In search of durable peace: the Comprehensive Peace

 Agreement and power sharing in Sudan. *The Journal of North African Studies,*18(1), 16-31.
- Zartman, I. W. (2009). Interest, leverage and public opinion in mediation. *Int'l Negotiation*, *14*, 1.
- Zartman, I. W., & Touval, S. (1985). International mediation: Conflict resolution and power politics. *Journal of Social Issues*, *41*(2), 27-45.
- Zureik, E. (2018). Qatar's humanitarian aid to Palestine. *Third World Quarterly*, 39(4), 786-798.
- Zweiri, M. (2006). The Hamas victory: shifting sands or major earthquake? *Third World Quarterly*, 27(4), 675-687.

Edited chapters

- Egeland, J. (1999). The Oslo accord: Multiparty facilitation through the Norwegian channel. *Herding cats: Multiparty mediation in a complex world*, 527-546.
- KAY, S. (2014). One result of the Gaza conflict: Iran and Hamas are back together.

 In: Time.

- Keashly, L., & Fisher, R. J. (1996). A contingency perspective on conflict interventions: Theoretical and practical considerations. *Resolving international conflicts: The theory and practice of mediation*, 235-261.
- Tvedt, T. (1994). The collapse of the state in southern Sudan after the Addis Ababa Agreement. Short-cut to decay: The case of the Sudan, 186-202.
- Zartman, I. W., Touval, S., Crocker, C. A., Hampson, F. O., & Aall, P. (2007).

 Leashing the Dogs of War: Conflict Management in a Divided World. In (pp. 437-454): United States Institute of Peace Press Washington, DC.

Zartman, W.I., & Touval S (2001). *International mediation in the post-Cold War* era. In: *Turbulent Peace*. Eds. Crocker, C., Hampson, F. & Aall, P., Washington: United States Institute of Peace.

Zartman, W. I., & Touval, S. (1996). International Mediation in the Post-Cold War

Era" dalam Crocker, Chester A (et. al)(eds.). Managing Global Chaos:

Sources of and Responses to International Conflict. In: Washington, DC: USIP

Press.

Internet sources

Aljazeera Channel.(2008). *Darfur Crisis/Interviewer: A. Mansour*. Bela Hudod "Without Boarders"

https://www.aljazeera.net/programs/withoutbounds/2008/9/21/مسؤولية-النخبة-ودور ها-في-/https://www.aljazeera.net/programs/withoutbounds/2008/9/21

- Aljazeera Channel. (2009). Darfur peace talks begin in Doha. Retrieved 20.08.2018

 https://www.aljazeera.com/news/middleeast/2009/02/2009210135517717910.
 https://www.aljazeera.com/news/middleeast/2009/02/2009210135517717910.
- Aljazeera Staff writer. (2010). Signing of the Darfur Peace Agreement in Doha.

 Retrieved from http://www.aljazeera.net/news/arabic/2010/2/23/- توقیع-اتفاق-سلام
- Aljazzera. (2018). The Qatri mediation *Aljazeera Satellite Channel*. Retrieved from http://www.aljazeera.net/home/search?qوي 20%في 20%دارفور = 100% مالكة 100% مالكة
- Aljazzera Staff write. (2017). The most prominent agreements between Fatah and Hamas in 10 years. Retrieved from

 http://www.aljazeera.net/encyclopedia/events/2017/10/12/
 وحماس-في-10-أعوام
- Alkhaldi, W. (2010). The Arab Narrative Critique of the 1948 War. Retrieved from http://palestine.assafir.com/Article.aspx?ArticleID=1517
- Amnsety International. (2007). OCCUPIED PALESTINIAN TERRITORIES: TORN

 APART BY FACTIONAL STRIFE. Retrieved from

 https://www.amnesty.org/en/documents/MDE21/020/2007/en/
- Blanford, N. (2008). Qatari Deal Defuses Lebanese Crisis. *Christian Science Monitor*,

 1. https://www.csmonitor.com/World/Middle-East/2008/0522/p01s05wome.html
- Chacko, K. (2009). Emir launches Qatar project on food security. *Farmlandgrab*. Retrieved from https://www.farmlandgrab.org/8949

- CIA. (2018). Qatar. Retrieved from https://www.cia.gov/library/publications/the-world-factbook/geos/qa.html
- Dabanga. (2010). Qatar charity to fund projects in rural Darfur. Retrieved from https://www.dabangasudan.org/en/all-news/article/qatar-charity-to-fund-projects-in-rural-darfur
- Dickinson, E. (2012b). Qatar builds a brand as mediator. *The Christian Science Monitor*. Retrieved from https://www.csmonitor.com/World/Middle-East/2012/0328/Qatar-builds-a-brand-as-mediator
- <u>Documents related to Oslo negotations.</u>

 https://almashriq.hiof.no/palestine/900/904/ds/
- Dougherty, J. (2001). The Gulf War: Moscow's role. Retrieved from http://edition.cnn.com/2001/WORLD/europe/01/16/russia.irag/index.html
- DW. (2014). Crisis in South Sudan activates international mediators. Retrieved from https://www.dw.com/en/crisis-in-south-sudan-activates-international-mediators/a-17345663
- El-Gamaty, G., (2017). Qatar, the UAE and the Libya connection. (2017). Qatar, the

 UAE and the Libya connection. Retrieved from

 http://www.aljazeera.com/indepth/opinion/2017/06/qatar-uae-libya-connection-170612080219306.html
- Elon, G. (2015). What Israelis Call Palestinians and Why It Matters. *Haaretz*.

 Retrieved from https://www.haaretz.com/israel-news/.premium-what-israelis-call-palestinians-and-why-it-matters-1.5424214

- Erlanger, S. (2007). Hamas Seizes Broad Control in Gaza Strip. *New York Times, 14*. https://www.nytimes.com/2007/06/14/world/middleeast/14mideast.html
- Fabra-Mata, J. (2014). Measuring the effectiveness of Norwegian peace facilitation.

 Retrieved from <a href="https://noref.no/Publications/Themes/Peace-processes-and-mediation/Measuring-the-effectiveness-of-Norwegian-peace-facilitation/(language)/eng-US.%20Access%2026.01.2018
- Faris, S. (2007). The Real Roots of Darfur. *The Atantic*. Retrieved from https://www.theatlantic.com/magazine/archive/2007/04/the-real-roots-of-darfur/305701/
- Farmlandgrab. (2008). Qatar, Sudan in farming tie-up. *Farmlandgrab*. Retrieved from https://www.farmlandgrab.org/post/view/2427
- Group, I. C. (2011). Yemen: Defusing the Saada time bomb. *Página:* http://www.crisisgroup.org/en/regions/middle-east-northafrica/iraq-iran-gulf/yemen/086-yemen-defusing-the-saada-time-bomb.aspx. Consultada, 26(04).
- Gulf Times staff writer. (2017). Qatari- Sudanes ties: Years of Co-operation and Excellence. *Gulf Times*. Retrieved from https://www.gulf-times.com/story/568527/Qatari-Sudanese-ties-years-of-co-operation-and-exc
- GulfBusiness. (2013). New Qatar Emir Says Will Not "Take Direction" In Foreign

 Affairs. Retrieved from https://gulfbusiness.com/new-qatar-emir-says-will-not-take-direction-in-foreign-affairs/
- Halevi, Y. K. (2018). The Real Dispute Driving the Israeli-Palestinian Conflict. *The Atlantic*. Retrieved from

- https://www.theatlantic.com/international/archive/2018/05/israel-palestine-embassy-legitimacy/560291/
- Hancock, S. (2010). Sudan signs ceasefire deal with Darfur rebel group. Retrieved 20.08.2018 http://news.bbc.co.uk/1/hi/8530033.stm
- IMF. (2017). Qatar Gross Domestic product based on purchasing power parity.

 Retrieved from https://www.imf.org/external/pubs/ft/weo/data/changes.htm
- International Crisis Group. (2002b). Dialogue or Destruction? Organising For Peace as the War in Sudan Scalates Retrieved from

 https://www.crisisgroup.org/africa/horn (AbuKoush, 2019)-
 africa/sudan/dialogue-or-destruction-organising-peace-war-sudan-escalates
- International Crisis Group. (2007). *Darfur's New Security Reality*. Retrieved from https://www.crisisgroup.org/africa/horn-africa/sudan/darfur-s-new-security-reality
- Issacharoff, A., & Khoury, J. (2012). Abbas to Lead Palestinian Unity Cabinet,
 Following Hamas-Fatah Deal. *Haaretz*. Retrieved from
 https://www.haaretz.com/1.5182154
- Jasem, A. B. (2012) The Role of Qatar in international politics/Interviewer: A.

 Mansour. Bila Hudud, Aljazeera Channel Doha. www.aljazeera.net
- Kerr, S. (2008, 27/06/2008). Man in the News: Hamad bin Jassim Al Thani. *Financial Times*. Retrieved from https://www.ft.com/content/40daa8a6-4472-11dd-b151-0000779fd2ac

- Kurz, A., Dekel, U., & Berti, B. (2018). *The Crisis of the Gaza Strip: A Way Out*: ha-Makhon le-meḥķere biṭaḥon le'umi. https://ecopeaceme.org/wp-content/uploads/2018/02/GazaCrisis WayOut Full.pdf
- Lyon, A. (2008). Qatar Pulls off Mediation Coup in Lebanon Crisis. *Reuters*.

 Retrieved from https://uk.reuters.com/article/uk-lebanon-qatar-analysis/qatar-pulls-off-mediation-coup-in-lebanon-crisis-idUKL2274043520080522
- Ma'an staff writer. (2012). Qatari emir calls for national reconciliation. Retrieved from http://www.maannews.com/Content.aspx?id=531630
- Maan staff writer. (2012). Abbas: Unity govt will commit to PLO principles. *Maan News Agency*. Retrieved from https://www.maannews.com/Content.aspx?id=461345
- MFA. (2016). Norway's approach to peace and reconciliation work. Retrieved from https://www.regjeringen.no/en/topics/foreign-affairs/peace-and-reconciliation-efforts/innsiktsmappe/norway-peace-work/id446704/
- Munro, A. (2019). Public diplomacy. Retrieved from https://www.britannica.com/topic/public-diplomacy
- Myre, G. (2006). Qatar Emerges as a Mediator Between Fatah and Hamas. *The New York Times*. Retrieved from https://www.nytimes.com/2006/10/10/world/middleeast/10mideast.html

- NORAD. (2002). Study on private sector development in Sri Lanka. Retrieved from Norway https://norad.no/en/toolspublications/publications/2009/study-on-private-sector-development-in-sri-lanka/
- Open Briefing. (2014). *Qatari aid, finance and foreign policy in Sudan*. Retrieved from https://www.openbriefing.org/docs/Qatari-aid-finance-and-foreign-policy-in-Sudan.pdf
- Pfeffer, A. (2014). The Israel-Palestine conflict is not just about land. It's a bitter religious war. Retrieved from https://www.theguardian.com/commentisfree/2014/nov/20/israel-palestine-conflict-religious-war
- Plaestine Center for Human Rights. (2007). Black pages in the absence of justice Report on bloody fighting in the Gaza Strip from 07 to 14 Jun 2007. Retrieved
 from https://reliefweb.int/report/occupied-palestinian-territory/black-pages-absence-justice-report-bloody-fighting-gaza-strip
- Public. (2008, 26.05.2008) The Arab and Foreign mediation in conflict resolution

 /Interviewer: M. Salman. Manber Aljazzera "Aljazeera Forum ", Aljazeera

 Satellite Channel, Doha www.aljazeera.net
- Qatar Red Crescent. (2017). A cooperation Agreemnt between Qatar Red Crescent and Qatar Develoment Fund. Retrieved from https://www.qrcs.org.qa/ar-QA/News/International%20News/2017/August/Development%20Project%20in%20Darfur
- Qatari Ministry of Foreign Affairs. (2012). *Ambassador Al Emadi Calls on Leaders of Palestinian Factions to Unify Ranks*. Qatar: Ministry of foreign Affairs

- Retrieved from https://www.mofa.gov.qa/en/all-mofa-news/details/2016/08/09/ambassador-al-emadi-calls-on-leaders-of-palestinian-factions-to-unify-ranks.
- Ravid, B. (2012). Netanyahu: PA President Must Choose Between Peace With Israel and Peace With Hamas. *Haaretz*. Retrieved from https://www.haaretz.com/1.5182238
- Roberts, D. B. (2015). Qatari Mediation. *Academia*. Retrieved from http://www.academia.edu/336597/Qatari Mediation
- Sawafta, A. (2012). Palestinian rivals agree to form unity government. Retrieved from https://www.reuters.com/article/us-palestinians-government/palestinian-rivals-agree-to-form-unity-government-idUSTRE8150KU20120206
- Sudan Tribune. (2011a). Sudan to hold a donor conference for Darfur development in October. Sudan Tribune. Retrieved from http://www.sudantribune.com/Sudanto-hold-a-donor-conference,39709
- Sudan Tribune. (2011b). Wikileaks: Egypt determined to thwart Qatar initiatives including Darfur. Retrieved from http://www.sudantribune.com/spip.php?article39222
- The Enough Project. (2010). A peace Process Play by -Play. RThe Enough Project.

 Retrieved from

 https://reliefweb.int/sites/reliefweb.int/files/resources/544A4ED52C3AFA64492

 576E300256328-Full Report.pdf

- The Nordic Pace. (2017). Arm Export Increases, Keeping Norway in Top 20 Arms

 Exporter List. Retrieved from https://www.tnp.no/norway/panorama/5498-arm-export-increases-keeping-norway-in-top-20-arms-exporter-list
- The Qatari Institution. (2003). CHAPTER I: THE STATE AND THE PRINCIPLES OF GOVERNING
- . Retrieved from
 - https://www.constituteproject.org/constitution/Qatar_2003.pdf?lang=en
- The Qatari Ministry of Foreign Affairs. (2018). Foreign Policy. Retrieved from https://www.mofa.gov.qa/الدبلوماسية-الوقائية/الوساطة/
- The World Bank. (2018). Norway. Retrieved from https://data.worldbank.org/country/norway
- Tuastad, D. (2013). Have the cake and eat it too: Hamas and PLO elections.

 Retrieved from Norway

 https://www.hf.uio.no/ikos/english/research/projects/new-middle-east/publications/2013/2 tuastad.pdf
- Tveit, Odd Karsten. (2015). Oslo negoatiations missing documents.

 https://almashriq.hiof.no/palestine/900/904/ds/
- Ulrichsen, K. C. (2012). Qatar: Emergence of a regional power with international reach. *E-International Relations. Retrieved from http://www. e-ir. info/2012/01/23/qatar-emergenceof-a-regional-power-with-international-reach.*
- UNAMID. (2011). Doha Document for Peace in Darfur (DDPD) [Press release].

 Retrieved from https://unamid.unmissions.org/doha-document-peace-darfur

- UNDP. (2016). Human Development Report 2016. Retrieved from http://hdr.undp.org/sites/default/files/2016 human development report.pdf
- Waage, H. H. (2000). Norwegians? Who needs Norwegians? Explaining the Oslo

 Back Channel: Norway's Political Past in the Middle East.

 https://www.regjeringen.no/en/find-document/dep/UD/Reports-programmes-of-action-and-plans/Reports/2001/norwegians-who-needs-norwegians/id451594/
- Walid, T. (2009). Qatar SWF food unit eyes PAVA stake, seals Sudan deal. *Reuters*.

 Retrieved from

https://af.reuters.com/article/investingNews/idAFJOE59S0LW20091029?page
Number=1&